

### TAPAL ENERGY (PRIVATE) LIMITED

Corporate Office: F-25 • Block 5 • Rojhan Street Kehkashan • Clifton • Karachi - 75600 • Pakistan Tel : +92-21-35876994 - 7 Fax : +92-21-35876991 & 35876993 Email : telcoff@tapalenergy.com.pk

The Registrar National Electric Power Regulatory Authority (NEPRA) NEPRA Tower Attaturk Avenue (East), Sector G-5/1 Islamabad Pakistan Ref:TEL/NEPRA/006/18 May 22, 2018

# Tapal Energy (Private) Limited ("Company") – Licensee Proposed Modification of the Generation License

The Company was granted Generation License No. IPGL/010/2003 on August 26, 2003 (the "Generation License") by NEPRA, under Section 15 of the Regulation of Generation, Transmission and Distribution of Electric Power Act 1997, for its thermal power generation facility at Deh Gondpass, Tapo Gabopat, Hub River Road, Taluka & District Karachi (West), Karachi (the "Project").

As stated in the Generation License, the Project was granted a Generation License for a term of fifteen (15) years. The Company has entered into a Power Purchase Agreement with KE dated September 26, 1995, ("PPA"), the term of which PPA is 22 years, such that the term of the PPA and Generation License are not aligned.

In order to discharge its contractual obligations under the PPA the Company is required to obtain and maintain a valid generation license granted by NEPRA. The grant of generation license is a critical regulatory consent without which the Company cannot perform its contractual obligations under the PPA and generate and sell electric power.

Given the fact that the term of the PPA is twenty-two years (from the COD) whereas, the term of Company's generation license no. IPGL/010/2003 is around ten months short of the Company's PPA term, the Company requires extension in the generation license in order to make it consistent with the term of the PPA.

The required modification would benefit K-Electric consumers as it would ensure that generation capacity continues to be available and maintained during the term of the PPA.

The Company, pursuant to Regulation 10(2) of the NEPRA Licensing (Application and Modification Procedure) Regulations, 1999 ("**Regulations**"), hereby seeks to apply for a modification of the Generation License granted to the Company to cater for extending the term of the Generation License to bring the same in line with the term of the PPA.

In relation hereto, we certify that the documents-in-support attached with this application are prepared and submitted in conformity with the provisions of Regulation 10(2) of the Regulations and we undertake to abide by the terms and provisions of the Regulations. We further undertake and confirm that the information provided in the attached documents-in-support is true and correct to the best of our knowledge and belief.



A Pay Order No. 00250381 dated May 16, 2018 amounting to Rs. 778,440/- (Pakistani Rupees Seven Hundred and Seventy-Eight Thousand Four Hundred and Forty only) drawn in favour of NEPRA, being the application fee for the present application, calculated in accordance with Schedule II to the Regulations, is also attached herewith.

In light of the submissions set out in this application and the information attached to the same, NEPRA is kindly requested to process the Licensee Proposed Modification of the Company's Generation License at the earliest, thereby enabling the Company to meet its obligations under the PPA

èrely, MUSTAFA LAKDAWALA

(Chief Financial Officer)

### LICENSEE PROPOSED MODIFICATION OF GENERATION LICENSE OF TAPAL ENERGY (PRIVATE) LIMITED TO NATIONAL ELECTRIC POWER REGULATORY AUTHORITY

22<sup>nd</sup> May 2018

CONFIDENTIAL



#### TAPAL ENERGY (PRIVATE) LIMITED

#### REGISTERED OFFICE

155-A, STREET NO. 37, SECTOR F-10/1, ISLAMABAD PAKISTAN TEL: +92-51-2344413 FAX: +92-51-2344411 EMAIL: TELCOFF@TAPALENERGY.COM.PK

.



#### Contents

1.0 Covering Letter	ł
2.0 Affidavit	5
3.0 Board Resolution	7
4.0 Certificate of Incorporation	)
5.0 Memorandum and Articles of Association	)
6.0 Text of the Proposed Modification	l
7.0 A Statement of the reasons in support of the modification; 12	2
8.0 A Statement of the impact on the tariff, quality of service and the performance by the licensee of its obligations under the license	
9.0 Glossary of Terms	4

~ -

.

-

.

.

÷ .-



.



#### AFFIDAVIT

#### BEFORE THE NATIONAL ELECTRIC POWER REGULATORY AUTHORITY

I, Mustafa Lakdawala S/o Abid Hussain Lakdawala, CNIC No. 42201-3140375-9, Chief Financial Officer, Tapal Energy (Private) Limited hereby solemnly affirm and declare on oath that the contents of the accompanying application for a Licensee Proposed Modification of the Generation License of Tapal Energy (Private) Limited, including all attached documents-in-support, are true and correct to the best of my knowledge and belief and that nothing has been concealed.

DEPONENT

Signature:

Name: Mustafa Lakdawala (Chief Financial Officer)

Dated: May 22, 2018



#### CERTIFIED TRUE COPY OF RESOLUTION OF THE BOARD OF DIRECTORS OF TAPAL ENERGY (PRIVATE) LIMITED PASSED ON MAY 2, 2018

#### "It is hereby unanimously resolved that:

- a) Tapal Energy (Private) Limited ("Company") be and is hereby authorized to file a Licensee Proposed Modification of Generation License Application for submission to the National Electric Power Regulatory Authority for modification of the Company's Generation License and in relation thereto, enter into and execute all required documents, make all filings, attend all hearings, provide all required information and pay all applicable fees, in each case, of any nature whatsoever.
- b) In respect of the Licensee Proposed Modification of Generation License of the Company, Mr. Tabish Tapal, Chief Executive Officer of the Company having CNIC No 42301-2385059-9 and/or Mr. Mustafa Lakdawala, Chief Financial Officer of the Company having CNIC No 42201-3140375-9 be and are hereby authorized and empowered for and on behalf of the Company to:
  - (i) Review, execute, submit, and deliver the Licensee Proposed Modification of Generation License and any related documentation required by National Electric Power Regulatory Authority for the award of Modified Generation License including any contracts, documents, powers of attorney, affidavits, statements, letters, forms, applications, deeds, guarantees, undertakings, approvals, memorandum, amendments, letters, communications, notices, certificates, request statements and any other instruments of any nature whatsoever;
  - (ii) Represent the Company in all negotiations, representations, presentations, hearings, conferences and/or meetings of any nature whatsoever with any entity (including, but in no manner limited to National Electric Power Regulatory Authority, any private parties, companies, partnerships, individuals, governmental and/or semi-governmental authorities and agencies, ministries, boards, departments, regulatory authorities and/or any other entity of any nature whatsoever);
  - (iii) Sign and execute the necessary documentation, pay the necessary fees, appear before the National Electric Power Regulatory Authority, as needed, and do all acts necessary for completion and processing of the award of Modified Generation License of the Company from the National Electric Power Regulatory Authority;
  - (iv) Appoint or nominate any one or more officers of the Company or any other person or persons, singly or jointly, in their discretion to communicate with,



make presentations to and attend any hearings in connection with the Licensee Proposed Modification of the Company's Generation License;

(v) Do all such acts, matters and things as may be necessary for carrying out the purposes aforesaid and give full effect to the above resolutions."

Mustafa Lakd vala Company Secretary

GOVERNMENT OF PAKISTAN CERTIFICATE OF INCORPORATION (Under section 32 of the Companies Ordinance, 1984 (XLVII of 1984) Company Registration No. 1-01546 "TAPAL ENERGY LIMITED" I hereby certify that ÷., is this day incorporated under the Companies Ordinance, 1984 (XLVII of 1984) and that sheres the company is limited by Rouse the 1255.71 Gluch Under ny hand. 4 1 March this dav one thousand nine hundled and -five Fee Rs.  $\frac{73,500}{(kupees}$  Seventy eight thousand & five hundred only)  $\mathcal{O}$ (NAZIR AHMED SHANEEN) DEPUTY REGISTRAR OF COMPANIES ISLAMABAD CRO-1 NOT 01/1872 04-1-3-95 FCPPK-Litho/2108/93-94/DCS&F-3-2-94-12000 Loois. ----CERTWIED TO BE TRUE COPY

Assistant Registrar Company Registration Office Islamaoad

32476 123-11=1-3



# THE COMPANIES ORDINANCE, 1984 (Company Limited by Shares)

## MEMORANDUM AND ARTICLES OF ASSOCIATION

### OF

## **TAPAL ENERGY (PRIVATE) LIMITED**

GOVERNMENT OF PAKISTAN CERTIFICATE OF INCORPORATION (Under section 32 of the Companies Ordinance, 1984 (XLVII of 1984) Company Registration No. 1-01646 I hereby certify that "TAPAL ENERGY LIMITED" is this day incorporated under the Companies Ordinance, 1984 (XLVII of 1984) and that shares the company is limited by-Speciet in **Islamab**ad band at Givei mγ March this reh one thousand nine hundred and T hinety-live 78,500/2(Rupess Seventy eicht thousand & rive bundred only) Fee Rs. (NAZIR AHMED SHANEEN) DEPUTY REGISTRAR OF COMPANIES 1 SLAM AB AD CRO-1 1.15- 01/1872 04-1-3-95 PCPPK-Litho/2105/93-94/DCS&F-3-1-94-12000 Loois. CERTIFIED TO BE TRUE COPY Asalstant Begister Company Nonisi/alion Office Islamanad

e sie **GOVERNMENT OF PAKISTAN** CERTIFICATE FOR COMMENCEMENT OF BUSINESS [Under section 146 (2) of the Companies Ordinance, 1984 (XLVII of 1984) COMPANY RECISERATION NO.1-01646 • I hereby contify that the Tapal Energy Limited which was incorporated under the. H Companies Ordinance, 1984 (XEVI) of 1984), on the 1st March and which has filed a duly verified day of. declaration in the prescribed form that the conditions of clauses (a) to (e) of sub-section (1) of section 146 of the said Ordinance have been complied with, is entitled to commence\_business; Islamabad Given under my hand at Мау thia day' ninety-five one undred (NAZIR AHMED SHAHEEN) DEPUTY REGISTRAR OF : C.R.O.4 ISLAMABAD NO. DR/2080 \_\_\_\_ PCPPK-Lund/1044/91-9 -1-7-97-2,000 los

á,

CERTIFIED TO BE TRUE GOPY

And Stant Registrer ست

Company Registration Office Islamatiat

No. JRI Dated\_\_\_\_

COVERNMENT OF PAKISTAN COMPANY REGISTRATION OFFICE ISLAMABAD AKNOWLEDGEMENT OF FILING [ Se-regulation-8 (2)] DR-31 6221-22. Daron; 3-11-99 No. In the author of TAPAT. ENERGY (PRIVATE) LIMITED No.3, First Flar, Malik Complex, Convert, Blue Ares, The carelpt of the undernamioned document (c) filled/regioned/recorded, pursuant to the provision of the Companies Ordinance, 1064 (XLVII of 1984) is hereby acknowledged. 1. Form 26 dated 19-4-1999-Dopy of Special Resolution parted on 19-4-1999 regarding convertion of the company into private company and substitution of Articles of Association. 2. Certified copy of the order dated 16-6-1999 passed by the Securities & Exchange Commission of Pakistan about the conversion of the company into private GULDY BISMILLAH ) DEPHT REGISTRAR OF COMPANIES,

Ħ

Şβ		:			
5					_ KNERGY
tı.			·. i		
		•			
	7	FHE COMPANI	ES ORDINAN	ICE, 1984	
		(A Company	Limited by Sl	hares)	
	M	IEMORANDU	M OF ASSO	CIATION	-
		•	OF		·
- - -	TAI	PAL ENERGY	(PRIVATE)	LIMITED <sup>1</sup>	
	l) The name	of the company is TAPAL E	VERGY (PRIVATE) LIMI	TED.	
	II) The Regist	stered office of the company v	vill be situated in Islamat	ad, Capital Territory.	
	underlakin Iransmissio	and exclusive object for wh ng in power sector to carry or ion and distribution thereof in d in such manner as may be	n the business of electric n all its branches and as	power generation, accu pects by the use of such	mulation, i
	To achieve	e the main and exclusive obje	ect the Company shall be	authorised:-	
		, purchase and acquire by so all, transmit and deliver the el			erial, and
	2. To purchas equip, mai constructio	se, or acquire by some other n intain or improve any buildir on thereof.	neans, any land or lands ng, structure, ediffice on	and build, grect, construc for Sale and bire, and to	t furnish, carry on
1	rights, sdv: marks, pai- and to emp	by purchase, exchange, hire, vanlages, moveable and imm lents or inventions, licenses to ploy, resell, sell, exchange, mo pertles, rights or privileges and	oveable property of any o use patents, of other pr ortgage, get on lease, lic	kind what soever, machin operties, plants and stock sinse to use or otherwise a	ery, trade k-in-irade
		: limited company to a private lumited co	24.0		i Maeting
			•••		
					,

 $i_{i}$ 

· · · · ·	, TAAALy		9 (
۱ <u>.</u>	4, To a , loca	arrange for buying all kinds of raw material, plant and machinery, equipments and tools whether al or imported, on cash, toan, deferred credit, pay-as-earn or non-repatriable investment basis.	(
ļ	5. To a <u>proj</u>	arrange electricity, water, gas, sewerage and other utilities required for efficient running of the ect.	
	6. To a the l	ppoint agents, sub-agents, attorneys, consultants, brokers and contractors in connection with business of the Company but not to act as managing agents.	
i	in pa tion	dopt such means of making known the products of the Company as may seem expedient, and articular by advertising in the press and electronic media, by circulars, by purchase and exhibi- of works of art or interest, by publication of books and periodicals, and by prizes, rewards, ands and donations.	
:	geou obje	arry on any other business or trade which can, in the opinion of the Company, be advanta- usly or conveniently carried on by the Company in connection with the main and exclusive ct. If calculated directly or indirectly, to develop any branch (es) of the Company's business or to asso the value any of the Company's property, assets or rights.	
,	local the C com( Com	Iter into any arrangement with any governments or authorities (Federal, Provincial, Municipal, or otherwise), or any corporations, companies, firms, or persons that may seem conductive to Company's exclusive object, and to obtain from any such government, authority, corporation, pany or person any charters, contracts, ducrees, rights, privileges and concessions which the pany may think desirable, and to carry out, exercise and comply with any such charters, con- s, decrees, rights, privileges and concessions.	
	eilhe orex	ell or otherwise dispose off all goods, materials, articles and things belonging to the Company or on cash or on credit and either for immediate or future delivery and to send the same for sale port to any place that may be deemed necrissary or expedient in the event of winding up of the pany.	
	the cy of set by the	It insured against losses, damages, risks, accidents and liabilities of all kinds which may affect ompany whether in respect of its contracts, agreements, advances or securilles or in respect rvants or employees of the company, or in respect of property belonging to or leased to or thired a company, either by setting apart funds of the company or by effecting such insurance and in case to pay the premium thereon.	
	draw, chequ	pen, close and operate bank accounts with scheduled banks or financial institutions and to make, accept, endorse, discount, execute and issue promissory noted, bills of exchange, ues, bills of lading, warrants, debentures and other negotiable or transferable instruments, eming the business of this Company.	
	acqui and to	vest the surplus moneys of the Company not Immediately required in and subscribe for, take, ine and hold shares, debentures, securities of any other company of corporation whatsoever, o invest moneys of the Company in any other manner, including the purchase of any book or ther debts without doing the business of an investment company within the meaning of the	
	foreig	frow or produce on mark-up, profil or return in any form, money or finances, in local or any in currency from any bank or financial institution and to receive money on mark-up and/or st by issuing debentures, and on security of any such money, so borrowed or received to	
,		Page 2	

the second s

and the second secon

F

• 1<sup>2</sup>

ليح<del>صر</del>ي ويدهمه المردوبين المحا

mortgage, plagge, charge or hypothecate whole or any part of property, assets or revenue of the Company (both present and future) including its capital by special assignment of otherwise, to transfer or convey the same conditionally, absolutely or in trust and to give, tender power to sell and other powers as may seem expedient, and to purchase or redeem such securities and pay for such borrowing and loans.

To procure or arrange finances from scheduled banks and financial institutions under any mode of Islamic financing scheme like, redeemable capital including modaraba and musharaka and to procure, raise or to secure the money in such manner as the company may deem fit and particularly by mortgage of its property in full or in part on both the present and future assets in accordance with Islamic Laws and/or by the issue of shares, bonds, depentures, participation term certificates, Term Finance. Certificates, or redeemable capital or any other securities charged or based upon the udentaking of the company, on any part of its property, both present and in future and generally to borrow money for the purposes of the business of the company, in such manner as the company shall deem fit. To issue debentures or participation term certificates, redeemable capital, either permanent or redeemable or repayable or convertible into shares and to secure any securities of the company by a trust or other assurances.

15

19.

20

21.

- 16. To pay commission or otherwise remunerate any company or firm or firms or person or persons (whether an officer of this Company or not) for services rendered in placing or assisting to place any of the shares of the Company's capital or any debentures or other securities of the Company, or for negotiating any of the purchases or sales by the Company, or for rendering any service of any kind whatsoever to the Company.
- 17. To produce the registration on other recognition of the Company in any country, state or place and to establish and regulate agencies and open branches in any part of the world for the purposes of the Company's business.
- 16. To pay the costs, charges and expenses preliminary and incidental to the formation, establishment and registration of the Company, and to remunerate any person or company for services rendered or to be rendered in placing or assisting to place or guaranteeing the placing of any of the shares in the Company's capital, or any debentures or other securities of the Company, or in or about the conduct of its business.

To grant pensions, allowances and bonuses to officers, ex-officers, employees or ex-employees of the company or its predecessors in business or the dependents or connections of such persons and establish and support or ald in the establishment and support of associations, institutions, funds, trusts and conveniences calculated to benefit employees and ex-employees and officers and ex-officers (including Directors and ex-directors) of the Company, or the dependents or connections of such persons, and to pay gratuities or grant pensions and allowances, and to make payments towards insurance, and to subscribe or guarantee more for charitable or benevolent objects, or for any exhibition, or for any public, general or useful objective matching.

To distribute any of the property of the Company amongs, the members in specie or kind and in particular any shares, debentures or securities of other companies belonging to this Company, or of which this Company may have the power of disposing in the event winding up of the Company.

To create any depreciation fund, provident fund, reserve fund, sinking fund, insufance fund, or any other special fund conducive to the interest of the company.

19 AND

Page 3

H	~~ ^j		· · · · · · · · · · · · · · · · · · ·	<b>1 )</b> F
				. (
1	<b>ત</b> ્ `	T		19 O I
		LAPAL Lineay		1 Or
		. 22.	To capitalise such portion of the profits of the Company as are not distributed amongst sharehold- ers of the Company in the form of dividend and as the Directors of the Company may think fit and to Issue bonus shares as fully pald-up in favour of the shareholders of the Company.	
		<u>23.</u>	To issue any shares of the Company as fully pald-up at par or at a premium or at a discount as provided by law.	
1		24.	To remunerate Directors, officials, servants of the Company or any other person or firm or company rendering services to this company, out of, or in proportion to the returns of profils of the company or otherwise as the Company may think proper, alther by cash payment or by the allotment to him or them shares or securities of the Company credited as paid-up in full as may be thought expedient.	-
-		25.	To appoint such persons or firms as may be seem expedient to be general managers, secretaries, managers, branch managers, or district representatives of the Company upon such terms as the Company may determine.	
		26	To establish and maintain branches, receiving offices and distributing centers and to enter into contracts or agency agreements (other than managing agency) with any other persons or firms or companies or for the distributing centers for the efficient carrying on of the business of the Com- pany.	
	1	27.	To undertake and execute any trusts which the company may think fit and expedient to undertake.	
		28	To apply for, purchase or other wise acquire, and protect and renew any patents, patent rights, brevets d'invention (trade marks, designs, etc.), licenses, concessions and the like, conferring any exclusive or non-exclusive or limited right to their use.	
		29.	To establish, Improve, manage laboratories, research and development centers to perform such research and development as the Company may deem advisable or feasible, and to expend money on experimenting upon testing and improving or securing any process or protecting any invention (s) which the Company may acquire or propose to acquire or deal with.	
	1	30.	To develop and/or transfer technology and to acquire or pass on technical know-how.	
			To train personnel and workers, in Pakistan and/or abroad, to obtain technical proficiency in various specialities connected with the business of the Company.	1 
		32.	It is hereby declared that:-	· ·
			(a) the word "company" in the above clauses except where used in reference to this Company shall be deemed to include any body of persons incorporated in Pakistan or elsewhere:	
	4	r,	(b) the main and exclusive object for which the company is set up are indicated in Clause III. The Company shall have full authority, power and competence to do any and all other inlings and acts to further the purposes specified in Sub-clauses 1 to 31 in support of and in relation, whether directly or indirectly, to the said exclusive object of the Company.	
	uť		(c) notwithstanding anything contained in the foregoing object clause of this Memorangum of us / Association, nothing herein shall be construed as empowering the Company to undertake or indulge in the business of banking, finance, leasing, investment or insurance, directly of	
			Page 4	( '
	ĺ			
			· · · · · · · · · · · · · · · · · · ·	<u> </u>

<u>,</u>,

I TAP indirectly, as restricted under law or in any unlawful operation and that nothing in the op-jects clause shall be construed to entitle it to engage in such business. N The liability of the members is limited, ΊV. The authorised capital of the Company is Rs. 1,500,000,000 (Rupees Fifteen Hundred Million) divided into 150,000,000 ordinary shares of Rs. 10 each with powers to increase and reduce the capital of the Company in such manner as may be consistent with the provisions of the Companies Ordinance, 1984. V. Ofdinance, 1984. Ĩ Page 5 ł

Ì.,1

	Nationality Korth auy Korthau Lationality PAKISTANI PAKISTANI PAKISTANI PAKISTANI PAKISTANI AMERICAN		Residentral Address Address 6-A. 15T GIZRI LANE, D.H.A., KARACHI. F898, CLIFTON, KARACHI. 6-A. 15T GIZRI LANE, D.H.A., KARACHI. D.H.A., KARACHI. T. YUL M. M. M.	Numble fol Anares (a) for by critic by critic c c c c c c c c c c c c c c c c c c	
		· ·	6-A. IST GIZRI LANE, D.H.a., KARACHI, K98, CLIFTON; KARACHI, 6-A. IST GIZRI LANE, D.H.A., KARACHI, D.H.A., KARACHI, T. FULX, M.D.A.	6.250 6.250	
			K98, CLIFTON, KARACHI. 6-A. IST GIZRI LANE. D.H.A., KARACHI. X.ET. 3990, M.	6,250 6,250 6,250	
			6-A. IST GIZRU LANE, D.H.A., KARACHL K.P.I. 39/B, M.	. e,250 . 6,250	
			K.ET 39/B. M. T VU AN POAD		
			LALAZAR, KARACHI.	-	. (
<u></u>	•		27-D, SEABIRD LAND DISCOVERY BAY LANTAU ISLAND HONG KONG.	ريه 	CENTRED TO AS TRUED
MR. DAVID WILFRED KARSNER	•••	Enspress	6-C. GREENBELT COURT DISCOVERY BAY, LANTAI ISLAND, HONG KONG	,, 	- Marine -
MR. EVALD BJORKMAN			65920 N. VAELGRUND VASA, FINLAND	۲۹ 	The second second
•		· · · · ·	14TH FLOOR, LUK KWOK CENTRE, 72 GLOUCESTOR ROAD, WANCHAI, HONG KONG	26.52	Company sees it another Ulthor
				50 <sup>,</sup> 000	
-21 -	995				
	-	iness to the above \$	signatures		ло то то т
Full Name. Father stHusband's Name	Signature	Occupat		dress 	
·····		•	. ·		1 2 1 4
•••••••	1		• • •		! 
		WILFRED AMERICAN FIINLAND 1995 Signature	NCE G. AMERICAN WILFRED AMERICAN FINLAND HONG KOW	NCE G. AMERICAN ENGINEER LALACAK KARA BUSINESS LANCONERY BAY AMERICAN BUSINESS LANDANT KONG. MILFRED AMERICAN BUSINESS CORENDELI PHNLAND MARKETING 65920 N. VALLOR HONG KONG FORER VASA FILLAND HONG KONG FORMER VASA FILLAND HONG KONG FORMER VASA FILLAND HONG KONG COMPANY SILAND I TH FLOOR LU 1995 Witness to the above signatures Signature Occupation	NCE G. AMERICAN ENGONEERU T. KHAN ROAD HOUGE C. AMERICAN ENGONEERU D. LAND ENGONEERU D. LAND ENGONEERU D. LAND BUSCOVERY BAX, LANTAU ISLAND HONG KONG ROAD HONG KONG HONG KONG ROAD HONG KONG T. I HONG KONG ROAD HONG KONG ROAD HONG KONG ROAD HONG ROAD COMPANT Signature Occupation Full Address Signature Occupation Full Address

્યુત્ર

and a state of the second s

E E

<u>(</u>)

$\langle$					i	<u>85</u>		
	) 				r 1		TAML	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)
Vij			,					
		,	THE COM	PANIES OI	RDINAN	CE, 1984		
			(Comp	any Limite	d by Sha	res)		
			ARTICL	ES OF AS	SSOCIA'	<b>TION</b> <sup>†</sup>	•	
-!				OF				
J		ТА	PAL ENEI	RGY (PRI	(VATE) I	LIMITEI	) <sup>2</sup>	
		÷-	\$ ·· * *	PRELIMIN		an the second		
KI K	1.	Table A n	ot to apply					
יזן י		apply to th	fations contained in t he Company excepts hese Articles or by the	so far as the same a				
. ' .	2.		pinal notes hereto sha cmething in the subje				resents, unless	Ň
		i) *4 Bi	Affiliate" shall mean, ng, Controlled by or u	with respect to a M inder common Con	ember, any Pers Irol with such Me	on directly or ind ember;	irectly Control-	т, т,
		Ċ	Agreemen!" means li company, Silhe, Maru mended and/or restal	beni, Kingpin and ti	he Tapel Shareh			
	,	b	Agreed Equity" mean eing US\$ 33,774,131 greement);					
ļ			Articles" méans these ine;	Anticles of Associat	ion as altered by	Special Resolut	on from time to	S. A.
			Board" means the Bo alled and convened;	oard of Directors fo	r the time being	i at à meeting du	ily constituted,	
. ,	<sup>7</sup> Conver		Resolution passed a) on Exi o timiled company to a prive			sed at an Extractionary	r Genoral Meeting	•
•	• •	•					-	
						. `		
_i								

¥¥,

.

$\sim$		Ś.	;	1
			. (	ג ר
	TAPAL ENERG	¥		
		VÌ)	"Chief Executive" means the Chief Executive for the time being of the Company, by what- ever name called, appointed pursuant to Section 198 of the Ordinance;	
•		vii)	"Company" means Tapal Energy (Private) Limited;	Υ.
		vili)	"Complex" means the approximately 126 MW, (gross ISO) power station utilising residual fuel oil, to be owned and constructed by the Company off Hub River Road, in the District of Karach) West, Province of Sindh, Pakisian whether completed or at any stage of its con- struction, including without limitation or regard to tevel of development, land, engineering and design documents, all energy producing equipment and it auxiliary equipment, fuel storage and handling facilities and equipment, a switchyard, interconnection facilities (other than the Metering System, the Interconnection Facilities and the Transmission Facilities, all as defined in the Power Purchase Agreement) necessary for delivery of electricity to the Karachi Electric Supply Corporation Limited at the Interconnection Point (as defined in the Power Purchase Agreement);	ر ب ب ب ب
		ix)	*Contractors" means each of Power Construction Company, B.V. and Wartslia NSD Nederland B.V. (formerly known as Stork-Wartslia Diese) B.V.);	
		ж)	"Construction Contract" means the Engineering and Construction Contract dated as of 21 December 1995 between the Company and Power Construction Company, B.V.;	
	,	xi)	"Control" shall mean the power of a Person (the "Controlling Person") to direct the manage- ment and policies of another Person whether through the ownership of voting securities, by contract or otherwise, and shall include the power to control material management and policy decisions of any Person by virtue of a contractual, corporate or partnership provision requiring the affirmative vote or approval of the Controlling Person, and "Controlling" and "Controlled" shall be construed accordingly;	8 
		xłi)	"Dividend" includes bonus;	
	ŀ	Xiii)	"Directors" means the Directors including an alternate director for the time being of the Company;	. •
		xiv)	"Equipment Supply Contract" means the Equipment Supply Contract dated as of 12 Octo- ber 1996 between the Company and Wartsila NSD Nederland B V. (formerly known as - Stork-Wartsila Diese) B.V.);	
	••	XV) .'	"EPC Guaranty" means the agreement dated 20 December 1995 between Wartslia NSD Nederland B V. (formerly known as Stork-Wartslia Diesel B.V.) and the Company	,
		xvi)	"Existing Loan" means the existing toan facility made available by Metra Finance OVAB on 14 December 1995 to the Company;	
. ,		xvii)	"Financing Documents" means any loan agreements, notes, indentures, security agreements and other documents, evidencing or securing debt incurred by the Company for the financing or refinancing of the Project;	
 		xviii)	Fiscal Quarter" means each three Month period ending on March 31, June 30, September 30 and December 31 of each Fiscal Year;	

н 7 Х.,

P.

· - - - -

		Page 3	
. 2 3 .!	(ii)	secondly, payment when due and payable of interest, fees and principal of leans of other obligations of the Company;	
-Tunbootter	()	first, payment of any expenses, including: (1) payroll and employee benefits, including pay- roll taxes;(2) taxes and assessments, and fees and expenses incurred in connection with the preparation of the Company's tax returns;(3) insurance premiums;(4) without limitation, all other expenditures, including capital expenditures, required or deemed advisable to be made by the Board; and (5) operating expenses; and	
	xxx)	"Net Cash Flow" means, for any period of the Fiscal Year, including a Fiscal Quarter, all of the Company's cash receipts, including the proceeds of borrowing and any other income from any source, and reduced by the following cash disbursements, which shall paid by the Company in the following order or priority;	
	xxix)	"Month" means calendar month;	
-	XXVİİI)	"Marubeni" means Marubeni Corporation, a corporation organised under the laws of Japan and such successors and assigns of it as may be permitted from time to time;	
-	XXVII)	"Member" means a member of the Company in accordance with the provisions of Section 2(1)(21) of the Ordinance;	
() • • •	xxvi)	"Long Term Maintenance Agreement" means the Long Term Maintenance Agreement en- tered or to be entered into between the Silhe Operator and Wartsila Pakistan or an Affikiate of Wartsila Pakistan reasonably acceptable to the Silhe Operator;	
	xxv)	"Lien" means any mongage, piedge, lien, glaim, security interest, charge or encumbrance or any obligation to create a lien, charge, piedge, security interest, encumbrance or mon- gage;	
d ·	xxiv)	"Kingpln" means Kingpln Investments Limited, a corporation organised under the laws of Hong Kong and such successors and assigns of it as may be permitted from time to time;	
	xxiii)	"Implementation Agreement" means the agreement made between the President of the Istamic Republic of Pakistan for and on behalf of the Istamic Republic of Pakistan and the Company dated 8 August 1995 as amended by Amendment No.1 made between the Presi- dent of the Islamic Republic of Pakistan for and on behalf of the Islamic Republic of Paki- stan and the Company dated 18 April 1996;	
	xxii)	"Groups means any of Sithe, Marubeni, Kingpin and collectively the Tapal Shareholders;	ľ
·   _	xxi)	"Guarantee" means the Guarantee made between the President of the Islamic Republic of Pakistan for and on behalf of the Islamic Pepublic of Pakistan and the Company dated 18 April 1996;	
	, xx)	"Fuel Supply Agreement" means the Fuel Supply Agreement made between Pakistan State Oil Company Limited and the Company as of 7 March 1996;	
للتنهك	xix)	Fiscal Year' means the year ending June 30;	· .
3_		TAMAL INENGY	
_ \ '			(n)

۲<sub>1</sub>

36.5		
ć. 🎽		
	· ·	
ENI	ÂĞY	// // _// _// _// _// _// _// //
	(lxox)	"New Capital" means any funding in addition to the Agreed Equity;
	xxxii)	"Office" means the Registered Office for the time being of the Company;
	xxxiii)	"Operating and Maintenance Agreement" means the Operating and Maintenance Agree- ment made on 1 December 1995 between the Company and the Operator as amended and/or restated from time to time;
	xxxiv)	"Operator" means Westsila Pakistan or, following the relinquishment by Wartsila Pakistan of its rights and the essumptions by the Sithe Operator of Wartsila Pakistan's obligations arising from and after the date of such assumption under the Operating and Maintenance Agreement, the Sithe Operator:
	`xxxxV)	"Ordinary Resolution" means a resolution passed at a general meeting of the Company when the votes cast (whether on a show of hands or in poli) in favour of a resolution by Members who being entitled to vote in person or by proxy, do so vote, exceed the number of votes of any cast against the resolution by Members so entitled and voting;
	XXXVI)	"Ordinance" means the Companies Ordinance, 1984 or any statutory modification or re- enactment thereof for the time being inforce;
•	xxxvil)	"Overruns and Shortfalls" means all amounts that increases the Agreed Equity and each Member's proportionate share thereof on account of the following factors;
	0	If due to constraints on availability of debt financing in the form of Working Capital and refinancing of the Existing Loan, the Members agree pursuant to Article 57 that the amount of equity financing required is higher that the amount of the Agreed Equity; or
	(ii)	any call or drawing in any guarantees, letters of credit or other financial support provided by the Members pursuant to the Agreement;
• · · · : : i	xxxxvili)	"Person" means an individual, partnershir, corporation, company, joint venture, associa- tion, trust, incorporated organisation, or a government;
	xxxix)	"Project" means the development, design, engineering, manufacture, financing, procure-
	,	ment of the Site, construction, permitting, installation, completion, testing, commissioning, insurance, ownership, operation and maintenance of the Complex and all activities inciden- tal thereto;
	xi)	"Project Costs" means the total cost of the Project approximately US\$141727,131
, <sup>,</sup> ,	, XII)	"Project Documents" means the Implementation Agreement, the Guarantee, the Fuel Sup- ply Agreement, the Power Purchase Agreement, the Construction Contract, the Equipment Supply Contract, the EPC Guaranty, the Operating and Maintenance Agreement, the Long Term Maintenance Agreement, the Memorandum and Anticles of Association of the Com- pany and any land purchase or title documents pertaining to the Sile for the location of the Project and each other material agreement and document relating to the Project, excluding the Financing Documents, entered into by the Company after 21 October 1997 (including, land purchase documents for the Islamabad property):
		And the state of t
		Pagé 4

· · · · ·			 
$\{\cdot, \cdot\}$			5
			3)
ļΟΥ		TAPAL	/
	xlii)	"Power Purchase Agreement" means the Power Purchase Agreement made between the Karachi Electric Supply Corporation Limited and the Company dated 26 September 1995 as amended by Amendment No.1 made between the Karachi Electric Supply Corporation Limited and the Company dated 9 May 1996;	
	xilii) Xliv)	"Proxy" includes an attorney duly constituted under a power of attorney; "Register" means the Register of Members to be kept pursuant to Section 147 of the Ord- nance;	1
	xivi)	"Seal" in relation to a Company means the Common Seal of the Company;	
	xlvii)	"Secretary" means the Secretary for the time being of the Company;	
	xlviii)	"Shareholder Loan" means a loan from a Member (or an Affiliate of a Member) to the Company to fund Agreed Equity, New Capital, Working Capital or Overruns and Shortfalts;	ļ
	xlix)	"Site" means the land, spaces, waterways, road, wells and any rights acquired or to be acquired by the Company for the purposes of the Complex on, through, above or below the ground on which the complex or any part thereof is to be built (including, without limitation, any working areas required by the Company and the Contractors, villages, township, and comps for the accommodation of the employees of the Company and the Contractors and any subcontractors; and, all rights of way and access from public highways, railways, and seaward access, if applicable):	!
	I)	"Sithe" means Sithe Maunitus Limited, a limited liability company organised under the laws of Mauritius and such successors and assigns of it as may be permitted from time to time.	
· · ·	ii)	"Sithe Operator" means one or more Affiliates of Sithe that are reasonably acceptable to the Company;	
↓ ′ - , ₂	16)	"Special Resolution" has the meaning assigned thereto by Section 2(1)(36) of the Ordinance;	
i Vi	(iii)	"Tapal Shareholders" means Moiz S. Tapal, Dannish A. Tapal, Tabish M. Tapal, Shakil A. Tapal, Tajwar M. Tapal, Muslafa A. Tapal, Amir S. Tapal, Zafar A. Tapal and the permitted transferees of such shareholders as provided under the Agreement;	
	liv)	"Term Sheet" means collectively (i) Summary of Key Terms and Conditions dated 23 De- cember 1996 initialled by the Company, ING Bank N.V. and Merita Bank Limited, (ii) Sum- mary of Terms and Conditions for the FMO Term Loan Facility dated 13 January 1997 between the Company and Nederlandse Financierings, Maatchappi), Voor Ontwikkelingslanden N.V. and, (ii) Dratt Term Sheet, Finaturd Loan Facility for Tabal En- ergy Limited dated 13 March 1997 from Finnish Fund for Industrial Cooperation Ltd. and accepted on behalf of the Company on 18 March 1997.	
g + 1	W)	Wartsila Pakistan" means Wartsila NSD Pakistan (Pvi) Limited formerly known as Wartsila Diesel Pakistan (Pvi) Limited, a private limited company organised and existing under the laws of Pakistan;	
1	•		
		Page 5	
er I		· · · · ·	
• •			

Ň.,

.

318	<b>)</b> .	
		.\$0
, ,	UBNERGY	W) "Working Capital" means short term loans in the amount of US \$10,500,000 and a short term loan in the amount of US\$ 453,000 supported or provided by Kingpin to finance part of the Project Costs.
		Miy Words importing the singular number shall include the plural number and vice versa:
		(vili) Words importing the masculine gender shall include the feminine gender.
		fix) Expression referring to writing shall unless the contrary intention appears, be construed as including references to printing, (ithography, photography and other modes of representing or reproducing words in a visible form.
	3.	Unless the context otherwise requires words or expressions contained in these Articles shall bear the same meaning as in the Ordinance.
		PRIVATE COMPANY
	4.	The Company is a private limited company and accordingly:
		(a) the right to transfer shares of the Company is restricted in the manner hereinafter provided;
•	• .	(b) the number of Members for the time being of the Company (not including persons who are for the time being in the employment of the Company) shall be limited to fifty (50): Provided that, for the purpose of this provision where two or more persons hold one or more shares in the Company jointly, they shall be treated as a single number; and
		(c) an invitation to the public to subscribe for any shares or dependences of the Company is hereby prohibited.
	·	CAPITAL
	<b>5</b> ,	The authorised share capital of the Company is Rs. 1,500,000 (Rupees fifteen hundred million) divided into 150,000,000 ordinary shares of Rs. 10 (Rupees Ten) each with powers to increase and reduce the capital of the Company in such manner as may be consistent with the provision of the Ordinance.
		SHARES
	6.	The Company shall not lesue partly paid shares.
<i>.</i>	7.	Except to the extent and in the manner allowed by Section 95 of the Ordinance nopart of the funds of the Company shall be employed in the purchase of, or in loans upon the security of the Compa-
	8.	ny's shares. The Directors may with the sanction of an Ordinary Resolution of the Companyin general meeting, Increase the authorised share capital by such surn as they think fit, to be divided into shares of such

Ħ

increase the authorised share capital by such such as they think fit, to be divided into shares of such amount as the resolution may prescribe, subject, nevertheless, to the provisions of Section 92 of the Ordinance.

Subject to compliance with the requirements of Articles 57 and 98 and Section 86 of the Ordinance where the Directors decide to increase the capital of the Company by the Issue of further shares,

nia niječki ministrije

Page 6

either at a premium or at par, such shares shall be offered on such terms and conditions and for such consideration, and at such times as the Directors deem fit, to the Members strictly in propotion to the existing shares held by each Member; Such offer shall be made by notice specifying the number of shares to which the Member is entitled and limiting a time within which the offer, if not accepted, will be deemed to be declined; and after the expiration of such time or on receipt of an intimation from the Member to whom such notice is given that such Member declines to accept the shares offered, the Directors may dispose of the same in such manner as they think most beneficial to the Company. The Company may at any time pay commission to any person for subscribing or agreeing to sub-10 scribe (whether absolutely or conditionally) for any shares, debentures or debenture stock of the Company, or procuring or agreeing to procure subscriptions (whether absolutely or conditionally), for any shares, debentures, debenture stock of the Company, but if the commission in respect of shares shall be gaid or payable out of capital, the conditions and requirements laid down in Section 82 of the Ordinance shall be observed. The amount or rate of commission shall not exceed any statutory limit thereon. The commission may be paid or satisfied in cash or in shares, debentures of debenture stock of the Company. 11. Except so far as otherwise provided by the conditions of issue or by these presents, any capital raised by the creation of new shares, shall be considered part of the original capital and shall be subject to the provisions herein contained with reference to the payment, transfer and transmission surrender, voting and otherwise. 12. The Company may from time to time by Special Resolution reduce its share capital in any way and in particular (without prejudice to the generality of the power) by paying off capital or cancelling capital which has been lost or is unrepresented by available assets or reducing the liability on the share or otherwise as may seem expedient and capital may be paid off which is in excess of the needs of the Company or otherwise, and paid up capital may be cancelled as aforesald withou reducing the nominal amount of the shares by the like amount to the extent that the unpaid and callable capital shall be increased by the like amount. Save as herein otherwise provided, the Company shall be entitled to treat the registered holder o any share as the absolute owner thereof and accordingly shall not, except as ordered by a court o 13. competent jurisdiction, be bound to recognise any equitable, contingent or partial interest in or an other right in respect of such share on the part of any other person. Shares may be registered in the name of any limited company or other body corporate but not in the 14. name of a minor or a firm. Not more than four persons shall be registered as joint holders of any shares. ansiny Roman If any share stands in the name of two or more cersons, the person first named in the Register shall 15. as regards receipt of dividend or service of nolice and all or any other matters connected with the Company except voting at the meeting and the transfer of shares, be deemed the sole holder In the case of the death of any one or more of the persons named in the Register as the joint-holder of any share, the survivor or survivors shall be the only person or persons recognized by the com-16 pany as having any title to or interest in such share, but nothing herein contained shall be taken to release the estate of a joint holder from any liability. Page 7 .

64.9	)		<u> </u>
:			S ~
	TAPAL	· · /	ر کسا جنہ
	17.	Every Member shall name to the Company a place in or out of Pakistan to be registered as his address and such address shall for all purposes be deemed to be his place of residence.	Â.
	18,	Subject to the provisions of Section 92 of the Ordinance the Company may by Ordinary Reso lution:	•
1 '		<ul> <li>a) consolidate and divide the whole or any part of its share capital into shares of larger amount than its existing shares;</li> </ul>	-
		<li>b) sub-divide its shares or any of them into shares of smaller amount than is fixed by the Memorandum of Association;</li>	
	•	c) cancel any shares which at the date of passing of the resolution have not been taken or agreed to be taken by any person.	•
		The resolution by which any share is sub-divided or consolidated may determine that as between holders of shares, resulting from sub-division or consolidation, rights of profits, votes and other benefit attaching to them will be proportionate to their paid up value and where shares issued as sub-divided or consolidated shares are of same class as those previously issued that rights attaching to them, subject as aforesaid, shall be the same as those attaching to the shares previously held.	
	19.	Subject to the provisions of the Ordinance and these Articles, the Directors may allot and issue shares in the capital of the Company as payment or part payment of any property, sold or trans- ferred, or in discharge of any indebtedness or obligations of the Company, or for goods or machin- ery supplied or for services rendered to the Company in or about the formation or promotion of the Company or the conduct of its business, and any shares which may be sold or allotted may be issued as fully paid-up shares, and, if so issued, shall be deemed to be fully paid-up shares.	4 17
		CERTIFICATE	
	•	Every person whose name is entered as a Member in the Register shall without payment be entitled to receive, after allotment or registration of transfer, one certificate for all his shares or several certificates each for one or more of his shares and upon payment of such charges, if any, as the Directors may determine for every certificate after the first.	
	i.	The certificates of title of shares and duplicates thereof when necessary shall be issued under the Seal and shall specify the share or shares held by a Member and the amount paid thereon including in particular and without limitation such legends as the Company shall be obligated to affix to certain of the certificates by law or as the Company shall have agreed to affix pursuant to any contractual, arrangements entered into by the Company in this respect.	·
•	22.	The Company shall not be bound to issue more than one share certificate in respect of a share or shares held jointly by two or more persons, and delivery of a certificate for a share to any one of joint holders shall be sufficient delivery to all.	
	23,	The Company shall, within ninety days, after the allotment of any of its shares, and within 45 days after the date on which the instrument of transfer has been lodged, complete and have ready for delivery the certificates of all shares, allotted or transferred, unless the conditions of issue of the	
		shares otherwise provide.	- }
		Page 8	Į
			}

and a subscription of the strength of the second second second second second second second second second second

8.5.7.5.5.6.9**5** 

. . .

N. C. Contractor

----

MICS T

rights the state of the state



SY.		· · · ·	
	TAPAL		
		(b) the transfer is in breach of the Agreement or any other contractual arrangements entered into by the Company; or	, Ç
		(c) the transfer is in violation of these Articles.	•
· ·	31.	If the Company refuses to register the transfer of any shares, owing to any attachment or prohibi- tory order of a competent authority or otherwise, the Company shall within forty five (45) days after the date on which the instrument of transfer was lodged with it, send notice to the transferee notice of the refusal indicating reasons for such refusal. No transfer of shares in any case shall be made to a minor or an insolvent or person of unsound mind.	
	.32. i	The transfer of shares shall be effected by an instrument in writing in the usual common form modified so as to suit the circumstances of the parties and shall be executed both by the transferor and the transferee and duty stamped according to law, and execution be attested by at least two mate witnesses, who shall add their address and occupation, and the transferor shall be deemed to remain the holder of such shares until the name of transferee shall have been entered in the Register of Members in respect hereof.	· 
,	-33,	Every instrument of transfer shall be left at the office for registration, duly stamped and shall be accompanied by the certificate of the shares to be transferred and such other evidence as the Company may require to prove the tills of the transferror or his right to transfer the shares. All instruments of transfer which will be registered shall be relained by the Company, but any instru- ment of transfer which the Directors decline to register shall, on demand, be returned to the person	
		depositing the same.	•
	•	Where It is proved to the satisfaction of the Directors that an Instrument of transfer signed by the transferor and the transferae has been lost, the Company may, if the Directors shall think fit, by an application in writing made by the transferee and bearing the stamp required by an instrument of transfer, register the transfer on such terms as to indemnity or otherwise as the Directors may think fit.	ت
••		Nothing contained in these Articles shall prejudice any power of the Company to register as Mem- ber any person to whom the right to any shares of the Company has been transmitted by operation of law.	· · · · · · · · · · · · · · · · · · ·
	36.	No fee will be charged for registering transfer of shares.	
•		The transfer books and Register of Members may be closed for any time of lines not exceeding in the whole forty five days in each year, but not exceeding thirty days at a time, in accordance with the manner specified in Section 151 of the Ordinance.	
		The nominees of a deceased Member as specified in Section 80 of the Ordinance, or executors or administrators of a deceased Member shall be the only persons recognised by the Company as having title to his share except in case of joint holders in which case the subving holders or the executors or administrators of the last surviving holder shall be the only persons interview of the last surviving holder shall be the estate of a deceased joint holder from any liability in respect of any share jointly held by him. The Company shall not be bound to recog- nise a nominee of a deceased Member or executor or administrator unless to shall have obtained probate or letters of administration, succession centificates or other tegal representation, as the case may be, from a court of competent jurisdic ion Provided nevertheless that where the Directors	-
	•	Page 10	
•	•		

Ē



· · · · ·		
		a O
	The Members entitled to receive notice of some particular meeting (other than the Annual General Meeting) at which a special resolution is to be passed that meeting may be convened by such shorter notice and in such manner as the Members may deem fit.	Chul
	45. The accidental omission to give notice of a meeting to or the non-receipt of notice of a meeting by, any Member or person entitled to receive notice shall not invalidate the proceedings at any general meeting.	•
	PROCEEDINGS AT GENERAL MEETINGS	्रम् अस् अस् अस्ति स्वा
· ·	46. The ordinary business of the Company at an Annual General Meeting shall be to receive and consider the balance sheet and profit and loss account, the reports of the Directors and of the auditors, to elect Directors, to declare dividends and to appoint auditors, and fix their remuneration. All other business transacted at an Annual General Meeting and all business transacted at Extraordinary General Meetings, shall be deemed special.	• • • • • • • • • • • • • • • • • • •
	47. No business shall be transacted at any general meeting unless a quorym of Members is present at the time when the meeting proceeds to business. Subject to the provisions of the Ordinance, three Members present personally who represent not less than twenty-five percent of the total voting power, either of their own account or as proxies, shall be a quorum.	
	48. If within half an hour from the time appointed for a general meeting, a quorum is not present, the meeting if called upon the requisition of Members shall be dissolved; in any other case it shall stand adjourned to the same day in the next week at the same time and place, and, at such adjourned meeting if a quorum is not present within half an hour from the time appointed for such meeting the Members present being not less than three shall constitute a quorum and may transact the business for which the meeting was called.	• • • • • • • • • • • • • • • • • • •
•••• <u>•</u>	49. The Chairman, if any, of the Board shall preside as Chairman at every general meeting of the Company, but if there is no such Chairman, or if he shall not be present within fifteen minutes after	
	The time appointed for the holding of the meeting or is triwilling to act, the Members present shalt elect one of the Directors present to be Chairman of the meeting, or if no Directors be present or if Directors present decline to take the chair, the Members present shall choose one of their member to be Chairman of the meeting, Every Director of the Company shall have the right to attend any general meeting of the Company and also to take part in the discussion thereat.	nanon annan an Antar gaolana sa an Antar gaolana sa an Antar
	<ul> <li>50. The Chairman may with the consent of any meeting at which a quorum is present (and shall, if so directed by the meeting), adjourn the meeting from time to time and from place to place, but no business shall be transacted at any adjourned meeting other than the business left until shed at the meeting from which the adjournment took place. When a meeting is adjourned for thirty days or more notice of adjourned meeting shall be given as in the case of dig(natmeeting. Save as afore-satd, it shall not be necessary to give any notice of an adjournment or of the business to be transacted at an adjourned meeting.</li> <li>51. At a general meeting a resolution put to the vote of the meeting shall be given as in the meeting shall be given as a former and provide any adjourned of the meeting.</li> </ul>	
	hands, unless a poil is (before or on the declaration of the show of hands) demanded in accordance with provisions of Section 167 of the Ordinance.	
norrado sarra arciana canta fu	a) by the Chairman of the meeting of his own motion; or	
· · · · · · · · · · · · · · · · · · ·	$P_{a00}$ (2)	
	a a bara a bara a bara da bara da bara da bara a a san bara da bara da a bara da a bara da bara da bara da bara Bara da bara da	

. ?

• •

,	1		·		
	). :	1	,		(S)
j.	α,		مې ور د ور د د د د د مې ور د	TAPAL	9
	b	) by one Member having the right to If not more than seven such Mem present in person or by proxy if mo	bers are personally present, and I	by Iwo such Members	ı I
	. C]	) by any Member or Members prese tenth of the total voting power in re		ng not less than one-	•
	ď	) by any Member or Members prese pany conferring a right to vole on t has been paid-up which is not less	the resolution, being shares on whi	ich an aggregate sum	
	· ·	ring that right.	± (		
	a to si	Inless a poll is so demanded, a declaration show of hands been carried of carried una o that effect in the book containing the min ive evidence of the fact without proof of the r against that resolution.	animously or by a particular majorit utes of the proceedings of the Corr	y, or lost, and an entry pany shall be conclu-	
	ta Is	If decisions required to be taken by the sken at a meeting and shall require the sound and paid-up share capital of the C reater percentage affirmative vote as set	affirmative vote of Members hold company, except in the case of a	ling a majority of the	
		ny poli duly demanded on the election of rent shall be taken at the meeting and wit		y question of adjourn-	,
-	0	a poll is duly demanded, it shall be taken I the Ordinance and the results of the pol high the poll was demanded.			
	Ď	he demand of a poll shall not prevent the usiness other than the question on which t e withdrawn at any time by the person or	he poll has been demanded. The c		
	m	he Chairman of any meeting shall be the neetings. The Chairman present at the tim I every vote tendered at such poll.	sole judge of the validity of every he of taking of a poll shall be the sol	vote tendered at such le judge of the validity	1
	ie	iolwithstanding anything contained in thes east eighty-five percent (85%) of all issue ecision regarding any of the following ma	d and paid up shares shall be requ	e Members holding at irred with respect to a	
	(i	) the issuance of, and the interest of provisions relating to the issuance Capital to any Person not a Memb	e of, any New Capital or the sale	hs and other financial or issuance of New	]
	(I	i) any merger or consolidation of the the Company into a joint venture w		erson or any ontry by	
	( <sup>1</sup>	<li>ii) the filing by the Company of any v Company;</li>	voluntary petition for the winding-U	p of dissolution of the	
	•	P	age 13.	+ No HEST	·
۲ ۲					•
		•			

ì.,

.

, in the second s	•				0 01
		TAPA	и сү	a A second second second second second second second second second second second second second second second seco	
	ĺ		(iv)	any amendment to the Memorandum and Anticles of Association;	SI
		•	(V), '	the Incurring of Indebtedness on the part of the Company of the provision of guarantees by the Company in an aggregate outstanding amount at any time of US\$500,000 or more (excluding the Existing Loan and any relinancing thereof as contemplated by the Term . Sheet and Working Capital;	
		ř	(vi)	the approval of all material agreements (Ir cluding any amendments thereto) entered or to be entered into by the Company after the date of the Agreement, including any Financing Document, any other agreement with respirat to the financing arrangements for the Project and any additional Project Document other than any document entered or to be entered	
	:			Into by the Company for the refinancing of the Existing Loan as contemplated by the Term Sheet :	
			(Vii)	any change in the number of Directors comprising the Board;	
	 		(vili)	any change in the status of the Company or other reorganisation of the Company or its assets;	1
		, <b>t</b>	([x)	approval of any plan of refinancing the Existing Loan other than the refinancing contemplated by the Term Sheet;	
			(×)	the requirement for any support from the Shareholders, other than for Working Capital and the refinancing of the Existing Loan as contemplated by the Term Sheet; and	and and a second s
		بودهه مراهه	(xi)	the transaction of any business by the Company with a Member or any Affiliate of a Mem- ber, or the causing of the Company to enter into any contract or other agreement with a	
· · · · · · · · · · · · · · ·		• • •	یک ایک ایک ایک رو	Member, an Affiliate of a Member of any employee, officer or director of a Member.	
لى تو تون بون بالله و تو		58.	On a si	ow of hands every Member who, being an individual, is present in person or by proxy or	and and a same of the state of the
		90,	being a have of	body comporate is present by a representative duly appointed pursuant to Anticle 69 shall e vote except in the case of an election of Directors in which case the provisions of Section ne Ordinance shall apply. On a poll every Member shall have voting rights as laid down in	
		4 <i>1</i> 1 4	Section	160 of the Ordinance. Nothing contained in these Articles shall prevent a Member from the case of an individual himself, and in the case of a body corporate itself by contract.	
			Will any	other Member or Members to vote his or its shares or other voting securities in any particu-	an ta gana anna an an an an an an an an an an a
	r taleque rea	11.35	the rem meeting	oval of Directors or for the removal of the Chief Execulive or otherwise at any general of the Company.	
·			shall be	ase of joint holders the vote of the sellior Member present whether in person or by proxy accepted to the exclusion of the votes of the other joint holders, and for this purpose shall be determined by the order in which their names stand in the Register.	
		60;	A.meml	er of unsound mind, or in respect of whom an order has been made by any court having on in lunacy, may vote, whether on a show of hands or on a poll, by his committee or other ==	
		22 543 	legal gu	ardian and any such committee or guardian may ion a poll, vote by proxy 57	
ى ئىرىيى بىلەت يەرىكە بەك يەك يەك ئەك ئەك يەك يەرىك يەرىكە بەرىك يەك يەك يەك يەك يەك يەك يەك يەك يەك يە	na ay wataka	in a sain a sain a sain - -	ណ្ដីជាភ្លេង ខេត្តស្ថា ខ្ម	Pege 14.	n de la constant de constant de la constant de la constant de la constant de la constant de la constant de la c La constant de la const
				A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF A CONTRACTOR OF	
		<u>.</u>			A CONTRACTOR
	ing the second second second second second second second second second second second second second second secon				
$ \begin{array}{c} & \left( \sum_{i=1}^{n} \frac{1}{2} \sum_{i=1}^{n} \frac$	an an an an an an an an an an an an an a	e tite ten i Vite of the second		an san bulan pulan san bulan kana kana kana kana kana kana kana k	
ang tana bagi pang tang tang tang tang tang tang tang t			والمديوت الراقسة	anaa darada ayaa gayaa ay ay ay ahaaya ay ahaaya ay ahaa ay ahaa ay ahaa ay ahaa ahaa ahaa ahaa ahaa ahaa ahaa Ahaa ahaa a	ى بى ئەرىپىيە بىرىكى بىرىكى بىرىكى بىلىكى r>بىلىكى بىلىكى br>بىلىكى بىلىكى
	ر به در کرد. در این این این این این در این این این این این این این	and a start of the second second second second second second second second second second second second second s		1997 - San Barnel Barnel Marine Constanting and State State State State State State State State State State Sta State State	
	si - sisi Alay ka				
	در بار آراد می می در اور از مربق می مربق در ایک در ایم	ada interna Maria de la composición Maria de la composición		가 가장 가지 않는 것은 것을 하는 것이 있는 것이 가장 물질을 가지 않는 것이 있다. 것이 있는 것이 있는 것이 있다. 같이 이렇게 같은 것은 것이 같은 것은 것이 같은 것은 것이 있을 것을 것이 하는 것이 같이 있다. 것이 같은 것이 같은 것이 같은 것이 같이 같은 것이 같이 같이 같이 같이 같이 같이 있다. 것이 한	
				에는 가장에 가장에 가장에 있는 것을 가장해 있다. 가장에 가장에 가장하는 것은 것을 가장하는 것을 수가 있다. 이렇게 가장하는 것을 가장하는 것을 가장하는 것을 가장하는 것을 가장하는 것을 것을 수가 없다. 이렇게 가장하는 것을 것을 수가 없다. 이는 것을 것을 수가 없다. 이는 것을 것을 가장하는 것을 가장하는 것을 가장하는 것을 것을 것을 것을 것을 것을 것을 것을 것을 수가 없다. 이는 것을 것을 것을 것을 것을 것을 것을 것을 것을 것을 수가 없다. 이는 것을 것을 것을 것을 것을 것을 것을 것을 것을 것을 것을 것을 것을	
	n ann an Annaich An Airte Airte Annaich An Annaichte Annaichte				
		· i		میشد. دو اگریا به برای در این ۲۰ در در در با برای در با در تشکیل در میشود و در این این میدود می با با کا مستقل از این از این این این این این این این این این این	
	د ا میں میوند رود چر چر				

F

.

. . . . .

. 19-

61. No Member shall be entitled to vote at any general meeting unless all sums presently payable by him in respect of shares in the Company have been paid. No objection shall be raised to the qualification of any voter except at the meeting or adjourned meeting at which the vote objected to is given or tendered, and every vote not disallowed at such 62. meeting shall be valid for all purposes. Any such objection made in due time shall be referred to the Chairman of the meeting, whose decision shall be final and conclusive. N. March Mill On a poll, voles may be given either personally or by proxy, or, in the case of a body corporate by a 63, representative duly authorised in accordance with Article 69. The instrument appointing a proxy shall be in writing under the hand of the appointer or of his 64. attorney duly authorized in writing, or if the appointer is a body corporate, under its common seal or the hand of an officer or altorney duly authorized by it and in default the instrument of proxy shall not be duly authorized. A proxy need not be a Member. A proxy who is appointed for a specified meet ing only shall be called a Special Proxy. Any other proxy shall be called a General Proxy. A Special Proxy shall be valid only for the meeting to which it relates and it may not be used for more than one meeting. 65. The instrument appointing a proxy and the power of attorney or other authority (if any) under which it is signed or a notarially certified copy of that power or authority, shall be deposited at the Office not less than fony-eight hours before the time for holding the meeting at which the person named in the Instrument proposes to vote, and in default the instrument of proxy or the power of altorney or other authority (if any) under which it is signed or a notarially certified copy of that power or authority shall not be treated as valid. 66. An instrument appointing a proxy may be in the following form, or in any other form which the Directors shall approve -I, the undersigned ...... being a member of TAPAL ENERGY (PRIVATE) LIMITED hereby appoint Mr./Mrs./Miss to vote for me and on my behalf at the Annual or Extraordinary (as the case 1.25 may be) General Meeting of the Company to be held on the ..... day of ...... and all every adjournment thereof (or all every general meeting of the " ....... Company to be held before the ..... day of ..... and at every adjournment of Astering, any such meeting). W. 0.00 . .... . .. Signed this ..... day of ..... The instrument appointing a proxy shall be deemed to confer authority to demand or join mending a poil. in de 67, manding a poll. 1. 11. a . . ۰. A vote given in accordance with the terms of an instrument of proxy shall be valid notwithstanding 68. the previous death or insanity of the principal or revocation of the proxy or of the authority under Page 15

		- F
(28)		
		0,
TAPAL	/ //	یا " تر
•	which the proxy was executed, or the transfer of the shares in respect of which the proxy is given, provided that no infimation in writing of such death, insanity, revocation or transfer as aforesaid shall have been received by the Company at the Office before the commencement of the meeting or adjourned meeting at which the proxy is used.	Ç.
• 69 • · ·	Subject to the provisions of Section 162 of the Ordinance, a body corporate which is a Member may, by resolution of its directors or in such manner as may be permitted or required by its constitution, authorise in writing any person to act as its representative at any meeting of the Company and the person so authorised shall be entitled to exercise the same powers on behalf of the body corporate which he represents as that body corporate could exercise if it were an individual Member of the Company present in person. A body corporate attending a meeting through such representative shall be deemed for all purposes of these Articles to be a Member present in person at the meeting.	
-	DIRECTORS	
	The number of Directors to be elected shall be fixed, pursuant to the provisions of Section 178 of the Ordinance.	
, żi.	Article 71 deleted and replaced with the Article 71 reproduced at the bottom of this page are	:
72	The first Directors of the Company shall be appointed by the Subscribers.	
73	Save as provided in Section 187 of the Ordinance no person shall be appointed as a Director unless he is a Member.	
74.	The first Directors shall stand retired at the first annual general meeting, and directors shall be elected in their place in accordance with Article 77 hereof.	
75.	Any person who seeks to contest an election to the office of Director shall, whether he is a retiring Director or otherwise, file with the Company, not later than fourteen days before the date of the meeting at which elections are to be held, a notice of his intention to offer himself for election as a Director. Provided that any such person may, at any time, before the holding of elections withdraw	
	such notice.	
76.	Reliring Directors shall be eligible for re-election.	en aler
77. ;	The Directors shall be elected in accordance with the provisions of the Ordinance by the Members In General Meeting from amongst the candidates eligible for election in the following manner (	
,	<ul> <li>every Member present in person or by privy or by representative shall have such number of votes as is equal to the product of the number of voting shares held by him and the number of Directors to be elected;</li> </ul>	<i>y</i>
:	b) the number of votes calculated in accordance with the preceding clause (a) may be given to a single candidate or may be divided between any two or more candidates in such manner as the person voting may choose; and	*
-	The second second second second second second second second second second second second second second second se	
	Page 16	<b>}</b>
71.	Unless otherwise determined by the Company in General Meeting in accordance with the provisions, of Article 98(b), the Board shall consist:	/
<u>(; )</u> ]	(i) ten (10) elected directors including a Chlef Executive, if the Chlef Executive is nominated from one of the elected directors; or	
	<ul> <li>(ii) ten (10) elected directors plus a Chief Executive; if the Chief Executive is not nominated from one of the elected directors.</li> </ul>	

\*\*\*\*\*\*\*

Ħ

873) 1

the candidate who gets the highest number of votes shall be declared elected as Director C) W. and then the candidate who gets the next highest number of votes shall be so declared and so on until the total number of Directors to be elected has been so elected. The Company may at any time, by a resolution in general meeting remove a director, appointed 78. under Section 176 or Section 180 or elected in the manner provided for in Section 178 of the Ordinance: \ a È Provided that a resolution for removing a director shall not be deamed to have been passed unless the number of votes cast in favour of such a resolution is not less than:-- <u>1</u> 0 <sup>(2</sup> the minimum number of votes that were cast for the election of a director in the manner provided in subsection (5) of Section 178 of the Ordinance; of - Q. .... the total number of votes for the time being computed in the manner laid down in subil) Section (5) of Section 178 of the Ordinance divided by the number of directors for the time being, If the resolution relates to removal of director appointed under Sections 176 or 180 nesti Noti of the Ordinance. .: 7 -Retiring Directors shall continue to perform thair functions until their successors are elected. 79. A Director elected by the Members in general meeting shall hold office for a period of three years 80. following the date from which his election is effective unless he earlier resigns, becomes disqualifled from being a Director or otherwise ceases to hold office. 🕷 斗 🕬 In the event a vacancy occurs on the Board by reason of the death, resignation or removal of a 81. • director, such vacancy shall be filled by the vole of the remaining Directors if still constituting a quorum, or if no quorum exists, by the Members at a general meeting called for that purpose. A director elected or eppointed to fill a vacancy caused by the death, resignation or removal of a director shall be elected or appointed only for the unexpired term of such director's predecessor in office. When any Director intends to be, or is absent for a period of not less than three (3) Months from 82 Pakistan, he may with the approval of the Directors appoint any person to be his alternate director. and such alternate director during the absence of the appointer from Pakistan, shall be entitled to receive notice of and to attend and vote at meeting of Directors and shall be subject to and entitled to the benefit of the provisions contained in these Articles with reference to directors and may exercise and perform all such powers, directions and duties as his appointer could have exercised or performed including the power of appointing another alternate director. An alternate directorso appointed shall not be entitled to receive any remuneration from the Company not be required to hold any qualification. Such appointment shall be recorded in the Directors minute book: A Director. may at any time by notice in writing to the Company remove an alternate director appointed by him upon the return of the appointer to Pakistan, or the death of, or the retirement or resignation as Director of the appointer, the alternate director shall cease to be such provided that if any Director retires but is re-elected at the meeting at which such retirement took effect any appointment made by him pursuant to this Article which was in force immediately prior to this retirement and re-election and which has not otherwise ceased to be effective shall continue to operate after his rejection as If he had not so retired. An alternate director shall not be deemed to be the agant of the Director appointing him but shall be reckoned as one with his appointer. All appointments and removals of alternate directors shall be effected by writing under the hand of the Director making or revoking Page 17
such appointment and left at the office. For the purpose of assessing a quorum in accordance with the provisions of Article 97 hereof an alternativ director shall be deemed to be Director. Any Director may act an

Alternate director for any one or more Directors, as well as being able to act as a Director in his own right. An allemate director may resion as such upon giving thirty (30) days prior netice to the Board to this effect. An alternate director need not be a Member.

The Directors shall be paid such travelling and hotel expenses as may be fixed by the Directors from time to time or if a Director has to come to attend the Board meeting from outstation.

83.

. 64,

85.

86

88.

89.

٩.,

.

-12 (F

The Directors shall elect one of their member as the Chairman of the Board and the Chairman shall nol have a second or casting vote.

The Directors may from time to time delegate any of their powers to a committee or committees consisting of two (2) or more members of their body as they think fit. A committee's duties and powers shall be restricted to the purpose for which it is formed. Any committee so formed shall' conform to any regulations that may by imposed upon it by the Directors and shall be governed, in the exercise of the powers so delegated, by the provisions herein contained for regulating meetings and proceedings applicable to the Directors.

#### POWERS AND DUTIES OF DIRECTORS

The business of the Company shall be managed by the Directors, who may pay all expenses incurred in setting up and registering the Company, and may exercise all such powers of the Company as are not by the Ordinance or by these Anticles, required to be exercised by the Company in general meeting provided that no regulation made by the Company in general meeting shall invalidate any prior act of the Directors which would have been valid if that regulation had not been made.

87: Subject to the provisions of Articles 57 and \$8 the Directors may exercise all the powers of the Company to borrow money and to mortgage its undertaking, property and capital or any part thereof, and to issue securities and debeniures, participation or term finance certificates or any other instru-ment, whether as security for any debt, liability or obligation of the Company or of any third party or otherwise. othenvise

The Directors may from time to time and at any time by power of attorney appoint any company, firm or person or body of persons, whether nomina ed directly or indirectly by the Directors, to be the attorney or attorneys of the company for such purposes and with such powers, authorities and discretions (not exceeding those vested in or exercisable by the Directors under these Articles) for such period subject to such conditions, if any, as they may think fit, and any such powers of attorney may contain such provisions for the protection and convenience of persons dealing with any such attorney to delegate all or any of the powers, authorities, and discretion vested in him.

A Director of the Company or a firm of which such Director is a partner or private company of which such Director is a director may with the consent of the Company in general meeting hold any office A Reguliano of profit in the Company.

5. Subject to the provisions of Section 195 of the Ordinance, the Directors Shall not be disqualified from contracting with the Company either as vendor, purchaser, or otherwise inbeshall any such contract or agreement entered into by or on behalf of the company or partnership of or provide any such of Director of the Company shall be a member or otherwise interested be avoided nor shall any such of Director so contracting or being such member or so interested be liable to account to the Company such any such or so interested be liable to account to the Company such member or so interested be liable to account to the Company set. 90. Page 18

को देशी हैं। देशी में ने में की साम की से के लोग की स

2.

·							
	· · · .	<i>•</i> •		1		(A)	
Ň				•		$\nabla$	
-	36			,		-	
* *	u,				TAPAL ENERG		
T.	W.			t or arrangement by reason of s labilshed, but the nature of his i		-	
				hich the contract or errangement the first meeting of the Directo			
	1	the interest. A ge	neral notice that any Dir	ector of the Company is a dire ed firm and is to be regarded as	ctor or a member of any		
	i i	quent transaction	with such company or f	irm shall, as regards any such such such such general notice it shall not	transaction, be sufficient		
		special notice rel	aling to any particular tr	ansaction with such firm or co led or concerned in a contract	mpany. In the case of a		
	·• • • •	chief executive, n	nanaging agent, whole li	me director, or secretary of the bserved and performed:			
	•			on 219 of the Ordinance a Reg	ister shall be kept by the		
		Directors in which	shall be entered particu	lars of all contracts or arranger Office during business hours.	nents and which shall be	. ·	
ĵ 				of exchange and other negoti	able instruments, and al		
- <mark>-</mark> -		receipts for money	is paid to the Company, s	hall be signed, drawn, accepted iner as the Directors shall from t	i, endorsed, or otherwise		
1. A A A A A A A A A A A A A A A A A A A		determine.		alanda ala mangana dina man	9. «7.		
				provisions of the Ordinance a the particulars of mortgages a			
	- 6 	properly of the Co	ompany or created by it,	and to keep a Register of the I I Members and a summary of p	Directors and managers		
د. مربع ،	s k . afit suu	and notice of any	consolidation or increase	of share capital and copies of olification of any changes there	special resolutions and a	· •~	
	• •	计记录 化化合物化	专人的名 就是这些问题的	ade in books provided for the p		<u></u>	
ne gag und gely distance during englishing with		and a start of the start of the start of the start of the start of the start of the start of the start of the s	sintments of officers mad	( Charles and the second of the	anna garan ang dalaman na baran na baran na baran na baran sa baran sa baran sa baran sa baran sa baran sa bar		y dydd ynyrholyff ry f yr Sodi Challennydyr (Pri y 2010au
	1 1 1			ent at each meeting of the Direc	ctors and of any commit		
	T.		Directors;			1 N	
		<ul> <li>of commit</li> </ul>	tee of Directors; and eve	of all meetings of the Company, any Director present at any mee	ling of Directors or com		
		minule of	such a meeting if purple	ame in a book to be kept for th orting to be signed by the Cha meeting of the same body, sha	irman thereof, or by the		
		without ar	of the flext succeeding w further proof of the fac	Is therein stated.			
	С. 1-		VACATION OF OF	FICE OF DIRECTORS			
: ا بنا اختا مستحمد المحمد المحمد الم		95. A Director shall ip	so facto cease to hold of		DEAD - S	· · · · · · · · · · · · · · · · · · ·	
		a) he is foun	d to be of unsound mind	by a court of competent jurisdle	allon lor Dialla		
			dged an insolvent, or				
22.52.4 <u>5.9</u> -39-32.0000000-0000-0000-0000-00000	ija, Prezistaj organizacija u pristana pri prista pri prista pri prista pri prista pri pri pri pri pri pri pris Na na	ระบบเขาะระบุรัง ให้สุดใจสมบัตรสมใจใหม่ไปที่สุดให้เรา 		n nam ng Ali sa s	endelska svensk testin bivelopatersko ogsatelsk	anen fiktor etterhete uppt of de	andrika in the state of the state of the state of the state of the state of the state of the state of the state
na an an an an an an an an an an an an a				198 19			
		•					
	n 1997 - Anna Star 1997 - Anna Star	• •					
					<u>wa ing ng kang kang ka</u>	<u></u>	
							· · · -
	-						
- Naga i tanan ing tuput a anna ing	and the second of				· ·		

----

. ....

----

s'

; ۴

C)

d)

e)

, f)

g)

h)

h

))

96

97.

98.

na the state of the second second second second second second second second second second second second second

(8)

he ceases to be a Member of the Company save as provided in Section 187(h) of the Ordinance, or

he or any firm of which he is a partner or any private company of which he is a director without the sanction of the company in general meeting accepts or holds any office of profit under the Company other than that of Chief Executive or a legal or technical advisor or banker, or

he absents himself from three consecutive meetings of the Directors or from all meetings of the Directors for a continuous period of three Months whichever is the longer without leave of absence from the Board, or

he acts in contravention of Section 195 of the Ordinance, or

he resigns his office by notice in writing to the Company, or

he suffers from any of the disabilities or discualifications mentioned in Section 187 of the Ordinance, or

he has been convicted by a court of competent jurisdiction for an offence involving moral urpitude or

he has betrayed lack of fiduciary behaviour and a declaration to this effect has been made by the Court under Section 217 of the Ordinance 🔬

#### **PROCEEDINGS OF DIRECTORS**

The Directors may meet logether for the despatch of business, adjourn and otherwise regulate their meetings, as they think fit. The Board of Directors shall meet at least four limes in every year to meetings, as they think it. The Board of Directors shall meet at least four times in every year to conduct the business of the Company. A Director may, and the Secretary on the requisition of Director shall, at any time, summon a meeting of Directors. No meeting of the Board of Directors may be held unless not less than seven (7) days hotice has been given to each Director on the Board of Directors; provided however that such meeting may be held upon shorter notice if such notice is waived in writing by all of the Directors. If and to the extent permitted by law, meetings of the Board of Directors may be held by video conference call, provided a quorum of Directors is present of such call and such directors can hear and see each other and are able to satisfy them-selves that the directors present at such meetings remain present for the duration thereof.

The quorum necessary for the transaction of the business of the Directors shall be five Directors actually present in person of by an alternate director, and at least one of whom shall have been nominated by each Group entitled to nominate a director pursuant to the Agreement For the purposes of this Article, an alternate director appointed by a Director shall be counted in a quorum at a meeting at which the Director appointing him is not present.

Except as set forth in Article 98(b) decisions and resolutions of the Board shall be taken by not less than two-thirds of the Directors present a meeting at which there is a quorufy No resolution, concerning any malter listed below when put to vote, shall be deened to be carried except when voted in favour of such resolution by at least saved Directors on the Board. Board:

and the second states of the second

1. Sec

Page 20

the issuance of, and the interest rate, terms, subordination provisions and other financial provisions relating to the issuance of New Capital, or the sale or issuance of New (i) Capital to any Person not a Member price to such sale; including any decision that New Capital shall be contributed as equity rather than Shareholder Loans; any decision regarding the declaration or payment of distributions to the Shareholders (and (ii) the amount thereof) whether in the form of dividends (including intertim dividends) or return of capital, including determination of Net Cash Flow; any decision to sell, pledge, lease, assign, mortgage or create a Lien on or a security interest in any material asset of the Company other than for purposes of the refinancing of (iiii) Interest in any material asset of the Company states the Existing Loan as contemplated by the Term Street; any decision to lend Company lunds to or invest Company lunds in any other company partnership, joint venture or other Person (other than funds maintained and invested by the Company under and in accordance with the Pinancing Documents); (iv) to the extent not otherwise determined by the Agreement or applicable law, any decision which could reasonably be expected to have the effect of a material change in (A) any of the Company's depreciation or accounting methods, tax policies or (ax elections, or (B) the (v) lax status of the Company or any Member; any change in the Company's Fiscal Year; (vi) any termination of, any material amendment or modification to or any walver or election of (vfi) any right under any Project Document, any Financing Document or any other agreement to which the Company is a party if the same could reasonably be expected to have a material adverse effect on the Company or would require the Company to make payments as a result thereof in an aggregate amount in excess of US\$500,000; any decision to cause the Company to enter into or conduct any business other than the (vilii) Project and activities reasonably related thereto; (IX) any increase in or volunlary prepayment, modification or extension of any indebtedness of the Company (other than for Working Capital and the refinancing of the Existing Loan as contemplated by the Term Sheet) in an aggregate amount outstanding at any time in excess of US\$500,000; the initiation, release, settlement or compromise by the Company of any arbitration, litiga-tion or other legal or administrative proceeding, or the settlement or compromise by the Company of any threatened claim, that in either case could have a material adverse effect on the Company or that could require its part of such settlement or litigation, payments by the Company (other than payments covered by insurance) that an emount in excess of US\$500,000; (X) the transaction of any business by the Company with a Member of any Affiliate of a Mem-ber, or the causing of the Company to enter into any contract or other agreement with a Member, an Affiliate of a Member or any employee, officer of director of a Member. (xl) the approval of each annual budget for the Company and any changes thereto in exc (xil) 115% of the budgeted amount contained in the relevant approved annual budget ÷ Page 21

(Zzz)		· .
		C
	(xIII) the approval of all material agreements (ir oluding any amendments thereto) entered or to be entered into by the Company after the date of the Agreement including any Finanoing Document, any other agreement with respect to the financing or refinancing arrangements for the Project, any additional Project Document and any amendment or modification to a Project Document or a Financing Document other them any document entered or to be entered into by the Company for the purposes of the refinancing of the Existing Loan as contemplated by the Term Sheet;	Ç.
	(xiv) the approval of any expenditure not contained in an approved budget in excess of US\$600,000;	
	(w) +: any change in the number of Olivectors comprising the Board;	•
	(xvi) the appointment of the Chief Executive;	• •
: A             	(Will a procedures for meetings of a protocol with respect to the place, time and procedures for meetings of the Board of Directors (and any replacement or amendment of such protocol); and	
- 	(Xvili) the requirement for any support from the Shereholders, other than for Working Gapital and the removed of the Existing Loan as contemplated by the Term Sheet.	
196 20 20	where the set of the s	
10	All acts done at any meeting of the Directors of by any person acting as a Director shall notwith- standing that it shall afterwards be discovered that there was some defect in the appointment or continuance in office of any such Director or person acting as aforesaid, or that they or any of them were disqualified or had vecalad office, or were not entitled to vote, be as valid as if every such person had been duly appointed of had duly continued in office and qualified and had continued to be a Director and had been entitled to be a Director.	<b>*</b> 4
10	Except as otherwise provided in the Ordinanct, any action which may be taken at a meeting of the Board shall be validly taken without such meeting if a resolution in writing approving such action is	
· · .	executed by the Directors that would otherwise comprise a quorum under Article 97 or. If Article 98(b) is applicable, the number of Directors regulared to take such action pursuant to Article 98(b). For this purpose, it shall be permissible to circulate the text of the proposed resolution duly signed by the Chairman, if any, of the Board or the Chief Executive and obtain the signatures of all the other Directors thereon separately by fax (the signed original whereof shall be sent in due course by mail, or courier by the Company for its record) and such resolution shall be effective as soon as the text of the resolution signed by each of the other Directors shall have been faxed to and face by the Company.	
- 10	If at any meeting the Cheirman is absent, the Directors present may elect any one of their number/	
, -		/ -
1 10	The Board shall within fourdeen days after the constitution of the Board or from the date of election	•

Ħ

4

100

urger.

e ji

------

1990-1991 1997 - 1997

Vorterna A

of Directors or the office of the Chief Executive failing vacant as the case may be, appoint, subject Page 22,



a

1 76.	_ ر	
	INAPA TENER	). d
	109,	The Directors may from time to time pay to the Members such Interim dividend as appear to the Directors to be justified by the profits of the Company.
:	110,	The Directors may, before recommending any dividends, set aside out of the profits of the Company, such sums as they think proper as a reserve or reserves, which shall, at the discre- tion of the Directors, be applicable for meeting contingencies, or for equalizing dividends, or for any other purpose to which the profits of the Company may be properly applied, and pending such application may, at the like discretion, either be employed in the business of the Company , or be invested in such investments (other than shares of the Company) as the Directors may from time to time think fit:
	111.	When any shareholder is indebted to the Company, all dividends payable to him, or a sufficient part thereof, may be retained and applied by the Directors in or towards satisfaction of the debt.
· ·	112.	Any dividend, interest or other moneys payable in cash in respect of shares may be paid by cheque or warrani sent through the post direct to the registered address of the holder or, in the case of joint holders, to the registered address of that one of the joint holders who is first named on the Register or to such persons and to such address as the holder or joint holders may in writing direct. Every such cheque or warrant shall be made payable to the order of the person to whom it is sent. Any of two or more joint holders may give effectual receipts for any dividends, bonuses, or other moneys payable in respect of the shares held by them as joint holders. The dividend shall be paid within the period laid down in the Ordinance.
	113.	Unpaid dividends shall not bear interest as against the Company.
		CAPITALIZATION OF PROFITS
	114.	The Company in general meeting may upon the recommendation of the Directors resolve that it is desirable to capitalize any part of the amount for the time being standing at the credit of any of the Company's reserve accounts or to the credit of the profit and loss accounts or otherwise available for distribution and accordingly that such sum he set free for distribution amongst the Members who would be entitled thereto if distributed by way of dividend and in the same proportion on condition that the same be not paid in cash but be applied either in or towards paying up any amounts for the time being unpeld on any shares held by such Members respectively or paying up in full un-issued shares of the Company-to be allotted and distributed/credited as fully paid up to and amongst such Members in the proportion aforesaid, or partly in the other, and the Directors shall give effect to such resolution.
		ACCOUNTS
	115.	The Directors shall cause the Company to keep proper books of account as required under Section. 230 of the Ordinance and in accordance with sound and generally accepted accounting principles and corporate practices.
· ·	116.	The books of account shall be kept at the Office of the Company or at such other place as the Directors shall think fit and shall be open to inspection by the Directors during business hours
•	117.	The Directors shall from time to time determine whether and to what extent and all what time and is places and under what conditions or on regulations that accounts and books or papers of the company or any of them shall be open to the inspection of Members not being Directors, and no Member (not being a Director) shall have any right of inspecting any account and book or papers of
	C	Page 24
•		

P

16 (25 )		•	
			7
	ø		$\gamma$
1.4 6			/
[ ]		TAPAL	
ß		ENERGY ENERGY	
		the Company except as conferred by law or authorised by the Directors or by the Company in general meeting or under this Aitlele 117. Notwithstanding the aforesaid any Member (or, in the case of the Tapal Shareholders, any group of the maching collectively) who holds shares sufficient to elect at least one director shall have the right to inspect, in person or through its authorised representatives, the books and records of the Company at all reasonable times. The Company, shall upon request provide each such Member with copies of all tax returns promptly after the filling of such tax returns with the appropriate governmental authorities and copies of proof of payment.	
		submitted in connection with such tax returns. The Members shall cause the Board to adopt appro- priate resolutions permitting the Members access to the books, records and other documents as they may reasonably request.	
	<b>*18</b> .	The Directors shall as required by Sections 233 and 236 of the Ordinance cause to be prepared and to be laid before the Company in general meeting such profit and loss account and balance sheets duly audited and reports as are referred to in those sections.	
	119.	The balance sheet, profit and loss account, and other reports referred to in Article 118 shall be made out in every year and laid before the Company in annual general meeting made up to a date not more than six Months before such meeting. The balance sheet and profit and loss account shall be accompanied by a report of the euditors' of the Company and the report of Directors.	
	120	A copy of balance sheet and profit and loss account and reports of Directors and auditors shall, at least twenty one days preceding the meeting, the sent to the persons entitled to receive notices of general meetings in the manner in which notices are to be given hereunder.	
	121, <sup>.</sup>	The Directors shall in all respect comply with the provisions of Section 230 to 236 of the Ordinance.	
↓ •1.		AUDIT	
	122. 	Auditors shall be appointed and their dutles regulated in accordance with Sections 252 to 255 of the Ordinance. The Company's annual financial statements shall be prepared in accordance with gen- erally accepted accounting principles of Pakistan.	
		NOTICES	
	<b>12</b> 3.	(a) A notice may be given by the Company to any Member either personally or by sending it by post to him to his registered address or (if he has no registered address in Pakislan) to the address, if any, within or outside Pakistan supplied by him to the Company for the giving of notices to him.	
		(b) Where a notice is sent by post, service of the notice shall be deemed to be effected by properly addressing, prepaying, and posting a letter containing the notice and, unless the contrary is proved, to have been effected at the time at which the letter would be delivered in the ordinary course of post.	
	124.	A notice may be given by the Company to the joint holders of a share by giving the notice to the joint , holder named first in the Register in respect of the share and a notice so given shall be sufficient notice to all the holders of such shares.	
	12 <del>5</del> .	A notice may be given by the Company to the persons entitled to a share in consequence of the death or insolvency of a Member through the post in a prepaid letter addressed to them by name or	
		Page 25	
í.			
1			

.

г., Эн

۶ ۲.

3,

	TAM		· ·	Ð
	EMERG	tion, be e	he title or representatives of the deceased, or assignce of the insolvent or by any like descrip- , at the address (if any) in or out of Pakistan supplied for the purpose by the persons claiming to initied, or (until such an address has been so supplied) by giving the notice in any manner in the same might have been given if the death or insolvency had not occurred.	
i., ; ; ,	128,	Men have them who	ca of every general meeting shall be given in the manner hereinbefore authorised to (a) every ober of the Company, except those Members who having no registered address within Pakistan a not supplied to the Company an address within or outside Pakistan for the giving of notice to n; (b) every person entitled to a share in consequence of the death or insolvency of a Member, but for his death or insolvency would be entitled to receive notice of the meeting; (c) the tors of the Company for the time being; and (d) the Directors.	
; . ; ;	•		WINDING UP	
: .	127.	(a)	If the Company is wound up, the Ilguidator may, with the sanction of Special Resolution of the Company and other sanctions required by the Ordinance, divide amongst the Mem- bers, in specie or kind the whole or any part of the assets of the Company, whether they consist of property of the same kind or not.	·
		<b>(</b> D)	For the purpose aforesaid, the liquidator must set a value as he deems in good failh fair upon any property having due regard for the property's fair marked value to be divided as aforesaid.	• •
¦     .	•	(c)	The liquidator shall distribute the cash and non-cash proceeds form the sale of the Compa- ny's assets, and any asset not sold, except as may be otherwise provided by applicable law, in the following order of priority:	
•		Ø	All of the Company's debts and liabilities shall be paid and discharged in the order of priority provided by applicable law; and	
	•	(ii)	The balance shall be distributed to the Members pro rate in accordance with the ratio which the number of shares held by each Member beers to the total number of issued and paid up shares of the Company.	
		<b>(d)</b>	The liquidator may, with the like sanction, vest the whole or any part of such assets for the benefit of the contributories as the liquidator, with the like sanction, thinks fit, but so that no member shall be compelled to accept any shares or other securities whereon there is any liability.	
		(e)	Upon completion of distribution pursuant to this Article the Infuldator shall take such ac- itions, including executing any documents or agreements as may be necessary to dissolve the Company under applicable law.	· .
•			SECRECY	
٠ ٠	128	(a)	Subject to Article 128(b) and (c) no Member nor its Affiliates or nominees shall reveal to any third party, without the prior consent of the other Members any information concerning the organisation, business, manufacturing processes, finances, transactions or affails of the Company, the other Members or any of the other Members' Affiliates or hominees, or any information related to the Agreement (the "Confidential Information") and shall not use	. • •
			Page 26	
			· ·	

Ē

: 1

any such Confidential Information in any manner which may directly or indirectly injure the any such Confidential Information in any manner which may directly or indirectly injure the Company, any other Member or any of the other Members' Affiliates or nominees, except, that such Confidential Information may be disclosed if, and only to the extent, required by law, or to any lender or prospective lender or other parties providing financing to the Com-pany, provided that such lender, prospective lender or party providing financing thas sup-plied an undertaking, which shall be acceptable (i) in the case of information pertaining to a Member, to that Member and (ii) in the case of information pertaining to the Company, to the case of information pertaining the supervision of the company, to the case of information pertaining to the company, to the case of information pertaining to the company, to the case of information pertaining to the company, to the case of information pertaining to the company. the Board in its sole, discretion, to keep such Confidential Information confidential on sub-3 stantially the same terms as are contained in this Article. The provisions of this Article shall į. not apply (i) in respect of information which is or has become publicly available through no act or omission of a Member, its Affiliates, agents or nominees, (ii) to the extent such information was in the possession of the Member, its Affiliates, agents or nominees prior to ż. Its earliest receipt from any other Member or the Company, or (III) as may be required in any report, statement or lestimony required to be submitted to any municipal, state or national regulatory body, any court or any arbitration tribunat having or claiming to have jurisdiction over it, or as may be otherwise required by law, regulation, court order or other governmental authority, or as may be required in response to any summons or subpoena or in connection with any litigation or arbitration under the Agreement. In the course of an offer to a third party to sell, transfer, assign, dispose of pledge, charge or hypothecate any shares, a Member shall be entitled to disclose Confidential Information to such third party, provided that: (b) the Member promptly informs the other Members of the substance of all Confidential Infor-mation disclosed; and (1) no information shall be disclosed to the third party unless the third party has first supplied to Űľ. the other Members an undertaking in writing on substantially the same terms as Article 128(a). The Board shall exclusively control all cublic disclosures and communications with the press regarding the Company. INDEMNITY Every Director or officer of the Company and every person employed by the Company as auditor shall be indemnified out of the funds of the Company against all liability incurred by him as such Director, officer or auditor in defending any proceeding, whether civil of criminal, in which judgment is given in his favour, or in which he is acquitted, or in connection with any application under Section 488 of the Ordinance in which relief is granted to him by the court. 129. RECONSTRUCTION On any sele of the undertaking of the Company, the Directors or the liquidator on a winding-up that if authorised by Special Resolution, accept fully paid or partly pak-up shares, debentures or secu-rities of any other company, whether incorporated in Pakistan or not either then existing of to be formed for the purchase in whole or in part of the property of the Company, and the Directors (if the profils of the Company permit) or the liquidator (in a winding-up) may distribute such shares, or any other property of the Company amongst the Members without realization, or vest the same intrus-tees for them, and any Special Resolution may provide for the distribution or appropriation of the 130 -Page 27

ŷ

۸.)



131.

cash, shares, or other securities, benefits or property, otherwise than in accordance with the strict legal rights of the Members or contributories of the Company, and for the valuation of any such securities or property at such price and in such manner as the meeting may approve, and all holders of shares shall be bound by any valuation or distribution so authorized, and waive all rights in relation thereto, save only in case the Company is proposed to be or is in the course of being wound up, such statutory rights if any, under Section 367 of the Ordinance as are incepable of being varied or excluded by these Articles. 9i 4

### CHANGE OF NAME

Page 28

.

If at any time the aggregate shareholding of the Tapal Shareholders fails below twenty six per cent (26%)of the issued and paid up shares of the Company, the Tapal Shareholders shall be entitied to require that the name of the Company be changed so as to remove the word. Tapal' from such name and each Member shall take all such steps as lies within its power or control to ensure that the Company effects such change.

We, the several percent whole names and actnesses are subscribed below are desirous of being formed into a Company set opposite our respectively agree to take the mather of abrees in the Capital of the Company set opposite our respective names.       Articlas of Association, and we respectively agree to take the mather of abrees in the Capital of the Company set opposite our respective names.     Membrane of the company set opposite our respective names.       Present and Isome     Perturbation     With solution     Membrane of the company set opposite our respective names.       Present and Isome     Perturbation     With solution     Names in the company set opposite our respective names.       Present and Isome     Perturbation     With solution     Names in the company set opposite our respective names.       Res Location     Activation     Names in the company set opposite our respective names.     Names in the company set opposite our respective names.       Res Location     Activation     Names in the company set opposite our respective names.     Names in the company set opposite our respective names.       Res Location     States     Names in the company set opposite our respective names.     Names in the company set opposite our respective names.       Res Lower     Names in the company set opposite our respective names.     Names in the company set opposite our respective names.       Res Lower     Names in the company set opposite our respective names.     Names in the company set opposite our respective names.       Res Lower     Names in the company set
---

1

3

ß

Dated...

## 6.0 Text of the Proposed Modification

Tapal Energy (Private) Limited ("**Company**") is a [126] MW thermal power project located at Deh Gondpass, Tapo Gabopat, Hub River Road, Taluka & District Karachi (West), Karachi, Sindh, Pakistan. ("**Project**").

As such, the Company, through the present Licensee Proposed Modification, desires to modify its Generation License by extending its term up to the expiry of their Power Purchase Agreement dated September 26, 1995, entered into with KE, ("**PPA**") i.e. June 19, 2019, pursuant to the Regulations read with Rule 5 of the Generation Rules, which provides that, upon expiry of the term of the generation license, NEPRA may renew the generation license "for such further term as it may deem appropriate in the manner provided for in the generation license."

In relation to the above, the modification required in Article 4 of the Generation License of the Company is as follows:

"(1) Pursuant to Rule 5 of the Rules, this License is granted until June 19, 2019."



## 7.0 A Statement of the reasons in support of the modification;

The Company is a private limited company registered under the Companies Ordinance, 1984.

The Company has developed a 126 MW power project in the province of Sindh. The Project is developed pursuant to the Power Policy of 1994 for the purposes of generation and sale of electric power to KE. In January 1997, the Company initially obtained license under the Electricity Act, 1910. After the establishment of NEPRA, the Company obtained the Generation License in accordance with the prevailing regulatory regime.

On August 26, 2003, NEPRA issued Generation License no. IPGL/010/2003 to the Company, under Section 15 of the Regulation of Generation, Transmission and Distribution of Electric Power Act, 1997 ("NEPRA Act") for a term of fifteen (15) years, the expiration date being 25<sup>th</sup> August 2018 which is around 10 months short of the expiry date of the Company's existing PPA term. The Regulations provides for the modification of a generation license including modification to extend the term of its generation license while the Generation Rules also provides for renewal of a generation license.

On 20<sup>th</sup> June 1997, the Project successfully achieved its commercial operation date and has been performing its contractual obligations, inter alia, under the Power Purchase Agreement (PPA) signed between the Company and KE. The term of the PPA between the Company and the KE is valid upto 19<sup>th</sup> June 2019 requiring the Company to perform its obligations and deliver electric power to KE. (Copy of the PPA is attached as Annexure A.

In order to discharge its contractual obligations under the PPA the Company is required to obtain and maintain a generation license from NEPRA for the term of the PPA. It is emphasized that the maintenance of a valid generation license is a critical regulatory consent without which the Company cannot perform its contractual obligations under the PPA

The required modification would benefit K-Electric consumers as it ensures generation continues from the generation facility for the entire term of the PPA. Moreover, there is no adverse impact on the quality of service and performance of the Licensee from approval of this Application.

The request by the Company does not seek any change in the existing generation license except for the modification in the expiry date of the license therefore the details and schedules attached with the existing generation license also remain valid for this Application.

We hope NEPRA will consider our request and provide us the Modified Generation License based on information/explanation provided herein as well as reasons stated in Section 8 below.



# 8.0 A Statement of the impact on the tariff, quality of service and the performance by the licensee of its obligations under the license

## **Impact on Tariff**

As the tariff given in the PPA is for the term of the 22 years therefore there is no change required in the tariff thereby the requested modification in the generation license do not have any additional financial implications.

## Impact on Quality of Service & Performance

The Company confirms that the quality of service and the performance under the Generation License shall not be affected during the proposed extended period. The Company has been fully diligent and dedicated in the performance of its services according to the provisions of the PPA and shall continue to assist in reducing the demand supply gap in KE network.

## Impact on the Obligations of the Company under the License

The proposed modification would facilitate the Company in fulfilling its obligations under the License and its obligations under the PPA.



9.0	Glossary	of Terms

"NEPRA Act"	Regulation of Generation, Transmission and Distribution of Electric Power Act, 1997
"Regulations"	NEPRA Licensing (Application and Modification Procedure) Regulations, 1999
"Generation Rules"	NEPRA Licensing (Generation) Rules, 2000
"PPA" with KE	Power Purchase Agreement dated September 26,1995 entered
"KE"	Karachi Electric Limited
"NEPRA"	National Electric Power Regulatory Authority
"NTDC"	National Transmission & Despatch Company
"O&M"	Operations & Maintenance



10.0 Attachments:

Annexure A - Power Purchase Agreement dated September 26, 1995

## THE KARACHI ELECTRIC SUPPLY CORPORATION LIMITED

1

Alt- ille

- and -

TAPAL ENERGY LIMITED

## POWER PURCHASE AGREEMENT

~ RELATING TO -

A POWER GENERATION COMPLEX AT

## HUB RIVER ROAD, IN THE DISTRICT OF KARACHI WEST, PROVINCE OF SINDH, PAKISTAN

MADE AT KARACHI ISLAMIC REPUBLIC OF PAKISTAN

DATED 26TH SEPTEMBER, 1995

# INDEX

<u>Page</u>

ARTICLE I	DEFINITIONS AND RULES OF INTERPRETATION	2
1.1	Definitions	2
1.2	Rules of Interpretation	, 13
ARTICLE II.	SALE AND PURCHASE OF ENERGY AND CAPACITY	15
2.1	Energy and Capacity	. 15
2.2	Provisions for Despatch	. 15
2.3	Emergency Despatch of the Complex	
2.4	Observance of Technical Limits	. 16
ARTICLE HI	. AGREEMENTS OF THE PARTIES	17
3.1	Permits, Licenses and Approvals	
3.2	Feasibility Report	
3.3	Submittal of Data	
3,4	Operating Procedures	
3.5	Delivery of Energy from RESC to the Complex	.21
3.6	Completion of Interconnection and Transmission Facilities	.21
3,7	Appointment of Construction Contractor	
3,8	Operations and Maintenance of Complex	
3.9	KESC Observation Visits	
3.10	KESC Consent	. 22
3.11	Failure to Submit Reports	.22
ARTICLE IV	TERM AND TERMINATION	24
4.1	Term of Agreement.	
4.2	Company Events of Default - Termination by KESC	
4.3	KESC Events of Default - Termination by the Company	
4,4	Termination Notices; Termination	
4.5	Notice to Lender of Company's Default	
4,6	Obligations Upon Termination.	
4.7	Reinbursement	
4.8	Other Remedies	
4,9	Notice to the GOP of KESC's Default	
4.10	Termination of Implementation Agreement	
1 15 m 1 / 1 1 - 1		
	REPRESENTATIONS, WARRANTIES AND COVENANTS	
5.1	Representations, Warranties and Covenants of the Company	
5.2	Representations. Warranties and Covenants of KESC	.35

Alk

. L.

**--**

. .

ļ.

i \_\_\_

ARTICLE V	I. CONTROL AND OPERATION OF THE COMPLEX, DESPATCH	36
6.1	Declared Available Capacity Declaration	
6.2	Scheduling and Despatch	
6.3	Scheduled Outage Periods	
6.4	Operation in Accordance with Despatch	
6.5	Recording of Telephoned Communications	40
6,6	Emergency Set-Up and Curtailment Plans	40
6.7	Supply of Power in Emergency.	
6.8	Employment of Qualified Personnel	
6.9	Operating Committee Membership and Duties	41
6.10	Maintenance of Operating Records	
6,11	Cessation of Operation of the Complex	43
ARTICLE V	II. INTERCONNECTION	44
7.1	Interconnection and Transmission Facilities	
7.2	Data Necessary for Construction of Interconnection and Transmission	
	Facilities	44
7.3	Granting of Easements and Rights of Way	
7.4	Construction of Interconnection and Transmission Facilities	
7,5	Interconnection Equipment on Company's Side of Interconnection Point	
7.6	Protective Devices	
7.7	Testing	
	III. METERING & TELECOMMUNICATIONS	47
	Metering System	
8.1 8.2	Installation of Metering Systems	
8.2 8.3	Testing of Metering System.	
8.4	Reading of Meters	
	Sealing of Metering System.	
8,5	Repair, Replacement or Recalibration of Metering System	
8.6	Telecommunications Circuit	50
8.7	Telecommunications Circuit	
	X. COMPENSATION, PAYMENT AND BILLING	51
9.1	Capacity Payments	
9,2	Energy Payments	
9.3	Energy Purchase Price Premium	
9.4	Liquidated Damages and Penalties	
9.5	Payment of Liquidated Damages	
9.6	Bonus	
9.7	Billing	
9,8	Payment Disputes	
9,9	Proration Resulting From KESC's Operation or Termination	
9.10	Supplemental Tariff Payments	
9.11	Retained Payments Fund	
M.	16	
ny	~ γ	
		k

K

ł

Ţ

1

.

( ) .

.

\_\_\_\_

ARTICLE X.	TESTING AND CAPACITY RATINGS	62
10.1	Testing of the Complex Prior to Commercial Operations Date	
10.2	Tests Prior to Synchronization of the Complex	
10.3	Tests After Synchronization of the Complex and Commercial Operations	
	Test,	., 63
10.4	Testing of Dependable Capacity of the Complex after Commercial	
	Operations Date	
10,5	Notice of and Compliance With Testing Procedures	66
10.6	Copies of Test Results	66
10.7	Testing Disputes	
10.8	Estimated Dependable Capacity	66
ARTICLE XI.	INSURANCE	. 67
11,1	Maintenance of Insurance Policies	
11.2	Maintenance of "Occurrence" Form Policies	
. 11.3	Policy Endorsements	
11.4	Endorsements to Fire and Perils and Machinery Breakdown Policies	68
11.5	Certificates of Insurance.	
11,6	Insurance Reports	
	LIABILITY AND INDEMNIFICATION	
12.1	Limitation of Liability	
12.2	Indemnification	
12.3	Assertion of Claims to Exceed Minimum Indemnification Amount	
12.4	Indemnification for Fines and Penalties	
12.5	Defense of Claims	71
ΔΡΥΤΟΙ Ε ΥΠ	1. FORCE MAJEURE	72
13.1	Definition of Force Majeure	
13.2	Notification Obligations	
13.3	Duty to Mitigate.	
13.4	Delay Caused by Force Majeure	
13.5	Payments During Force Majoure Event	
13.6	Supplemental Tariff for Pakistan Political Force Majeure Events	
13.7	Supplemental Tariff for Change in Law	
15,7	Supportional faint for Onlingo in Daw	
ARTICLE XI		
14.1	Taxes Applicable to the Company	78
14.2	Taxes Applicable to KESC	78
	مال	
	l	

Ale

F

. )

\_\_\_\_

. 1

\_\_\_\_\_

.

]

-

1

t.

ARTICLE XV	/. RESOLUTION OF DISPUTES	79
15.1	Resolution by Parties	79
15.2	Mediation by Expert	79
15,3	Arbitration	81
15.4	Commercial Acts	82
ARTICLE XY	VI. NOTICES	83
16,1	Addresses and Addressees	83
16.2	Changes of Address	84
ARTICLE X	VII. MISCELLANEOUS PROVISIONS	
17.1	Amendment	
17.2	Headings	
17.3	Third Parties	
17.4	No Walver	85
17.5	Relationship of the Parties	
17.6	Survival	
17.7	Language	. 86
17.8	Choice of Law.	
17.9	Entirety	.86
17.10	Assignment	
17.11	Confidentiality	
17,12	Successors and Assigns	
17.13	No Liability for Review	
17.14	No Third Party Beneficiaries	, 89
17.15	Affirmation	
17.16	Approval Not to be Unreasonably Withheld or Delayed	
17.17	Double Jeopardy	. 89

## SCHEDULES

SCHEDULE 1 MINIMUM FUNCTIONAL SPECIFICATIONS SCHEDULE 2 TECHNICAL LIMITS SCHEDULE 3 INTERCONNECTION AND TRANSMISSION FACILITIES SCHEDULE 4 COMMISSIONING AND TESTING SCHEDULE 5 METERING STANDARDS AND TESTING SCHEDULE 6 TARIFF, INDEXATION AND ADJUSTMENT SCHEDULE 7 CONSTRUCTION REPORTS SCHEDULE 8 INSURANCE SCHEDULE 9 FORM OF LETTER OF CREDIT SCHEDULE 10 TESTING SCHEDULE

116

BL-

THIS POWER PURCHASE AGREEMENT is made at Karachi, dated September 26, 1995 between THE KARACHI ELECTRIC SUPPLY CORPORATION, a public limited company with its principal office located in Karachi, Pakistan ("KESC") and TAPAL ENERGY LIMITED, a public limited company incorporated under the laws of Pakistan, with its principal office located at Campbell Street, P.O. Box No. 51, Karachi 74200, Pakistan (the "Company"). Both the Company and KESC are herein referred to individually as a "Party" and collectively as the "Parties."

#### WHEREAS:

- (1) The Company plans to build, own and operate a power generation facility with capacity of approximately 126 MW gross ISO (the "Complex") in Sindh Province, Pakistan, using residual fuel oil purchased from the Pakistan State Oil Company Limited (the "Fuel Supplier"), pursuant to the terms and provisions of the Fuel Supply Agreement;
- (2) The Company wishes to sell to KESC, and KESC wishes to purchase from the Company, the Dependable Capacity of the Complex and all of the Net Electrical Output (as hereinafter defined) pursuant to the terms and conditions set forth herein; and
- (3) The Company has entered into the Implementation Agreement with the Government of Pakistan and will enter into the Fuel Supply Agreement with the Fuel Supplier;

I

**NOW, THEREFORE,** in consideration of the mutual benefits to be derived and the representations and warranties, conditions and promises herein contained, and intending to be legally bound hereby, the Company and KESC hereby agree as follows:

## ARTICLE I DEFINITIONS AND RULES OF INTERPRETATION

## 1.1 Definitions.

Whenever the following terms appear in this Agreement or the schedules hereto, whether in the singular or in the plural, present, future or past tense, they shall have the meanings stated below unless the context otherwise requires:

1.1 "Abandonment" - Voluntary cessation of operation of the Complex, and the withdrawal of all, or substantially all, personnel by Company from the Site for reasons other than KESC's acts or omissions or a Force Majeure Event; provided, however, that the Company shall not be deemed to have Abandoned the Complex so long as it is using all reasonable efforts to regain control of the Complex or reinstate such operation;

1.2 "Actual Initial Dependable Capacity" or "AIDC" - The Dependable Capacity of the Complex as at Commercial Operations Date, as determined by testing in accordance with Article X and established by the Company; provided, however, that such Dependable Capacity shall not exceed one hundred five percent (105%) of the Estimated Dependable Capacity.

1.3 "AGC" - Automatic generation control equipment.

1.4 "Agent" - The meaning ascribed thereto in Section 4.5.

1.5 "Agreement" - This Power Purchase Agreement, together with all schedules attached hereto, dated as of the date first entered above between KESC and the Company, as may be amended from time to time.

1.6 "Agreement Year" shall mean a period of twelve (12) consecutive months commencing on each consecutive anniversary of the Commercial Operations Date and ending as of the end of the Day preceding the next anniversary of the Commercial Operations Date, except for the first Agreement Year which shall start on the Commercial Operations Date; <u>provided</u>, <u>however</u>, that each Agreement Year shall automatically be extended by the occurrence of a Force Majeure Event declared by the Company (other than a Pakistan Political Force Majeure Event or a Change in Law) within such Agreement Year for a period equal to the sum of the Days the Company was unable to perform fully due to the Force Majeure Event, each such Day multiplied by one (1) minus the FM Ratio in effect on that Day; <u>provided</u>, <u>further</u>, that in the event of such extension, the immediately succeeding Agreement Year shall commence on the Day following the last Day of such extension and end after a period of twelve (12) consecutive months, and each Agreement Year thereafter shall have a period of twelve (12) consecutive months, in each case subject to any extension for Force Majeure Event(s) declared by the Company (other than a Pakistan Political Force Majeure Event or a Change in Law) as provided above.

1.7 "Annual Capacity Test" - The test to be conducted each Agreement Year during the Demonstration Period in accordance with Section 10.4 to determine the Dependable Capacity of the Complex.

1.8 "Average Dependable Capacity" - An amount equal to (a) the sum of (i) each Dependable Capacity, determined by an Annual Capacity Test conducted pursuant to Article X, in effect during an Agreement Year multiplied by (ii) the number of hours that each such Dependable Capacity was in effect during such Agreement Year, divided by (b) the number of hours in the Agreement Year; provided, that periods of Force Majeure shall be excluded from each determination of the number of hours in the preceding formula.

1.9 "Back-Up Metering System" - Any meters and metering devices installed, owned and maintained by the Company for backup purposes.

1.10 "Base Rate" - The repurchase rate for six-month Pakistan Treasury notes, as announced by the State Bank of Pakistan from time to time.

1.11 "Business Day" - Any Day that banks in Karachi, Pakistan are legally permitted to be open for business.

1.12 "Capacity Damages Amount" - The amount expressed in Rupees per MWh equal to the product of (a) the average Capacity Purchase Price provailing during the relevant Agreement Year and (b) 1.61.

1.13 "Capacity Payment" - The meaning ascribed thereto in Section 9.1.

1.14 "Capacity Purchase Price" - For each Agreement Year, the amount expressed in Rs. per kW per Month and shown in Table I to Schedule 6, as such amount is adjusted from time to time in accordance with Schedule 6.

1.15 "Certification Date" - The meaning ascribed thereto in Section 3.6.

1.16 "Change in Law" - (a) the adoption, promulgation, modification, repeal or reinterpretation after the date of this Agreement by any Public Sector Entity of any Law of Pakistan, or (b) the imposition by a Public Sector Entity of any material condition in connection with the issuance, renewal, extension, replacement or modification of any Consent after the date of this Agreement, that in either case establishes requirements for the construction, operation, or maintenance of the Complex that are materially more restrictive than the most restrictive requirements (i) in effect as of the date of this Agreement, (ii) specified in any applications, or other documents filed in connection with such applications, for any Consent filed by the Company on or before the Commercial Operations Date or (iii) agreed to by the Company in any agreement in the Security Package.

1.17 "Commercial Operations Date" - The Day following the date on which the Complex is Commissioned in accordance with Article X.

1.18 "Commercial Operations Tests" - The tests to be performed to establish the Commercial Operations Date in accordance with Article X.

Ulb Ale

1.19 "Commissioned" - The performance by the Complex during Commissioning at or above the criteria for attaining commercial operations specified in Schedule 4 hereto, as certified in writing to the Company and KESC by the Engineer.

1.20 "Commissioning" - For the Complex, engaging in testing the Complex in accordance with Section 10.3.

1.21 "Company" - Tapal Energy Limited, a public limited company incorporated under the laws of Pakistan, with its principal office located at Ameejee Chambers, Campbell Street, P.O. Box 51, Karachi 74200, Pakistan and its permitted successors and assigns.

1.22 "Company Event of Default" - An event described in Section 4.2 for which KESC may, subject to Section 4.4, issue a Termination Notice.

1.23 "Company Letters of Credit" - Unconditional and irrevocable direct-pay letters of credit issued by a bank or banks reasonably acceptable to KESC, in the form of Schedule 9 hereto, provided by the Company to KESC pursuant to Sections 9.4(f)(i)(A), (B) and (C), which shall provide for draws by KESC in immediately available funds on a Monthly basis and must permit presentation at a bank located in Karachi, Pakistan.

1.24 "Complex" - The approximately 126 MW gross ISO, power station utilizing residual fuel oil to be owned and constructed by the Company at off Hub River Road, in the District of Karachi West, Province of Sindh, Pakistan whether completed or at any stage in its construction, including without limitation or regard to level of development, land, engineering and design documents, all energy producing equipment and its auxiliary equipment, fuel storage and handling facilities and equipment, a switchyard, interconnection facilities (other than the Metering System, the Interconnection Facilities and the Transmission Facilities) necessary for delivery of electricity to KESC at the Interconnection Point, together with a colony to house certain employees of the Company, the Contractors or any subcontractors.

1.25 "Consents" - All such approvals, consents, authorizations, notifications, concessions, acknowledgements, agreements, licenses, permits, decisions or similar items required to be obtained from any Public Sector Entity for the Company or for the construction, financing, ownership, operation, and maintenance of the Complex, including without limitation those Consents listed in Schedule 1 of the Implementation Agreement; provided, however, that in no event shall the Consents include any concessions or exemptions from the Laws of Pakistan unless such concessions or exemptions are expressly granted to the Company pursuant to the terms of the Implementation Agreement.

1.26 "Construction Contract" - The agreements entered into between the Company and the Construction Contractor for the design, manufacture, engineering, procurement, construction, installation, completion, start-up, testing, and Commissioning of the Complex, as amended from time to time.

1.27 "Construction Contractor" - Stork-Wartsila Diesel B.V. and any successor or successors thereto appointed by the Company and not objected to by the GOP pursuant to Section 8.2(b) of the Implementation Agreement.

4

Ale

1.28 "Construction Financial Closing" means the execution and delivery of one or more loan agreements (following the resolution of any objections raised by the GOP to the term sheet or schedule in accordance with Section 15.3 of the Implementation Agreement that sets out a principal repayment schedule and the other principal terms of the transaction between the Company and the Lenders) and/or the receipt of commitments for equity by the Company, which debt and equity provide, in the aggregate, sufficient funding for the construction, testing and completion of the Complex; provided, however, that such equity commitments may not be less than required by the Letter of Support.

1.29 "Construction Start" - The issuance of the "notice to proceed" by the Company to the Construction Contractor and the unconditional release by the Company to the Construction Contractor of funds equaling seven percent (7%) or more of the total capital cost of the Project.

1.30 "Contractors" - The Construction Contractor and the O&M Contractor and any other direct contractors and any of their direct sub-contractors integrally involved in the Project.

1.31 "Control Centre" - KESC's Control Centre located in Karachi, or such other control centre designated by KESC from time to time (but not more than one at any time) from which KESC shall Despatch the Complex.

1.32 "Day" - Each twenty-four (24) hour period beginning and ending at 12:00 midnight Pakistan time.

1.33 "Declared Available Capacity" - The estimated net capacity of the Complex announced daily by the Company pursuant to Section 6.1 which shall equal the Dependable Capacity adjusted for expected temperature conditions and, in the event of a Scheduled Outage, Forced Outage, Partial Forced Outage, or Maintenance Outage, shall equal the Dependable Capacity, adjusted for expected temperature conditions, less the reduction due to such Scheduled Outage, Forced Outage, Partial Forced Outage or a Maintenance Outage.

1.34 "Demonstration Period" - A period of ninety (90) consecutive days designated in accordance with Sections 6.2(a) and 6.3 during which the Annual Capacity Test shall be conducted by the Company.

1.35 "Dependable Capacity" - The amount of capacity (adjusted to Reference Conditions), expressed in kW's, at the outgoing busbars of the 132kV substation of the Complex as determined by testing from time to time in accordance with Article X.

1.36 "Despatch" - The right of KESC to issue instructions or its issuance of instructions from the Control Centre in accordance with Prudent Utility Practices and this Agreement, including, without limitation, the Technical Limits, to schedule and control (or, in the case of AGC, to control) the generation of the Complex in order to commence, increase, decrease or cease the Net Electrical Output delivered to the Grid System.

1.37 "Dispute" - Any dispute or disagreement of any kind whatsoever between KESC and the Company in connection with or arising out of this Agreement.

5

· ph Mk

1.38 "Dollars" or "\$" - The lawful currency of the United States of America.

1.39 "Economic Despatch" - The distribution of the total KESC energy needs among available sources for optimum system economy, security and reliability with due consideration of incremental generating costs, incremental power purchase costs, incremental transmission losses, load flow considerations and other operational considerations as reasonably determined solely by KESC; <u>provided</u>, <u>however</u>, that Economic Despatch shall not require the Company to operate the Complex in any manner that is contrary to the Technical Limits or Prudent Utility Practices, nor require KESC to alter its then-current mode of operation and maintenance of the Grid System because of the Complex.

1.40 "Emergency" - A condition or situation that, in the sole but reasonable opinion of KESC, does materially and adversely, or is likely materially and adversely to (i) affect the ability of KESC to maintain safe, adequate and continuous electrical service to its customers, having regard to the then-current standard of electrical service provided to its customers, or (ii) present a physical threat to persons or property or the security, integrity or reliability of the Grid System.

1.41 "Energy Payment" - The meaning ascribed thereto in Section 9.2.

1.42 "Energy Purchase Price" - The lesser of (a) the amount, expressed in Rs. per kWh, identified as the Energy Purchase Price in Schedule 6, as such amount is adjusted from time to time in accordance with the provisions of Schedule 6 or (b) the amount nominated by the Company pursuant to Section 9.2(c).

1.43 "Engineer" - The independent consulting engineer, or engineering company, of international repute acceptable to KESC, the Company and the Lenders for the purposes of monitoring the construction and certifying the results of Commissioning.

1.44 "Equivalent Weighted Forced Outage Energy" - The meaning ascribed thereto in Section 9.4(b)(ii).

1.45 "Escrow Account" - The account that may be established by KESC pursuant to Section 9.4(f)(ii)(B) for the payment of amounts owed to the Company under this Agreement.

1.46 "Escrow Agreement" - The agreement between KESC, the Company and the Escrow Bank in Pakistan controlling the disposition of funds deposited in the Escrow Account by KESC.

1.47 "Escrow Bank" - A scheduled Pakistani bank in Pakistan reasonably acceptable to the Company with which KESC establishes the Escrow Account.

1.48 "Estimated Dependable Capacity" - The Dependable Capacity to which the Company commits pursuant to Section 10.8 hereof.

1.49 "Feasibility Report" - Any report prepared by or on behalf of the Company evaluating the technical feasibility of the Project, including an environmental impact study for the Complex.

1.50 "Federal Entity" - Any Public Sector Entity subject to the overall control or direction as to matters of policy, of the GOP or which is otherwise under or controlled by the GOP which includes,

No Als

without limitation, but only for so long as such entities are under the control of the GOP, KESC, the Pakistan State Oil Company Limited, the National Bank of Pakistan and the State Bank of Pakistan.

1.51 "Financial Closing" - (a)(i) The execution and delivery of one or more loan agreements, or (ii) the conversion or extension to term loans of the loan Agreements executed at the Construction Financial Closing, or (iii) any combination of (a)(i) and (a)(ii) above, that together evidence the financing for the construction, testing, and completion and/or the permanent financing of the Complex (and in each case, following the resolution of any objections raised by the GOP to a term sheet or schedule in accordance with Section 15.3 of the Implementation Agreement that sets out a principal repayment schedule and the other principal terms of the transaction between the Company and the Lenders or following the date on which the GOP shall be deemed to have not objected to such term sheet and principal repayment schedule pursuant to Section 15.3 of the Implementation Agreement) and (b) the receipt of commitments for such equity as is required by the Company to satisfy the requirements of the Lenders and the Letter of Support.

1.52 "Financing Documents" - The loan agreements, notes, indentures, security agreements, guarantees and other documents (including refinancings and revolving credit facilities) relating to the construction and/or permanent financing of the Company in relation to the Project, including, without limitation, shareholder loans and guarantees.

1.53 "FM Ratio" - As provided in Section 13.5.

F

1.54 "Forced Outage" - A total interruption of the Complex's generating capability (excluding periods, if any, that the Complex is being operated by KESC pursuant to Section 6.11) that is not the result of (i) a request by KESC in accordance with this Agreement; (ii) a Scheduled Outage or a Maintenance Outage; (iii) a Force Majeure Event; or (iv) a condition caused by KESC or by the Grid System.

1.55 "Force Majeure Event" - The meaning ascribed thereto in Section 13.1.

1.56 "Fuel Supplier" - Pakistan State Oil Company Limited, its successors and permitted assigns.

1.57 "Fuel Supply Agreement" - The agreement dated on or after the date hereof, between the Fuel Supplier and the Company for the supply of residual fuel oil to be used by the Complex to generate electricity, as amended from time to time, on terms reasonably acceptable to the GOP.

1.58 "GOP" - The Government of the Islanuc Republic of Pakistan, and its successors.

1.59 "Grid System" - The transmission or distribution facilities owned by KESC, other than the Interconnection Facilities and the Transmission Facilities, through which the Net Electrical Output of the Complex will be received and distributed by KESC to users of electricity and electrical energy delivered by KESC to the Complex, as required.

1.60 "Guarantee" The guaranty by the GOP of (i) the payment obligations of KESC under this Agreement (ii) the payment obligations of the Fuel Supplier under the Fuel Supply Agreement, and (iii) if the Company obtains Exchange Risk Insurance, the obligations of the National Bank of Pakistan

Who Alber

under the Exchange Risk Insurance (as such term is defined in the Implementation Agreement), substantially in the form set out in Schedule 3 to the Implementation Agreement, as may be amended from time to time by agreement of the parties thereto.

1.61 "Implementation Agreement" - The Implementation Agreement dated August 8, 1995, between GOP and the Company, as may be amended from time to time.

1.62 "Initial Shareholders" - The meaning ascribed thereto in the Implementation Agreement.

1.63 "Interconnection Facilities" - All facilities and equipment described in Schedule 3, including any telemetering equipment; transmission lines and associated equipment; transformers and associated equipment; relay and switching equipment; and protective devices and safety equipment (but not including the Metering System or the Back-up Metering System) -- which must be constructed or installed by or for KESC on the Site on KESC's side of the Interconnection Point to enable KESC to receive Net Electrical Output and Dependable Capacity and to deliver electrical energy, as required, in accordance with this Agreement.

1.64 "Interconnection Point" - The physical points where the Complex and the Grid System are connected at the Site as specified in Section 1(a) of Schedule 3 at which the transfer of Net Electrical Output occurs between the Company and KESC.

1.65 "Invoice for Liquidated Damages" - A written invoice for liquidated damages provided to the Company by KESC pursuant to Section 9.5.

1.66 "Involce Dispute Notice" - The meaning ascribed thereto in Section 9.8.

1.67 "kW" - Kilowatt.

the May

1.68 "kWh" - Kilowatt-hour.

1.69 "KESC" - The Karachi Electric Supply Corporation Limited and its successors and permitted assigns.

1.70 "KESC Event of Default" - An event described in Section 4.3 for which the Company may, subject to Section 4.4, issue a Termination Notice.

1.71 "KESC Letter of Credit" - The unconditional and irrevocable direct-pay letter of credit issued by a scheduled Pakistani bank in Pakistan reasonably acceptable in form and substance to the Company, which shall provide for draws by the Company for the purposes of Section 9.7 in immediately available funds and which shall be in an amount equal to an aggregate of two (2) Months of Capacity Payments plus Energy Payments (each as adjusted in accordance with Schedule 6) computed on the basis of the AIDC and assuming that the Complex will be Despatched at sixty percent (60%) of AIDC or, with respect to any replacement letter of credit, computed on the basis of the then-prevailing Dependable Capacity and assuming that the Complex will be Despatched at sixty percent (60%) of the then-prevailing Dependable Capacity and assuming that the Complex will be provided by KESC to the Company pursuant to Section 9.4(f)(ii)(A).

1.72 "Lapse of Consent" - Any Consent (a) ceasing to remain in full force and effect or (b) not being issued or renewed upon application having been properly and timely made and diligently pursued or (c) being made subject, subsequent to its grant, upon renewal or otherwise, to any terms or conditions that materially and adversely affect the Company's ability to perform its obligations under any document included within the Security Package, in each of the above instances despite the Company's compliance with the applicable procedural and substantive requirements as applied in a "non-discriminatory" manner, as explained in Section 16.4 of the Implementation Agreement.

1.73 "Laws of Pakistan" - The Federal, Provincial and local laws of Pakistan, and all orders, rules, regulations, statutory regulatory orders (SRO's), executive orders, decrees, Policies, judicial decisions, notifications, or other similar directives made pursuant thereto, issued by any executive, legislative, judicial, or administrative entity, as any of them may be amended from time to time.

1.74 "Leuders" - The lenders party to the Financing Documents together with their respective successors and assigns.

1.75 "Letter of Support" - That certain Letter of Support issued by the Ministry of Water and Power dated 18 September 1994, as the same may have been amended or clarified prior to the date hereof.

1.76 "Loss" - Any loss, damage, liability, payment and obligation (excluding any indirect or consequential loss, damage, liability, payment or obligation), and all expenses (including without limitation reasonable legal fees).

1.77 "Maintenance Months" - The Months of November, December, January & February or any four Months in a Year, at least three of them consecutive Months, designated by KESC in accordance with Section 6.3.

1.78 "Maintenance Outage" - An interruption or reduction of the Complex's generating capability that: (i) is not a Scheduled Outage; (ii) has been scheduled and allowed by KESC in accordance with Section 6.3(e); and (iii) is for the purpose of performing work on specific components, which should not, in the reasonable opinion of the Company, be postponed until the next Scheduled Outage.

1.79 "Metering System" - All meters and metering devices (including remote terminal units) specified or provided by KESC in a timely manner and owned by KESC and used to measure Dependable Capacity and, for payment purposes, the delivery and receipt of Net Electrical Output.

1.80 "Minimum Functional Specifications" - The minimum functional specifications for the construction of the Complex as set forth in Schedule 1.

1.81 "Minimum Indemnification Amount" - The amount, equal to one hundred thousand Dollars (\$100,000) that a Party's claims for indemnification pursuant to Article XII must exceed in the aggregate before that Party will be entitled to indemnification.

plo Als

ł

"Month" - A calendar month according to the Gregorian calendar beginning at 12:00 1.82midnight on the last Day of the preceding month and ending at 12:00 midnight on the last day of that month.

1.83 "MVAR" - Megavar.

1.84 "MW" - Megawatt.

1.85 "MWh" - Megawatt-hour,

"Net Electrical Output" - Net electrical energy expressed in kWh delivered to the 1.86 Interconnection Point by the Company for sale to KESC during testing and Commissioning of the Complex and, following the Commercial Operations Date, when Despatched by KESC.

"Notice of Intent to Terminate" - A notice delivered by the Company or KESC, as the 1.87 case may be, of its intent to terminate this Agreement pursuant to Section 4.4.

1.88 "O&M Agreement" - The agreement to be entered into between the Company and the O&M Contractor for the operation and maintenance of the Complex, as may be amended from time to time.

"O&M Contractor" - Wartsila Diesel Pakistan (Private) Limited and any successor 1.89 thereto appointed by Company and not objected to by the GOP in accordance with the Implementation Agreement.

1.90 "Operating Committee" - The committee established pursuant to Section 6.9 for the purposes described therein.

"Operating Day" - Each period of twenty-four consecutive hours commencing on the 1.91 Commercial Operations Date beginning at 12:00 midnight; provided, that such periods during Scheduled Outages of the entire Complex shall not be Operating Days.

"Other Force Majeure Events" - The meaning ascribed thereto in Section 13.1(c). 1.92

"Pakistan Political Force Majeure Events" - The meaning ascribed thereto in Section 1.93 13.1(a).

"Partial Derating" - In the event of a Partial Forced Outage, the Dependable Capacity 1,94 of the Complex, expressed in MW, minus the Declared Available Capacity of the Complex, expressed in MW,

"Partial Forced Outage" - A decrease, but not total interruption, in the Complex's 1.95 generating capacity, including a shortfall as provided in Section 9.4(c), that is not the result of (i) a request by KESC in accordance with this Agreement; (ii) a Scheduled Outage or a Maintenance Outage; (iii) a Force Majeure Event; or (iv) a condition caused by KESC or the Grid System.

"Parties" - Both KESC and the Company. 1.96 10

HE

1.97 "Party" - Either KESC or the Company.

1.98 "Pass-Through Items" - Certain costs or charges identified as Pass-Through Items in Schedule 6.

1.99 "Policies" - Such policies adopted by the GOP, any Federal Entity, the Provincial Government or, where applicable, any political subdivision thereof, as have been published in writing and made available to the public.

1.100 "Premium Date" - The meaning ascribed thereto in Section 9.3.

1.101 "Prescribed Fee" - With respect to any Consent, the charge or fee, if any, prescribed by the Laws of Pakistan

1.102 "Prescribed Form" - With respect to any Consent, the form, if any, (including all information and details) prescribed by the Laws of Pakistan for the application for, or renewal of, such Consent.

1.103 "Project" - The development, design, engineering, manufacture, procurement, financing, construction, permitting, completion, testing, Commissioning, insurance, ownership, operation and maintenance of the Complex and all activities incidental thereto.

1.104 "Provincial Government" - The government of the Province of Sindh.

1.105 "Prudent Electrical Practices" - The use of equipment, practices or methods, as required to comply with applicable industry codes, standards, and regulations in Pakistan (i) to protect KESC's system, employees, agents, and customers from malfunctions occurring at the Complex and (ii) to protect the Complex and the Company's employees and agents at the Complex, from malfunctions occurring on the Grid System.

1.106 "Prudent Utility Practices" - The prudent utility practices followed from time to time by the electric utility industry in Pakistan, having regard to engineering and operational considerations, including manufacturers' recommendations. Prudent Utility Practices are not limited to optimum practices, methods or acts to the exclusion of all others, but rather are a spectrum of possible practices, methods and acts which could have been expected to accomplish the desired result at reasonable cost consistent with reliability and safety.

1.107 "Public Sector Entity" - The GOP and the Provincial Government, any subdivision of either, any local governmental authority with jurisdiction over the Company, the Project or any part thereof, courts and tribunals in Pakistan and any department, authority, instrumentality, agency, or judicial body of the GOP, the Provincial Government or any such local governmental authority, and including, without limitation, so long as they are Federal Entities, Pakistan Insurance Corporation, National Insurance Corporation, KESC, the Fuel Supplier, the National Bank of Pakistan and the State Bank of Pakistan, but excluding National Development Finance Corporation.

Mb Hk

ł

1.108 "Reactive Power" - The wattless component of the product of voltage and current, which the Complex shall provide to or absorb from the Grid System within the Technical Limits and which is measured in MVAR.

1.109 "Reference Conditions" - the reference conditions set forth in Schedule 1

1.110 "Required Commercial Operations Date" - The date sixteen (16) months after the date of the earlier to occur of either Construction Financial Closing or Financial Closing, which date may be extended in accordance with the terms of this Agreement as a result of Force Majeure Events and delays in testing pursuant to Section 10.1(b) or in the completion of the Interconnection Facilities and/or the Transmission Facilities pursuant to Section 3.6.

1.111 "Rupee" or "Rs." - The lawful currency of Pakistan.

1.112 "Scheduled Commercial Operations Date" - The date advised to KESC by the Company, as may be revised from time to time based on the scheduled construction programme, for the completion of Commissioning of the Complex.

1.113 "Scheduled Outage" - A planned interruption of the Complex's generating capability, or any portion thereof, that (i) has been scheduled and allowed by KESC in accordance with Section 6.3, and (ii) is for inspection, testing, preventive maintenance, corrective maintenance, repairs, replacement or improvement of the Complex.

1.114 "Security Package" - The meaning ascribed thereto in Article I of the Implementation Agreement.

1.115 "Site" - The land, spaces, waterways, roads, wells and any rights acquired or to be acquired by the Company for the purposes of the Complex on, through, above or below the ground on which the Complex, or any part thereof is to be built, (including, without limitation, any working areas required by the Company and the Contractors, villages, townships and camps for the accommodation of the employees of the Company and the Contractors and any subcontractors; and, all rights of way and access from public highways, railways, and seaward access, if applicable).

1.116 "Standards" - The environmental guidelines and occupational health and safety standards of the World Bank as in effect on the date of this Agreement.

1.117 "Start-Up" - After the Commercial Operations Date, any start up of the Complex requested by KESC that results in synchronization with the Grid System.

1.118 "Supplemental Charges" - Certain costs or charges identified as Supplemental Charges in Section 10 of Schedule 6, including the Pass-Through Items.

1.119 "Supplemental Tariff" - Additional compensation payable by KESC to the Company pursuant to Section 13.6 or Section 13.7 and approved in accordance with Section 17.7(d) of the Implementation Agreement.

Alle Alle

"Supplemental Tariff Payments" - Payments for the Supplemental Tariff in accordance 1.120with the payment schedule agreed to by the Company and KESC, which shall be designed to permit the Company to recover the Supplemental Tariff as provided in Schedule 6.

"Technical Limits" - The limits and constraints described in Schedule 2 relating to the 1.121 operation, maintenance and Despatch of the Complex.

"Term" - The term of this Agreement as specified in Section 4.1. 1.122

"Termination Notice" - A written notice of termination of this Agreement issued by 1.123 KESC or the Company, as the case may be, pursuant to Section 4.4.

"Transmission Facilities" - All facilities and equipment described in Schedule 3, 1.124including telemetering equipment; transmission lines and associated equipment; transformers and associated equipment; relay and switching equipment; and protective devices and safety equipment which are necessary to be constructed or installed by or for KESC between the dead-end tower(s) referenced in Schedule 3 and the Grid System to enable KESC to receive Net Electrical Output and Dependable Capacity and to deliver electrical energy, as required, in accordance with this Agreement.

1.125 "Week" - Each period of seven (7) consecutive Days beginning at 12:00 midnight Pakistan time falling between a Friday and a Saturday.

1.126 "Weighted Complex Partial Derating" - The meaning ascribed thereto in Section 9.4(b)(iii).

"Weighting Factor" - The meaning ascribed thereto in Section 9.4(b)(v). 1.127

1.128"World Bank" - The International Bank for Reconstruction and Development.

1.129 "Year" - Each twelve (12) Month period commencing on 12:00 midnight on December 31 and ending on 12:00 midnight the following December 31 during the Term.

1.2 Rules of Interpretation.

In this Agreement:

The headings are for convenience only and shall be ignored in construing this (a) Agreement;

(b) The singular includes the plural and vice versa;

References to Sections, Recitals and Schedules are, unless the context (c) otherwise requires, references to Sections of, and Schedules and Recitals to, this Agreement;

Unless otherwise provided herein, whenever a consent or approval is required (d)by one Party from the other Party, such consent or approval shall not be unreasonably withheld or delayed; and Nb Mb

13

(e) In carrying out its obligations and duties under this Agreement, each Party shall have an implied obligation of good faith.

pb Ale

i....

L

....

~--
# ARTICLE II SALE AND PURCHASE OF ENERGY AND CAPACITY

# 2.1 Energy and Capacity.

(a) Subject to and in accordance with the terms of this Agreement, from and after the Commercial Operations Date and until the expiration of this Agreement, the Company shall make available to KESC, and KESC shall purchase from the Company for the consideration described in Article LX:

(i) the Dependable Capacity up to one hundred five percent (105%) of the Estimated Dependable Capacity, and

(ii) the Net Electrical Output of the Complex to the extent that it is Despatched.

(b) Subject to and in accordance with the terms of this Agreement, prior to the Commercial Operations Date, KESC shall accept and purchase for the consideration provided in Section 9.2(b) all Net Electrical Output produced by testing performed pursuant to the Construction Contract and Commissioning; provided, however, that in no event shall the provisions of Sections 6.2 through 6.6 apply prior to the Commercial Operations Date.

(c) The Parties agree that, during the Term and any extension thereof, the Company shall not, without the prior written consent of KESC, sell or deliver electric energy produced by the Complex to any entity other than KESC.

# 2.2 Provisions for Despatch.

KESC shall have the right to Despatch the Complex, provided that (i) Despatch of the Complex must at all times be consistent with the Technical Limits and the operating procedures developed pursuant to Section 3.4; (ii) Despatch shall not exceed at any time the then-prevailing Declared Available Capacity and (iii) prior to the Commercial Operations Date, such right shall only be during an Emergency, subject to the provisions of Section 6.7. Except as provided in Sections 2.3 and 6.7, KESC shall Despatch the Complex in accordance with Economic Despatch.

2.3 <u>Emergency Despatch of the Complex</u>.

Ale\_

UL.

(a) Notwithstanding the provisions of Sections 2.1 or 2.2, KESC shall not be obligated to purchase or receive energy from the Complex, and may require the Company to disconnect or reduce energy deliveries:

(i) if in KESC's sole opinion, an Emergency exists; or

(ii) for so long as a disconnection or a reduction in energy deliveries is necessary to enable KESC to construct, install, maintain, repair, replace, remove, investigate,

15

inspect or test any part of the Interconnection Facilities, the Transmission Facilities or any affected part of the Grid System.

(b) KESC will use all reasonable efforts to notify the Company of reductions in the Complex's Net Electrical Output due to reasons specified in (i) above. With respect to (ii) above, to the extent that such reductions can be rescheduled or delayed in a manner that does not affect the reliability or integrity of the Grid System, KESC shall use reasonable efforts to coordinate such reductions in Net Electrical Output with the Company's Scheduled Outages, and the Company shall use reasonable efforts to reschedule its Scheduled Outages with such reduction in Net Electrical Output, so that long outages can be scheduled to coincide with Scheduled Outages. KESC shall provide the Company with at least forty-eight (48) hours prior notice of such reductions in Net Electrical Output when practicable. Any reduction required of the Company in accordance with (i) or (ii) above shall be implemented and completed as soon as possible consistent with Prudent Utility Practices and the Technical Limits.

# 2.4 Observance of Technical Limits.

Nothing contained in this Agreement shall be construed to require the Company to operate the Complex, at any time, including during an Emergency, in any manner inconsistent with the Technical Limits or the Laws of Pakistan.

JUS ME

1-

### <u>ARTICLE III</u> AGREEMENTS OF THE PARTIES

# 3.1 Permits, Licenses and Approvals.

(a) Prior to the Commercial Operations Date, pursuant to Section 6.2 of the Implementation Agreement, the Company shall, at its own expense, (i) make or cause to be made all applications (whether initial applications or renewal applications) for the Consents in the Prescribed Form and with the Prescribed Fee to the appropriate Public Sector Entities, shall diligently pursue all such applications and shall use all reasonable efforts to maintain in effect Consents once obtained; (ii) give all required notices and allow all required inspections under all Consents obtained by it in connection with the Complex; and (iii) pay all Prescribed Fees in connection with such Consents. The information supplied in the applications shall be complete and accurate and shall satisfy the substantive and procedural requirements of the applicable Laws of Pakistan applied in a "non-discriminatory" manner, as explained in Section 16.4 of the Implementation Agreement.

(b) After the Commercial Operations Date, the Company, at its own expense, shall (i) use all reasonable efforts to obtain and maintain in effect all Consents required in order to enable it to perform its obligations under this Agreement; (ii) give all required notices and allow all required inspections under all Consents obtained by it in connection with the Complex; and (iii) pay all Prescribed Fees in connection with such Consents.

(c) KESC shall, at its own expense, (i) use all reasonable efforts to obtain and maintain in effect all permits, licenses and approvals required by all Public Sector Entities with jurisdiction over KESC, the Grid System, the Interconnection Facilities, the Transmission Facilities and the Metering System in order to enable it to perform its obligations under this Agreement; (ii) give all required notices and allow all required inspections under all consents, permits, licenses and approvals obtained by it in connection with the Complex; and (iii) pay all prescribed fees in connection with such consents, permits, licenses and approvals.

#### 3.2 Feasibility Report

Within five (5) Days of receipt or completion of any Feasibility Report prepared by or on behalf of the Company or the Lenders, the Company will deliver a copy of each such Feasibility Report to KESC.

# 3.3 Submittal of Data.

(a) The Company shall submit to KESC the following documents on or before the specified dates:

(i) As soon as available, but no later than the earlier to occur of either Construction Financial Closing or Financial Closing, a copy of the Implementation Agreement as executed, with any amendments thereto;

ph are

(ii) Beginning within ninety (90) Days after the earlier to occur of either Construction Financial Closing or Financial Closing and ending on the Commercial Operations Date, (A) monthly construction progress reports substantially in the form set forth in Schedule 7 (or such other form as may be agreed to by the Parties), (B) such other reports as are submitted to the Company by the Engineer and (C) reports, when and as the Company becomes aware, of any new condition or event which will have a material and adverse effect on the timely completion of the Complex.

(iii) At least ten (10) Days prior to Construction Start, evidence demonstrating that the Company and the Contractors have obtained all material Consents as of the Construction Start.

(iv) At least sixty (60) Days prior to the scheduled commencement of testing and Commissioning of the Complex, a preliminary start-up and test schedule for the Complex;

(v) Not later than thirty (30) Days following the Commercial Operations Date, for the major items of plant incorporated into the Complex, copies, as received by the Company under the Construction Contract, of all manufacturers' specifications, manufacturers' operation manuals, and a certificate of the designated site engineer attesting to the fact that all equipment is new and unused;

(vi) As soon as available, signed and sealed copies of such as-built drawings for the Complex, including the civil and architectural works, as and when provided to the Company under the Construction Contract;

12

(vii) On or before the Commercial Operations Date, a certificate from the Engineer to the effect that the Complex has been constructed in all material respects in compliance with the terms of this Agreement and the Construction Contract such that the Company will be able to operate the Complex or cause it to be operated, in accordance with Prudent Utility Practices and the terms of this Agreement;

(viii) On or before the Construction Start, a copy of the Construction Contractor's All Risk Insurance Policy and as soon as available, but in any event on or before the Commercial Operations Date, and as and when updated, copies of all initial insurance policies and certificates of insurance or other evidence of insurance for policies detailed in Schedule 8;

(ix) Not later than thirty (30) Days prior to the Commercial Operations Date, (i) evidence demonstrating the Company has obtained from the Public Sector Entities having jurisdiction over the Company or the Project all of the Consents then required to be obtained for the ownership, operation and maintenance of, and the supply of power from, the Complex, together with (ii) a list identifying Consents not yet required to be obtained for the operation and maintenance of, and the supply of power from, the Complex, together with (ii) a list identifying such Consents and an estimate of the time within which such Consents will be obtained;

11/2--

(x) As soon as available, but not later than the Commercial Operations Date, copies of all preliminary results of tests performed on the Complex, including tests of major equipment included in the Complex, in accordance with the Construction Contract, and as soon as available but not later than thirty (30) Days after the Commercial Operations Date, copies of all final results of such tests;

(xi) As available, draft copies of the Fuel Supply Agreement, and no later than the earlier to occur of either Construction Financial Closing or Financial Closing, a copy of the Fuel Supply Agreement as will be executed on or before the earlier to occur of either Construction Financial Closing or Financial Closing, together with such other information as appropriate demonstrating that the Company has contracted for a reliable supply of fuel necessary to generate Net Electrical Output at the Dependable Capacity level for the Term (and a full requirements contract with the Fuel Supplier for a term not less than the Term shall constitute such evidence);

(xii) As soon as available, but no later than the earlier to occur of either Construction Financial Closing or Financial Closing, the Company shall provide KESC with any environmental assessment or study relating to the Complex that has been provided to the Company or to its Lenders;

(xiii) As of the date of Construction Financial Closing and again at Financial Closing, a certificate, dated as of such date, signed by the Chairman of the Board, President or a Vice-President and by a principal financial or accounting officer of the Company that (1) all representations and warranties of the Company in and under this Agreement are true, with the same effect as though such representations and warranties had been made on and as of the date of Financial Closing and (2) no Company Event of Default shall have occurred and be continuing;

(xiv) As of the Commercial Operations Date, a certificate, dated as of such date, signed by the Chairman of the Board, President or a Vice-President and by a principal financial or accounting officer of the Company that (1) all representations and warranties of the Company in and under this Agreement are true, with the same effect as though such representations and warranties had been made on and as of the Commercial Operations Date and (2) no Company Event of Default shall have occurred and be continuing; and

(xv) As soon as available, but no later than the earlier to occur of either Construction Financial Closing or Financial Closing, necessary technical documents and data required by Schedule 3, sufficient for KESC to design, construct and install the Interconnection Facilities and the Transmission Facilities.

(b) KESC shall provide to the Company (a) within thirty (30) Days after receipt of a written request from the Company, information on the Grid System and the design of the Interconnection Facilities, Transmission Facilities, and Metering Systems reasonably necessary for the Company to design, operate and maintain the Complex, (b) not later than sixty (60) Days prior to the Scheduled Commercial Operations Date, Despatch and communications procedures currently in use by

19

plas

KESC and (c) at least thirty (30) Days prior to their implementation, any planned changes to the Despatch and communication procedures then in effect.

(c) The Company shall notify KESC of any changes to the schedule described in subsection (iv) of Section 3.3(a) in a timely manner. The receipt of the above schedules, data, certificates and reports by KESC shall not be construed as an endorsement by KESC of the design thereof, does not constitute a warranty by KESC of the safety, durability or reliability of the Complex, nor otherwise relieve the Company of its obligations or potential liability under this Agreement, the Implementation Agreement, and the other documents comprising the Security Package or, except with respect to the obligations of KESC to maintain the confidentiality of documents and information received by it, create any obligation or liability on the part of KESC under any document contained in the Security Package.

(d) Within ten (10) Days after the earlier to occur of either Construction Financial Closing or Financial Closing, the Company shall deliver to KESC copies of each of the documents referenced in Section 3.3(a) as executed which execution copies were not previously submitted to KESC.

3.4 <u>Operating Procedures</u>.

(a) The Company shall develop, with the input and comments of KESC, written operating procedures for the Complex that are acceptable to the Parties no later than one hundred twenty (120) Days prior to the then-existing Scheduled Commercial Operations Date. The operating procedures shall be based on the designs of the Complex, the Metering System, the Interconnection Facilities, the Transmission Facilities and the Grid System and shall be consistent with the Technical Limits. The operating procedures shall deal with all operational interfaces between KESC and the Company, including, but not limited to, the method of Day-to-Day communication, Despatch procedures, identification and contact of key personnel, clearances and switching practices, outage scheduling, capacity and energy reporting, operations logs and Reactive Power support. The operating procedures are to be developed in accordance with the following process:

(i) KESC shall advise the Company of all matters and information KESC believes are to be included in the operating procedures at least ten (10) months prior to the then existing Scheduled Commercial Operations Date and within fourteen (14) Days of a request from the Company, shall make its representatives available to review such matters and information with the Company procedures;

(ii) the Company shall provide KESC with draft written operating procedures not later than eight (8) months prior to the Scheduled Commercial Operations Date;

Hes.

(iv) the Company shall revise the draft operating procedures to incorporate the comments of KESC and provide them to KESC not later than six (6) months before the Scheduled Commercial Operations Date. Any remaining disagreements concerning the operating procedures shall be referred to the Operating Committee for resolution and, if not resolved by the Operating Committee within thirty (30) Days of such referral, to the expert for final resolution. The expert shall be directed to render his decision at least one hundred twenty (120) days prior to the then-prevailing Scheduled Commercial Operations Date. If the expert has not rendered his decision by the date ninety (90) Days prior to the then-prevailing Scheduled Commercial Operations Date, the operating procedures as modified by KESC's comments shall be used for operating the Complex pending the decision by the expert and such delay by the expert shall not postpone the Commercial Operations Date pursuant to Section 3.11. Upon the rendering of the decision by the expert, the operating procedures for the Complex shall be modified accordingly.

(b) The Company and KESC shall mutually develop an inter-tripping schedule no later than sixty (60) Days prior to the required implementation of such schedule. Such inter-tripping schedule shall be based on a proposed schedule submitted to the Company by KESC at least one hundred and twenty (120) Days prior to the date implementation of such schedule is required.

# 3.5 Delivery of Energy from KESC to the Complex.

Upon the Company's request, KESC shall use its reasonable efforts to provide, subject to availability and KESC's ability to deliver to the Complex, at the sole cost and expense of the Company, energy for testing, Commissioning, and emergencies and for Start-Ups at the relevant tariff then charged by KESC for the provision of such class of energy service.

### 3.6 <u>Completion of Interconnection and Transmission Facilities</u>

(a)In the event that for any reason other than the reasons specified in Section 7.4, KESC has not completed the Interconnection Facilities and the Transmission Facilities by the completion date specified in Section 7.4, or such later date as may be determined by the Parties in accordance with this Agreement, and such delay causes any programme of tests referred to in Article X to be delayed beyond the date (the "Certification Date") as of which the Interconnection Committee (as defined in Section 9.3) certifies that the Complex is ready for the commencement of Commissioning tests but such testing cannot commence due to the unavailability of the Interconnection Facilities or the Transmission Facilities and that the delay in the programme of tests would not have occurred but for KESC's failure to timely complete the Interconnection Facilities and the Transmission Facilities, then, from and after the Certification Date, KESC shall pay to the Company Monthly, in arrears, (and pro rated for any portion of a Month) an amount equal to the carrying cost on the debt related to the Complex incurred under the Financing Documents plus fifty percent (50%) of the escalable component of the Capacity Purchase Price. Such payments shall continue until the completion of the Interconnection Facilities and the Transmission Facilities. In addition, the Required Commercial Operations Date and the Premium Date shall be extended on a Day-for-Day basis by the number of Days in the period commencing on the Certification Date and ending on the date KESC completes the Interconnection Facilities and the Transmission Facilities in accordance with Section 7.4. Except as provided in Section 4.3(e), the Company shall be entitled to no other penalty or claim for damages as a result of delay by KESC in completing the Interconnection Facilities and the Transmission Facilities.

phis Alt

21

KESC shall have no obligation to make the payments provided in this Section 3.6 if and to the extent that the delay in the programme of tests would have nevertheless occurred regardless of KESC's failure to timely complete the Interconnection Facilities and the Transmission Facilities. If payments by KESC under this Section 3.6 shall have commenced or the obligation for such payments shall have accrued, the Complex shall be tested at the first available opportunity thereafter.

### 3.7 Appointment of Construction Contractor.

KESC acknowledges the Company's appointment of Stork-Wartsila Diesel B.V. as Construction Contractor; <u>provided</u>, <u>however</u>, that such acknowledgement shall not relieve the Company of any of its obligations or potential liability regarding the design, financing, insuring, acquisition, construction or completion of the Complex or any liability whatsoever resulting from a breach of any term or condition of this Agreement.

### 3.8 Operations and Maintenance of Complex.

The Company shall operate and maintain the Complex; <u>provided</u>, <u>however</u>, that the Company may contract with the O&M Contractor to operate and maintain the Complex; <u>provided</u>, <u>further</u>, that the appointment of the O&M Contractor by the Company shall not relieve the Company of any of its obligations or potential liability regarding the insuring, operation or maintenance of the Complex or any liability whatsoever resulting from a breach of any term or condition of this Agreement.

#### 3.9 KESC Observation Visits.

KESC shall have the right, on a recurring basis and upon reasonable prior notice to the Company (except that no notice shall be required during an Emergency), to observe the progress of the construction of the Complex, the testing and Commissioning of the Complex in accordance with Article X, and the operation of the Complex. The Company shall comply with all reasonable requests of KESC for, and assist in arranging, any such observation visits to the Complex. All persons visiting the Complex on behalf of KESC shall comply with the reasonable instructions and directions of the Company or its Contractors. Such visits to the Complex shall not be construed as an endorsement by KESC of the design, construction or operation of the Complex nor as a warranty by KESC of the safety, durability of the Complex.

3.10 KESC Consent.

16

For the purposes of Section 28 of the Electricity Act, 1910, KESC hereby irrevocably consents to the generation of electricity by the Company at the Complex during the Term.

# 3.11 Failure to Submit Reports.

Any failure of either Party to timely submit any reports, information or certifications required by this Agreement, including, without limitation, the items required by Section 3.3, shall, in addition to any rights and remedies available to the receiving Party under law, give the receiving Party the right to

22

delay reciprocal action for which such information is provided, or the date or event in connection with which the information is provided, for a period equal to any such delay by the delivering Party.

UB Me

F

1

ſ

I

i.

~~

# ARTICLE IV TERM AND TERMINATION

# 4.1 <u>Term of Agreement</u>.

(a) This Agreement shall commence and be effective on the date hereof, and shall, unless extended or terminated earlier in accordance with the terms of this Agreement, continue in full force and effect for twenty-two (22) Agreement Years (the "Term").

(b) The Term shall automatically be extended by the aggregate of Days all Pakistan Political Force Majeure Events or Changes in Law declared by the Company were in existence, each such day multiplied by one (1) minus the FM Ratio in existence on each such Day. During each Month of such extension, the Company shall be paid the non-escalable portion of the Capacity Payment as in existence during such Pakistan Political Force Majeure Event, plus the escalable component of the Capacity Purchase Price applicable to the twenty-second (22nd) Agreement Year (as shall be further escalated pursuant to the provisions of Schedule 6) multiplied by the then-prevailing Dependable Capacity. The Term shall automatically be extended by the aggregate of all Days Force Majeure Events declared by KESC were in existence. During each Month of such extension, the Company shall be paid the escalable component of the Capacity Purchase Price applicable to the twenty-second (22nd) Agreement Year (as shall be further escalated pursuant to the provisions of Schedule 6) multiplied by the then-prevailing Dependable Capacity. During any such extensions, the Company shall also receive the Energy Purchase Price for each kWh of Net Electrical Output and any additional amounts payable, if any, due under the terms of this Agreement.

(c) Following the end of the twentieth (20th) Agreement Year, at the request of either Party, the Company and KESC agree to enter into good faith negotiations for a renewal of this Agreement for an additional term of five (5) Agreement Years on terms and conditions mutually agreed to by the Parties. If the Parties cannot agree to terms and conditions for the renewal of this Agreement, the Company will be permitted to contract with any other party for the sale of dependable capacity and electrical energy from the Complex and KESC shall deliver to the Company any necessary consents for such sale, including, without limitation, any consent required by Section 28 of the Electricity Act, 1910; <u>provided</u>, <u>however</u>, that KESC shall have no obligation to assist in such sale (including the transmission of electrical energy) unless otherwise required by law.

#### 4.2 <u>Company Events of Default - Termination by KESC</u>.

طرال

Each of the following events shall be events of default by the Company (each a "Company Event of Default"), which, if not cured within the time permitted (if any), shall give rise to the right on the part of KESC to terminate this Agreement pursuant to Section 4.4; provided, that KESC has received the prior written consent of the GOP, a copy of which consent shall be provided to the Company with any notice provided under Section 4.4, and provided further that no such event shall be a Company Event of Default (i) if it results from a breach by KESC of this Agreement or by the GOP of the Implementation Agreement or the Guarantee; or (ii) if it occurs as a result of or during an Force Majeure Event for the period provided pursuant to section 13.4:

24

(a) the failure of the Company to achieve the earlier to occur of either Construction Financial Closing or Financial Closing by the date twelve (12) months following the date of the Letter of Support, as such date may be extended pursuant to the terms of Section 19.6 of the Implementation Agreement;

(b) the failure of the Company (a) to achieve Construction Start and (b) to satisfy all conditions precedent to the initial availability of funds necessary to complete construction of the Complex (either under the Financing Documents or through equity commitments), each within ninety (90) Days after the earlier to occur of either Construction Financial Closing or Financial Closing;

(c) the failure of the Company to achieve the Commercial Operations Date within fifteen (15) months after the Required Commercial Operations Date;

(d) after the Construction Start, but prior to the achievement of the Commercial Operations Date, the abandonment of the Project by the Company or the failure of the Company to prosecute the Project in a diligent manner for a period of thirty (30) consecutive Days without the prior written notice to, and prior written consent of, KESC; <u>provided</u>, <u>however</u>, that after the commencement of on-site construction of the Complex, the Company shall not be deemed to have abandoned its construction of the Complex (and therefore its prosecution of the Project) so long as it is using all reasonable efforts to regain control of the Complex or reinstate such construction;

(e) after the Commercial Operations Date, the Abandonment by the Company of the operation of the Complex for a consecutive period of thirty (30)-Days without prior written notice to, and the prior written consent of KESC; provided, however, that the Company shall not be deemed to have Abandoned its operation of the Complex so long as it is using its best efforts to regain control of the Complex or reinstate such operation or is subject to a Lapse of Consent;

(f) the incurrence of cumulative operating losses (determined in accordance with generally accepted accounting principles in Pakistan, consistently applied, without taking into account any depreciation) in an amount greater than or equal to the sum of (i) fifty-five percent (55%) of the amount of the equity finds committed at Financial Closing and (ii) any additional equity contributed by the shareholders of the Company after Financial Closing;

(g) the Company's failure to operate, maintain, modify or repair the Complex in accordance with Prudent Utility Practices, such that safety of persons and property, the Complex or KESC's service to its customers is adversely affected;

(h) except for the transfer of the Complex to the GOP pursuant to the terms of the Implementation Agreement and the assignments to and by the Lenders contemplated under Section 17.10 of this Agreement, (i) the assignment or transfer of the Company's rights or obligations in the assets identified in Section 17.10(b) without the prior consent of KESC; or (ii) the transfer, conveyance, loss or relinquishment to any person or entity of the Company's right to own and/or operate the Complex or any material part thereof or to occupy the Site to any person without the prior written approval of KESC;

(i) except for the purpose of amalgamation or reconstruction (provided, that such amalgamation or reconstruction does not affect the ability of the amalgamated or reconstructed

entity, as the case may be, to perform its obligations under this Agreement), the occurrence of any of the following events; (i) the passing of a resolution by the shareholders of the Company for the winding up of the Company; (ii) the voluntary filing by the Company of a petition of bankruptcy, moratorium or other similar relief; (iii) the appointment of a provisional liquidator in a proceeding for the winding up of the Company after notice to the Company and due hearing, which appointment has not been set aside or stayed within ninety (90) Days of such appointment; or (iv) the making by a court with jurisdiction over the Company of an order winding up the Company which is not stayed or reversed by a court of competent authority within thirty (30) Days;

(j) any statement, representation or warranty by the Company in this Agreement proving to have been incorrect, in any material respect, when made or when deemed to have been made and such failure or incorrect statement, representation or warranty having a material and adverse effect on the Company's ability to perform its obligations under this Agreement;

(k) The exercise by Lenders of their remedies under the Financing Documents with respect to either the Complex assets or the pledged Ordinary Share Capital (as defined in the Implementation Agreement) such that either the Company or its management are removed by the Lenders from control of the Complex or the Company and the failure by the Lenders to deliver an Election Notice (as defined in Section 4.5) or to transfer the Complex and the rights and obligations of the Company under the Implementation Agreement and this Agreement to a Transferee (as this term is defined in Section 17.10(d)) within two hundred and forty (240) Days thereafter.

(1) any material breach by the Company of this Agreement that is not remedied within thirty (30) Days after notice from the GOP or KESC stating that a material breach of such agreement has occurred and is continuing that could result in the termination of such agreement, identifying the material breach in question in reasonable detail, and demanding remedy thereof, or a termination of the Implementation Agreement resulting from a default of the Company thereunder; g

(m) tampering by the Company, its employees, Contractors or subcontractors of any tier with the Interconnection Facilities or the Metering System; or

(n) failure to post security as required by Sections 9.4(f)(i)(A) or (B) or (C). The Company agrees that in no event shall the Company be entitled to any extension for the posting of security pursuant to Sections 9.4(f)(i)(A) or (B) or (C), including by reason of a Force Majeure Event pursuant to Article XIII; or

(o) any default or defaults by the Company in the making of any payment or payments required to be made by it hereunder within thirty-five (35) days of the due date therefor.

# 4.3 KESC Events of Default - Termination by the Company.

fle .

Each of the following events shall be events of default by KESC (each a "KESC Event of Default"), which if not cured within the time period permitted (if any) to cure, shall give rise to the right on the part of the Company to terminate this Agreement pursuant to Section 4.4, provided, however, that no such event shall be a KESC Event of Default (i) if it results from a breach by the Company of

26

this Agreement or the Implementation Agreement or (ii) if it occurs as a result of or during a Force. Majeure Event during the period provided pursuant to Section 13.4:

(a) the dissolution, pursuant to law, of KESC, except for (i) the privatization of KESC's thermal power stations or area electricity boards (provided, however, that in the case of the area electricity boards, all of KESC's obligations under this Agreement must be assigned pursuant to law to or contractually assumed, through a novation, by one or more entities, each of which has the legal capacity and appropriate commercial function to perform such obligations) or (ii) an amalgamation, reorganization, reconstruction, or further privatization of KESC, in each case where the GOP without interruption guarantees the performance of the succeeding entity or entities on the same terms and conditions as the Guarantee or such other commercial security is provided for the obligation of the succeeding entity or entities that in the reasonable business judgment of the Company provides an adequate alternative to the Guarantee and all of KESC's obligations under this Agreement are assigned pursuant to law to or contractually assumed, through a novation, by one or more entities, each of which has the legal capacity and appropriate commercial function to perform such obligations under this Agreement are assigned pursuant to law to or contractually assumed, through a novation by one or more entities, each of which has the legal capacity and appropriate commercial function to perform such obligations.

(b) any default or defaults by KESC in the making of any payment or payments required to be made by it within thirty (30) Days of the due date therefor and then, upon notice to the GOP, any default or defaults by the GOP in the making of any payment in accordance with the terms of the Guarantee which continues unpaid for five (5) Business Days;

(c) any statement, representation or warranty made by KESC herein proving to have been incorrect, in any respect, when made or when deemed to have been made and such failure or incorrect statement, representation or warranty having a material adverse effect on KESC's ability to perform its obligations under this Agreement; or

(d) any material breach by KESC of this Agreement, which is not remedied within thirty (30) Days after notice from the Company to KESC, which notice states that a material breach of such agreement has occurred that could result in the termination of such agreement, identifies the breach in question and demands remedy thereof; or

(e) failure of KESC to complete construction of the Interconnection Facilities and the Transmission Facilities by fifteen (15) Months after the Scheduled Commercial Operations Date;

(f) failure of KESC to post security as required by Section 9.4(f)(ii). KESC agrees that in no event shall KESC be entitled to any extension for the posting of security pursuant to Section 9.4(f)(ii), including by reason of a Force Majeure Event pursuant to Article XIII;

(g) failure of the Company to obtain and have in full force and effect the Consents specified in Schedule 1 of the Implementation Agreement as and within the six (6) Months period provided in Section 6.3(b) of the Implementation Agreement, so long as such failure is not a result of delay by the Company or the failure by the Company or its contractors to have been and to be in compliance with the substantive and procedural requirements applied in a "non-discriminatory" manner as described in Section 16.4 of the Implementation Agreement.

(h) any change in or change in the intrepretation of any applicable Laws of Pakistan in either case that continues in effect for ninety (90) days or more(including the Constitution of Pakistan and any other Law of Pakistan that gives effect to the injunctions of Islam) making (A) unenforceable, invalid, or void any material undertaking of KESC under this Agreement or (B) it unlawful for the Company, its Lenders or the Investors to make or receive any payment, to perform any obligation or to enjoy or enforce any material right under this Agreement, or (C) any such payment, the performance of any such material obligation or the enjoyment or enforcement of any such material right unenforceable, invalid or void as a result of any such change in law.

#### 4.4 Termination Notices; Termination.

(a) Upon occurrence of a KESC Event of Default or a Company Event of Default, as the case may be, that is not cured within the applicable period (if any) for cure, the nondefaulting Party may, at its option, initiate termination of this Agreement by delivering a written notice ("Notice of Intent to Terminate") of its intent to terminate this Agreement to the defaulting Party. The Notice of Intent to Terminate shall specify in reasonable detail the Company Event of Default or the KESC Event of Default, as the case may be, giving rise to the Notice of Intent to Terminate.

(b) Following the giving of a Notice of Intent to Terminate, the Parties shall consult for a period of up to forty-five (45) Days in case of a failure by either Party to make payments or provide security when due, and up to ninety (90) Days with respect to any other Event of Default (or such longer period as the Parties mutually may agree), as to what steps shall be taken with a view to mitigating the consequences of the relevant event taking into account all prevailing circumstances. During the period following delivery of the Notice of Intent to Terminate, the Party in default may continue to undertake efforts to cure the default, and if the default is cured at any time prior to the delivery of a Termination Notice in accordance with Section 4.4(c), then the non-defaulting Party shall have no right to terminate this Agreement in respect of such cured default.

(c) Subject to the provisions of Section 4.5 or 4.9, as the case may be, upon expiration of the consultation period described in Section 4.4(b) and unless the Parties shall have otherwise agreed or unless the Event of Default giving rise to the Notice of Intent to Terminate shall have been remedied, the Party having given the Notice of Intent to Terminate may terminate this Agreement by delivering a Termination Notice to the other Party, whereupon this Agreement shall immediately terminate.

(d) In the event that the Complex is transferred to the GOP under the terms of the Implementation Agreement, this Agreement shall immediately terminate and the Company shall have no further obligations to KESC hereunder except those obligations which arose prior to or upon the termination of this Agreement.

1th Ale

#### 4.5 Notice to Lender of Company's Default.

Anything in this Agreement notwithstanding, from and after the occurrence of the earlier to occur of either Construction Financial Closing or Financial Closing, KESC shall not seek to terminate this Agreement as the result of any default of the Company without first giving a copy of any notices required to be given to the Company under Sections 4.2 and 4.4 to the Lenders, such notice to be coupled with a request to the Lenders to cure any such default within the cure period specified in Section 4.4(b), and such cure period shall commence upon delivery of each such notice to the Lenders. If there is more than one Lender, the Lenders will designate in writing to KESC an agent (the "Agent") and any notice required hereunder shall be delivered to such Agent, such notice to be effective upon delivery to the Agent as if delivered to each of the Lenders. Each such notice shall be in writing and shall be deemed to have been delivered (a) when presented personally to the Lender or the Agent, (b) when transmitted by facsimile to the number specified in accordance with the procedure set forth below, or (c) five (5) Days after being deposited in a regularly maintained receptacle for the Postal Service in Pakistan, postage prepaid, registered or certified, return receipt requested, addressed to the Lender at the address indicated at Construction Financial Closing and Financial Closing (or such other address or to the Agent at such address as the Lenders may have specified by written notice delivered in accordance herewith). Any notice given by facsimile under this Section 4.5 shall be confirmed in writing delivered personally or sent by prepaid post, but failure to so confirm shall not void or invalidate the original notice if it is in fact received by the Lender or the Agent. Notwithstanding the foregoing, if the address of the Lender or Agent is outside Pakistan, any notice delivered to the Lender or Agent pursuant to this Section 4.5 shall be sent by international courier or facsimile, and if sent by facsimile, confirmed by international courier. The address and facsimile number for Lender or Agent shall be provided to KESC by the Company at Construction Financial Closing and Financial Closing and thereafter may be changed by the Lender or the Agent by subsequent delivery of a notice to KESC at the address or facsimile number for KESC provided in Section 16.1 (or at such other address or facsimile number subsequently delivered to the Lender or the Agent in accordance with this Section 4.5) and otherwise in accordance with the requirements of Section 16.1.

No rescission or termination of this Agreement by KESC shall be valid or binding upon the Lenders without such notice, the expiration of such cure period, and the expiration of the Extended Cure Period (as defined below) provided in this Section 4.5. The Lenders may make, but shall be under no obligation to make, any payment or perform any act required to be made or performed by the Company, with the same effect as if made or performed by the Company. If the Lenders fail to cure or are unable or unwilling to cure any Company Event of Default within the cure period under Section 4.4(b) as provided to the Company in this Agreement, KESC shall have all its rights and remedies with respect to such default as set forth in this Agreement; provided, however, that the Lenders, upon the termination of the cure period provided to the Company specified in Section 4.4(b), such cure period commencing on the delivery of such notice to the Lenders shall be offered a further period (the "Evaluation Period"), during which the Lenders shall evaluate such default, the condition of the Complex, and other matters relevant to the actions to be taken by the Lenders concerning such default, and which Evaluation Period shall end on the sooner to occur of (i) the Lenders' delivery to KESC of a notice that the Lenders have elected to pursue their remedies under the Financing Documents and assume the rights and obligations of the Company under this Agreement (an "Election Notice"), or (ii) thirty (30) Days following the end of the cure period. Upon the delivery of the Election Notice, the Lenders shall be granted an additional period of six (6) months to cure any Company Event of Default pet

29

11-

(the "Extended Cure Period") if required pursuant to Section 17.10(c). If KESC has assumed the operation of the Complex and the Complex is being operated by KESC to its satisfaction, KESC shall extend the Extended Cure Period for an additional period of six (6) months to cure any Company Event of Default if required pursuant to Section 17.10(c). In the event that the Lenders fail to cure any Company Event of Default required to be cured pursuant to Section 17.10(c) on or before the expiration of the Extended Cure Period, as it may have been extended, KESC may exercise its rights and remedies with respect to such default set forth in this Agreement, KESC may immediately terminate this Agreement, and such termination shall be effective on delivery to the Lenders or the Agent of notice of such termination.

### 4.6 Obligations Upon Termination.

Upon expiration or termination of this Agreement, the Parties shall have no further obligations hereunder except for obligations that arose prior to such expiration or termination and obligations that expressly survive such expiration or termination pursuant to this Agreement, including without limitation, the obligation to pay expenses under Section 4.7 and liquidated damages under Section 9.4.

4.7 <u>Reimbursement</u>.

In the event of a termination of this Agreement prior to the Commercial Operations Date for any reason other than (i) a KESC Event of Default; (ii) a GOP Event of Default under the Implementation Agreement; (iii) a Pakistan Political Force Majeure Event or (iv) a Change in Law, the Company shall reimburse KESC for all costs and expenses (including reasonable attorneys' fees) relating to the Project incurred by the KESC prior to such termination, which amount in any event shall not exceed One Hundred Thousand Dollars (\$100,000) plus all costs incurred by KESC on construction of the Transmission Facilities and the Interconnection Facilities. The amount of such construction costs shall be subject to independent audit, at the request and sole expense of the Company.

### 4.8 Other Remedies.

The exercise of the right of a Party to terminate this Agreement, as provided herein, does not preclude such Party from exercising other remedies that are provided herein or are available at law Remedies are cumulative, and the exercise of, or failure to exercise, one or more of them by a Party shall not limit or preclude the exercise of, or constitute a waiver of, other remedies by such Party; <u>provided</u>, <u>however</u>, that the remedies provided in Section 3.6 and Section 4.3 are the exclusive remedies available to the Company with respect to any claim by the Company against KESC for KESC's failure to construct the Interconnection Facilities or the Transmission Facilities in accordance with Section 7.4 and Schedule 3.

#### 4.9 Notice to the GOP of KESC's Default.

Anything in\_this Agreement notwithstanding, the Company shall not seek to terminate this Agreement as a result of any default of KESC without first giving a copy of any notices required to be given to KESC under Sections 4.3 and 4.4 to the GOP, such notices to be coupled with a request to the GOP to cure any such default within the same cure period as provided to KESC hereunder and

such cure period to commence upon delivery of each such notice to the GOP. Each such notice shall be deemed to have been delivered (a) when presented personally to the GOP, (b) when transmitted by facsimile, or (c) five (5) Days after being deposited in a regularly maintained receptacle for the Postal Service in Pakistan, postage prepaid, registered or certified, return receipt requested, addressed to the GOP, at the address indicated in Section 23.1 of the Implementation Agreement (or such other address as the GOP may have specified by written notice delivered in accordance therewith). No rescission or termination of this Agreement by the Company shall be of any effect without such notice and expiration of such cure period. The GOP may make or perform, but shall be under no obligation to make any payment (other than is required under the Guarantee) or to perform any act required of KESC hereunder with the same effect as if the payment or act had been made or performed by KESC. If the GOP fails to cure or is unable or unwilling to cure a default of KESC within the cure periods provided to KESC under this Agreement, the Company shall have all of its rights and remedies with respect to such default as set forth in this Agreement; provided, however, that if the GOP is diligently attempting to cure such default, other than a payment default of KESC, and demonstrable progress toward affecting such cure is being made, the GOP shall be granted an additional period not exceeding ninety (90) Days to affect such cure before the Company may exercise its rights and remedies with respect to such default set forth in this Agreement.

### 4.10 <u>Termination of the Implementation Agreement</u>

In the event that the Implementation Agreement is terminated under the provisions of Article XVII thereof and the Complex is transferred to GOP following a Pakistan Political Force Majeure Event or Change in Law or a GOP Event of Default thereunder, this Agreement shall be assigned to the GOP upon transfer of the Complex or, at the option of the GOP, terminate immediately.

lik

# ARTICLE V REPRESENTATIONS, WARRANTIES AND COVENANTS

#### 5.1 <u>Representations, Warranties and Covenants of the Company</u>.

Corporate Representations and Warranties. The Company represents and (a) warrants to KESC that (i) the Company is duly incorporated, existing and in good standing under the Laws of Pakistan, and has, so far as is material to KESC, complied fully with all requirements of the Companies Ordinance of 1984 and all other applicable Laws of Pakistan and has all requisite power and authority to conduct its business, to own its properties and to execute, to deliver and to perform its obligations under this Agreement; (ii) there are no proceedings pending, or to the best of its knowledge, threatened for the liquidation of the Company or that could materially adversely affect the performance by the Company of its obligations under this Agreement, the Implementation Agreement and any other documents comprising the Security Package, (iii) this Agreement has been duly authorized, executed and delivered by it and constitutes the legal, valid and binding obligation of it; and (iv) to the best of its knowledge, the execution and delivery of, and performance of its obligations under this Agreement by the Company, subject to the granting and maintenance of the requisite Consents, does not and, subject to the granting and maintenance of the relevant additional Consents in the future, will not constitute a violation of any statute, judgment, order, decree or regulation or rule of any court, governmental authority or arbitrator of competent jurisdiction applicable or relating to it, its Y assets or its businesses.

# (b) <u>Company Covenants</u>. The Company hereby covenants as follows:

Design and Construction of the Complex. The Company shall (i) design, engineer, finance, construct, complete and own the Complex in all material respects in accordance with (A) the Minimum Functional Specifications set forth in Schedule 1, (B) the plans and specifications and accompanying data submitted to KESC pursuant to this Agreement, (C) to the extent not inconsistent with the Laws of Pakistan, the Standards, (D) so far as it is material to KESC's rights and obligations hereunder, all applicable Laws of Pakistan and the Consents and (E) sound engineering and construction practices and Prudent Utility Practices and (F) such requirements as KESC may reasonably deem necessary in order for the Interconnection and Transmission Facilities to be designed and constructed in accordance with sound engineering practices, Prudent Electrical Practices and Prudent Utility Practices, and shall use all reasonable efforts to ensure that the Complex is Commissioned on or before the Required Commercial Operations Date. The Company covenants that the Complex will be designed, constructed and completed in a good and workmanlike manner, only with materials and equipment that are brand new and unused, utility grade and suitable for their intended use, and in such a manner as to provide that the useful life of the Complex, with proper operation and maintenance, will be at least equal to twenty-two (22) years and in accordance in all material respects with sound engineering practices and Prudent Utility Practices;

(ii) <u>Operation and Maintenance of the Complex</u>. The Company will operate and maintain the Complex in all material respects in accordance with (A) the operating procedures developed pursuant to Section 3.4, (B) the Technical Limits set forth in Schedule 2,

32

Mb

(C) if not inconsistent with the Laws of Pakistan, the Standards, (D), so far as it is material to KESC hereunder, all applicable Laws of Pakistan and the Consents, and (E) sound engineering practices and Prudent Utility Practices, including without limitation, synchronizing, voltage and reactive power control and (F) Prudent Electrical Practices.

(iii) <u>Company's Reliable Fuel Supply</u>. Beginning with the Commercial Operations Date and at all times thereafter until the termination of this Agreement, the Company will (i) have a reliable supply of fuel of quality and in quantity sufficient to meet the Net Electrical Output and Dependable Capacity delivery requirements hereunder and (ii) maintain at least a thirty (30) Day (at full load) supply of oil stored within one (1) mile of the Site. From time to time, as KESC may reasonably request, the Company shall provide KESC evidence of its compliance with this obligation. The Fuel Supply Agreement, the Company's inventory of fuel and availability of alternate supplies of fuel will be considered in determining whether the Company has a reliable supply of fuel;

(iv) <u>KESC Voltage Level or Waveforms</u>. Subject to the Technical Limits, the Company shall operate and maintain the Complex in such a manner so as not to have an adverse effect on KESC's voltage level or voltage waveform;

(v) <u>Complex Voltage Levels</u>. The Complex will be operated at the voltage levels determined pursuant to Schedule 2;

(vi) <u>Maintenance of Insurance</u>. The Company shall obtain and maintain the insurance coverage required by Article XI of this Agreement, in accordance with the provisions of Article XI;

(vii) <u>Applicable Laws</u>. The Company shall, (a) at all times, maintain its corporate existence in compliance with the Laws of Pakistan, (b) at all times, so far as it is material to KESC hereunder, comply with all Laws of Pakistan applicable to the Company and, to the extent not inconsistent therewith, the Standards, (c) use all reasonable efforts to procure and maintain all Consents necessary for its performance under this Agreement, give all required notices and allow all required inspections under all Consents obtained or applied for by it in connection with the Complex, and (d) pay all Prescribed Fees in connection with such Consents.

(viii) <u>Opinion of Counsel</u>. Upon request of KESC and at no cost to KESC, the Company shall cause its counsel to issue an opinion to KESC affirming the representations and covenants in Sections 5.1(a) and 5.1(b)(vii) and setting forth such further matters as KESC may reasonably request;

(ix) <u>Certificates of Officers, Accountants, Engineers or Agents.</u> The Company will, upon request of KESC, deliver or cause to be delivered from time to time to KESC certifications of its officers, accountants, engineers, or agents as to the performance of its obligations under this Agreement, including a certificate by the Engineer regarding, compliance of the Complex with the Standards, and as to such other matters as KESC may reasonably request; <u>provided</u>, <u>however</u>, that each certificate from such accountants, engineers or agents shall be requested not more than twice in any twelve (12) month period;

All

llh

Ħ

(x) <u>Maintenance of Books of Record and Account</u>. The Company will keep proper books of record and account, in sufficient detail to permit KESC to calculate and verify Pass-Through Items and Supplemental Tariff Charges recovered by the Company pursuant to Article IX and Schedule 6, in which all correct entries will be made of all dealings or transactions of or in relation to its business and affairs, in accordance with generally accepted accounting principles in Pakistan, consistently applied. The Company will furnish the following to KESC so long as this Agreement is in effect:

(A) <u>Periodic Reports</u>. As soon as available but in any event within sixty (60) Days of filing, furnish to KESC two (2) copies of all documents filed in compliance with the requirements of the Companies Ordinance, 1984, as amended or as may be superseded from time to time; and

(B) <u>Other Reports</u>. As soon as available a report on any factors materially and adversely affecting or that might materially and adversely affect the Project and its business and operations;

Examination. Without limiting the foregoing, the Company will (C)permit KESC or its accountants to examine all relevant books, records, reports and other papers of the Company for the purpose of verifying (i) the determination of Supplemental Tariff Payments; (ii) the amount or calculation of the Capacity Purchase Price or the Energy Purchase Price (including indexed and Pass-Through Items) in accordance with Schedule 6; and (iii) upon the occurrence of a Company Event of Default, compliance with the Company's financial obligations under the Financing Documents. The Company shall keep such books, records, reports and other papers in sufficient detail to permit KESC to calculate and evaluate such matters. Any such examination by KESC shall be made during regular business hours and upon reasonable advance notice to the Company. In connection with such examination, the Company will permit KESC and KESC's accountants to make copies and extracts of such books, records reports and other papers and to discuss such matters with the Company's officers and employees. If after discussion of such matters with the Company's officers and employees, KESC wishes to review such matters with the Company's accountants, the Company shall direct its accountants (whose fees and expenses shall be for the account of the Company) to communicate directly with KESC concerning such matters. The Company shall not be required to pay or reimburse KESC for expenses which KESC or KESC's accountants may incur in connection with any such examination;

(xi) <u>Maintenance Evaluations or Reports</u>: The Company shall provide KESC with copies of any maintenance evaluations or reports it performs for or obtains from any third party including those with a financial security interest in or lien on the Complex. The Company shall use all reasonable efforts to obtain for KESC copies of any evaluations or reports generated at the request of such third parties or performed by an engineer employed by such third party; and

(xii) <u>Protective Relays</u>. The Company shall, in accordance with the Minimum Functional Specifications, install protective relays in the Complex having ratings and characteristics approved by KESC, and, subject to the Technical Limits, shall maintain the

plb

settings of all such relays at the levels agreed by the Company and KESC and the Company shall not change such settings without the prior written consent of KESC.

# 5.2 <u>Representations, Warranties and Covenants of KESC</u>.

1

(a) <u>KESC Representations and Warranties</u>. KESC hereby represents and warrants that:

(i) Existence. Etc. It is duly incorporated under the Laws of Pakistan, and has, so far as it is material to the Company, complied fully with all applicable Laws of Pakistan, and that there are no proceedings pending, or to the best of its knowledge, threatened, for the dissolution of KESC or that would adversely affect the performance by KESC of its obligations under this Agreement; this Agreement has been duly authorized, executed and delivered by it and constitutes the legal, valid and binding obligation of it; and to the best of its knowledge, the execution and delivery of, and performance of its obligations under, this Agreement by KESC does not and, under the existing Laws of Pakistan, will not constitute a violation of any statute, judgment, order, decree or regulation or rule of any court, governmental authority or arbitrator of competent jurisdiction applicable or relating to it, its assets or its businesses; and

(ii) <u>KESC Approvals</u>. It has, as of the date hereof, such permits, licenses, and approvals required by all Public Sector Bntities with jurisdiction over KESC as are reasonably adequate to enable KESC to operate its business as presently operated, to own and operate the Grid System and to construct, own and operate the Interconnection Facilities and the Transmission Facilities.

(b) <u>KESC Covenants</u>. KESC hereby covenants as follows:

(i) <u>Interconnection and Transmission Facilities</u>. KESC shall design, construct, own, operate, and maintain the Interconnection Facilities and the Transmission Facilities substantially in accordance with Schedule 3 and the operating procedures developed pursuant to Section 3.4 and, so far as it is material to the Company's rights and obligations hereunder, all applicable Laws of Pakistan;

(ii) <u>No Adverse Effect on Complex</u>. KESC shall operate and maintain the Interconnection Facilities and the Transmission Facilities in such a manner so as not to have a material adverse effect on, and within the Technical Limits of, the Complex, and in accordance with Prudent Utility Practices, but KESC shall not be liable for any adverse effect on the Complex resulting from its normal operation and maintenance of the remainder of the Grid System; and

(iii) <u>Opinion of Counsel</u>. At the Construction Financial Closing and the Financial Closing, KESC shall deliver a certificate, executed by a duly authorized officer of KESC, affirming the representation in Section 5.2(a) and shall cause its counsel to issue an opinion affirming, to the best of its knowledge, the validity of the representations of KESC made in Section 5.2(a) and setting forth such further matters as the Lenders may reasonably request.

# ARTICLE VI CONTROL AND OPERATION OF THE COMPLEX, DESPATCH

### 6.1 Declared Available Capacity Declaration.

Prior to the beginning of each Operating Day, at the time and in the manner established in the operating procedures developed pursuant to Section 3.4, the Company shall inform the KESC Control Centre of the Declared Available Capacity for the coming Operating Day of the Complex.

### 6.2 Scheduling and Despatch.

116

(a) In order to assist with scheduling of the Complex to meet the requirements of KESC, the Parties agree that, from and after ninety (90) Days prior to the Scheduled Commercial Operations Date, the following procedures will be adhered to:

(i) Year Ahead Notification. Not less than ninety (90) Days before the Scheduled Commercial Operations Date, and thereafter not less than ninety (90) Days before the beginning of each Year, KESC shall provide to the Company estimated requirements on a Monthly basis for Net Electrical Output for the remainder of the Year in which the Commercial Operations Date is scheduled to occur, and thereafter for each subsequent Year, but KESC shall not be bound by these figures. When providing such year ahead notification, KESC shall have the right to revise what Months shall be deemed Maintenance Months; provided, however, that such revision by KESC shall only be made in connection with long range planning of load requirements; provided, further, that any such revision shall also include a revision to the Weighting Factors such that the Weighting Factors associated with the Maintenance Months before and after such revisions shall be unchanged. The year ahead notification for the first full Year following the Commercial Operations Date and each subsequent Year shall also include a designation of a Demonstration Period of not more than ninety (90) consecutive days for the Year to which such notification applies.

(ii) <u>Quarter Ahead Notification</u>. Not less than sixty (60) Days before the beginning of each calendar quarter KESC shall provide to the Company estimated requirements on a Week-by-Week basis for Net Electrical Output during that quarter. KESC shall also provide to the Company revised estimates for the remaining quarters for the current year. KESC shall not be bound by these figures.

(iii) <u>Month Ahead Notification</u>. Not less than fourteen (14) Days before the beginning of each Month, KESC shall provide to the Company estimated requirements on a Day-by-Day basis for Net Electrical Output during the Month and also, provisionally, for the following two (2) Months, but KESC shall not be bound by these figures.

(iv) <u>Week Alread Notification</u>. Not less than forty eight (48) hours before the beginning\_of each Week KESC shall provide to the Company\_estimated requirements, on an hour by hour basis, for Net Electrical Output during that Week and also, provisionally, during the following Week, but KESC shall not be bound by these figures.

36

(v) <u>Plant Availability Notification</u>. To enable KESC to give final schedules of requirements as in (vi) below the Company shall, by 1200 hours each Day, inform KESC of the estimated Declared Available Capacity of the Complex available during each hour of the Day commencing thirty-six (36) hours ahead and, provisionally, for the Day immediately thereafter. Such estimates shall not be binding upon the Company and the Company shall advise KESC as soon as possible of any changes in its Declared Available Capacity for such Days.

F

1

W

(vi) <u>Day Ahead Notification</u>. Not less than seven (7) hours before the start of each Day, KESC shall provide to the Company firm requirements, on an hour by hour basis, for Net Electrical Output, Start-Ups, and Reactive Power during that Day and also, provisionally, during the following Day. The firm requirements shall be not be binding upon KESC and KESC may subsequently alter its requirements.

Actual operation level requested by KESC will be determined by the requirements for operation in accordance with Economic Despatch and/or AGC and may be substantially different from the information provided in accordance with this Section; <u>provided</u>, <u>however</u>, that actual operation levels requested by KESC shall at all times be subject to compliance with the Technical Limits.

(b) Notice of Change of Operating Levels. In connection with its rights to Despatch the Complex in accordance with this Agreement, KESC will provide the Company with at least five (5) minutes advance notice of changes in operating levels to be achieved by the Complex (or such greater period as may be required by the Technical Limits) except that when the Complex is operated with AGC, KESC shall not be required to provide such notice but KESC shall observe the Technical Limits in effecting the changes in operating levels.

(c) <u>Compliance</u>. The Company agrees to comply with the instructions it receives from KESC pursuant to subsections (a) and (b) of this Section 6.2. In the event that the Company is unable to comply with any such instructions, KESC's sole remedy shall be the collection of damages, if any, payable in accordance with Article IX.

(d) <u>Adjustment to Fuel Requirements Arising From Despatch</u>. If as a result of a change or changes in KESC's Despatch levels from those identified in the quarter ahead notification delivered by KESC to the Company pursuant to Section 6.2(a)(ii) as modified through the notice given forty-five (45) Days before the scheduled delivery date of fuel at Site, the Company has to change any scheduled fuel deliveries from the Fuel Supplier, and

(i) If the cumulative effect on the Company of such change or changes in Despatch by KESC is a reduction of more than thirty-five percent (35%) in the actual Monthly fuel deliveries from the Monthly scheduled fuel deliveries calculated according to the quarter ahead notification of the estimated requirements for Net Electrical Output delivered by KESC pursuant to Section 6.2(a)(ii), as may be modified through the notice given forty-five (45) Days before the scheduled delivery date of fuel at Site, damages charged to the Company under the Fuel Supply Agreement as a result of such cumulative effect of such change or changes, which shall be based on the actual costs incurred by the Fuel Supplier as a result of such cumulative effect of such change or changes, shall be a Pass-Through Item under Section 14 of Schedule 6 of this Agreement as follows:

37

// \_\_\_\_

(A) In case KESC's Despatch level of the Complex is changed such that Company has to amend the quarterly order with a cumulative effect between thirty-five percent (35%) and forty-five percent (45%) lower than such Monthly figure with less than forty-five (45) Days' notice prior to scheduled delivery date of fuel at Site, the Company shall be obliged to take any consignment of fuel whether by pipeline, rail or road already in transit for the Complex, failing which the Company shall pay to the Fuel Supplier any freight charges for the consignment of Fuel to the Complex or the terminal, railway demurrage charges, road halting charges and diversion charges therefrom to the decantation point at the facility of the alternative buyer to the extent incurred by the Fuel Supplier; and

(B) In case KESC's Despatch level of the Complex is changed such that the Company has to amend the order and the cumulative effect of such an amendment falls by more than forty-five percent (45%) below the monthly figure as modified through the notice given forty-five (45) Days before the scheduled delivery date of fuel at Site, the Company shall pay as damages, in US Dollars to the Fuel Supplier, the tanker demurrage charges to the extent incurred by the Fuel Supplier as a result of such effect of such amendment; provided, that the above charges shall be subject to third party audit at KESC's option.

The Company, through management of its fuel inventory, shall use all reasonable efforts to avoid any changes in its scheduled fuel deliveries resulting from reductions in KESC's Despatch requests. The Company shall notify KESC of any cancellation or rescheduling of fuel deliveries at the time of such cancellation or rescheduling.

(ii) If the cumulative change in Despatch levels over the Despatch levels in the quarter ahead Notification, as modified through the notice given forty-five (45) Days before the scheduled delivery date of the fuel at Site, is more than ten percent (10%) higher than the scheduled level for despatch for such month, the Company shall use its best efforts to meet such amended Despatch request; <u>provided</u>, <u>however</u>, that failure by the Company to meet fully Despatch requests in the first two (2) weeks of the following Month resulting solely from such amended Despatch requests shall be a Force Majeure Event or Events under Section 13.1.

# 6.3 Scheduled Outage Periods.

1110-

plb

Within forty-five (45) Days following receipt by the Company of the Year (a)ahead notification to be provided by KESC pursuant to Section 6.2(a)(i), the Company shall submit to KESC its proposed schedule of Scheduled Outage periods for the Year to which such notification applies. Such Scheduled Outage periods shall, in the aggregate, not exceed thirty (30) Days in each Agreement Year (computed by multiplying thirty (30) Days by twenty-four (24) hours and by the thenprevailing Dependable Capacity, expressed in MWh), except that every fifth (5th) Agreement Year, an additional thirty (30)Scheduled Outage Days (computed by multiplying thirty (30) Days by twenty-four (24) hours and by the then-prevailing Dependable Capacity, expressed in MWh), may be scheduled. The Company shall only propose Scheduled Outage periods of the Complex during the Maintenance Months, subject, however, to the requirements of the Technical Limits relating to equipment. maintenance; provided, however, that in no event shall any Scheduled Outage be scheduled during the for the purpose of conducting except October, Months of April, May, June, and

inspection, testing and preventive maintenance not exceeding seven (7) Days in the aggregate during such months (computed by multiplying seven (7) Days by twenty-four (24) hours and by the thenprevailing Dependable Capacity, expressed in Mwh); <u>provided</u>, <u>further</u>, that not more than one generating unit shall be subject to inspection, testing and preventive maintenance at a time during such period; <u>provided</u>, <u>further</u>, that KESC may, at the time of its Year ahead notification to the Company, designate any other three consecutive Months and up to four (4) Months total for the Year to which such notification applies as Months during which no such Scheduled Outage shall be allowed. The Company shall take into consideration the Demonstration Period designated by KESC in the Year ahead notification when establishing the proposed schedule of Scheduled Outage periods; <u>provided</u>, <u>however</u>, that if the requirements of the Technical Limits or the scheduling of Scheduled Outages during Maintenance Months would result in a Scheduled Outage during the designated Demonstration Period, then the Company shall propose an alternative Demonstration Period at the time of the submission by the Company of its proposed Scheduled Outage periods.

(b) Within thirty (30) Days after receipt by KESC from the Company of the Company's submittal of requested Scheduled Outage periods, KESC shall notify the Company in writing whether the requested Scheduled Outage periods are acceptable. If KESC cannot accept any of the requested Scheduled Outage periods, KESC shall advise the Company of a period during the Maintenance Months when KESC determines such unacceptable Scheduled Outage period can be rescheduled. Such rescheduled period shall be as close as reasonably practicable to the requested period, shall comply with the Technical Limits, and shall be of equal duration as the requested period. Based upon the foregoing, KESC shall include in its notice to the Company pursuant to this Section 6.3(b) a final designation for the Demonstration Period for the Year to which the schedule for Scheduled Outage.

(c) The Company shall undertake Scheduled Outages only as agreed to in writing by KESC as aforesaid.

(d) KESC may, upon sixty (60) Days prior written notice, request the Company to reschedule a Scheduled Outage, <u>provided</u>, <u>however</u>, that KESC shall not request that such Scheduled Outage be rescheduled in a manner or time inconsistent with the Technical Limits or Prudent Utility Practices. The Company shall use reasonable efforts to comply with such request and shall notify KESC within fifteen (15) Days of receipt of KESC's request whether the Company is able to comply with such request and, if it is unable to comply, giving the reasons therefor and the periods in which the Scheduled Outage may be rescheduled. In such event, the Parties shall discuss in good faith an alternate period in which the Scheduled Outage may be conducted, but if they are unable to reach agreement within 15 days, the Scheduled Outage shall be rescheduled by KESC during an alternate period provided by the Company in its reply to KESC's request for an alternate period.

(e) As the need arises for a Maintenance Outage, the Company shall advise KESC of such need and of the commencement and estimated duration of the works to be carried out and KESC shall allow the Company to schedule such Maintenance Outage at a time and within a period of time reasonable in light-of its need for energy from the Complex and the necessity-of the Maintenance Outage.

Ħ

(f) For each MWh that the Company reduces the aggregate Scheduled Outage periods below seven-hundred and twenty (720) hours multiplied by the then-prevailing Dependable Capacity in each Agreement Year, the  $\Omega$  (omega) provided in Section 9.4(b) shall, for purposes of computing liquidated damages payable by the Company, be increased by 0.5 MWh.

### 6.4 Operation in Accordance with Despatch.

From and after the Commercial Operations Date, the Company shall operate the Complex consistent with KESC's Despatch of the Complex. The Company shall comply with all Despatch instructions received from KESC in accordance with Section 6.6. In the event that KESC from time to time elects to place the Complex under the control of AGC, KESC shall advise the Company of the periodic estimated requirements from the Complex and the Company shall schedule the operation of the Complex to meet the said estimated requirements.

### 6.5 <u>Recording of Telephoned Communications.</u>

Each Party hereby authorizes the other Party to tape record all telephoned voice communications relating to Declared Available Capacity control and Despatch of the Complex received from the other Party pursuant to this Agreement and shall supply, at the request of the other Party, a copy or transcript of any such recording.

j,

### 6.6 Emergency Set-Up and Curtailment Plans.

The Company shall cooperate with KESC in developing emergency procedures for the Complex, including, without limitation, recovery from a local or widespread electrical blackout and voltage reduction in order to effect load curtailment, and shall, within the Technical Limits, comply with such emergency procedures. The Company shall make technical references available to KESC concerning Start-Up times, black-start capabilities, and minimum load-carrying ability.

# 6.7 Supply of Power in Emergency.

Su-

M

The Company shall, during an Emergency and upon request by KESC, supply such power as the Complex is able to generate within the Technical Limits and KESC is able to receive; <u>provided</u>, <u>however</u>, that prior to the Commercial Operations Date the Company shall not be obligated to supply such power if doing so would materially delay the completion of construction, testing or Commissioning for the Complex, <u>provided</u>, that the Company shall use all reasonable efforts to comply with any such request by KESC prior to the Commercial Operations Date without materially delaying construction or the testing of the Complex or meeting the Premium Date. If the Complex has a Scheduled Outage or a Maintenance Outage, and such Scheduled Outage or Maintenance Outage occurs or would occur coincident with an Emergency, the Company shall make all reasonable efforts to reschedule the Scheduled Outage or Maintenance Outage or, if the Scheduled Outage or Maintenance Outage has begun, expedite the completion of the work to restore power supply as soon as possible. Nothing contained in this Agreement or the Schedules hereto shall be construed to require the Company to operate the Complex at any time, including during an Emergency, in any manner\_ inconsistent with the Technical Limits.

# 6.8 Employment of Qualified Personnel.

The Company shall only employ personnel (management, supervisory and otherwise), directly or indirectly through its O&M Contractor, who are adequately qualified and trained, and who have experience as necessary and appropriate for operating, maintaining, and monitoring the Complex and for coordinating operations of the Complex with the Grid System. The Company and KESC shall ensure that their personnel are on duty at the Complex and the Control Centre, respectively, at all times, twenty-four (24) hours a Day and seven (7) Days a Week commencing with the first date on which electrical energy is generated.

# 6.9 Operating Committee Membership and Duties.

(a) On or before twelve (12) Months prior to the Scheduled Commercial Operations Date, the Parties shall establish an Operating Committee comprising six (6) members. Each Party shall designate three (3) members to represent it on the Operating Committee, and either Party may remove or replace any of its Operating Committee members at any time upon notice to the other Party. The Operating Committee shall develop procedures for the holding of meetings, the keeping of minutes of meetings and the appointment and operation of sub-committees. The chairmanship of the Operating Committee shall rotate each six (6) months between the Parties and the Parties agree that the first chairman shall be nominated by KESC. Decisions of the Operating Committee shall require the approval of a majority of members of the Operating Committee.

(b) The Operating Committee shall be responsible for finalizing the operating procedures to be developed by the Company pursuant to Section 3.4; for approving procedures for the Commissioning of the Complex pursuant to Article X for establishing other procedures relating to the interaction of the Complex, the Metering System, the Interconnection Facilities, the Transmission Facilities and the Grid System; and, where appropriate, for proposing solutions to other issues and attempting to resolve Disputes arising under this Agreement. These matters shall include:

(i) the coordination of the respective programmes and procedures of the Parties for the construction, commissioning and operation of the Transmission Facilities and Interconnection Facilities, and the Metering System and the Complex, and agreement where necessary upon the respective commissioning procedures;

(ii) the discussion of the steps to be taken on the occurrence of any Force Majeure Event or the shutdown or reduction in capacity for any other reason of the Interconnection Facilities, the Transmission Facilities or the Complex;

(iii) the coordination of Scheduled Outages;

Contractors;

(iv) safety matters affecting the Complex, the Parties or their

(v) clarification of emergency plans developed by KESC for recovery from a local or widespread electrical blackout;

No Ale.

(vi) review and revision, subject to KESC approval, of protection schemes; and

(vii) any other matter mutually agreed to by the Parties.

(c) The Parties shall instruct their representatives on the Operating Committee to act in good faith in dealing with matters considered by the Operating Committee. The Parties shall consider and use reasonable efforts to incorporate decisions of the Operating Committee. Actions taken in conformance with the decisions of the Operating Committee shall be presumed to be in compliance with the terms of this Agreement.

(d) The Operating Committee shall use reasonable efforts to resolve Disputes, where appropriate, in accordance with the procedures established in Section 15.2

6.10 Maintenance of Operating Records.

(a) Each Party shall keep complete and accurate records and all other data required by each of them for the purposes of proper administration of this Agreement. Among other records and data required hereby or elsewhere in this Agreement, the Company shall maintain an accurate and up-to-date operating log, in a format reasonably acceptable to KESC, at the Complex with records of:

(i) real and Reactive Power production for each clock hour, and 132 kV bus voltage at all times;

(ii) changes in operating status, Scheduled Outages, Maintenance Outages, Forced Outages and Partial Forced Outages;

(iii) any unusual conditions found during inspections; and

(iv) other matters agreed to by the Operating Committee.

All such records and data shall be maintained for a minimum of sixty (60) Months after the creation of such record or data and for any additional length of time required by Public Sector Entities with jurisdiction over either Party; <u>provided</u>, <u>however</u>, that each Party shall not dispose of or destroy any such records after such sixty (60) Month period unless the Party desiring to dispose of or destroy any such records gives thirty (30) Days prior written notice to the other Party, generally describing the records to be destroyed or disposed of, and the Party receiving such notice does not object thereto in writing within ten (10) Days.

(b) Either Party shall have the right, upon ten (10) Days prior written notice to the other Party, to examine the records and data of the other Party relating to this Agreement or the operation and Despatch of the Complex within the Grid System at any time during normal office hours during the period such records and data are required hereunder to be maintained.

Hu.

flb

### 6.11 <u>Cessation of Operation of the Complex</u>.

If, after the Commercial Operations Date, without the prior written consent of KESC, the Company shall have ceased to operate the Complex for a period of forty-eight (48) consecutive hours other than because of a Force Majeure Event, a Forced Outage, a Maintenance Outage, a Scheduled Outage or as a result of Despatch instructions from KESC, then KESC shall be (i) entitled to enter the Complex and operate it until the Company demonstrates to the reasonable satisfaction of KESC that it can and will resume normal operation of the Complex, and (ii) as soon as is practicable, send written notice of such entry to the Agent (as defined in Section 4.5) in accordance with the procedure set forth in Section 4.5. During any period that KESC shall operate the Complex pursuant to this Section, KESC shall bear all costs of such operation (including fuel costs), operate the Complex within the Technical Limits, and shall continue to pay to the Company the non-escalable portion of the Capacity Payments as the Company would otherwise be entitled to during such period. Notwithstanding any other provision in this Agreement to the Complex incurred, suffered or sustained by the Company barmless from any loss or damage to the Complex incurred, suffered or sustained by the Company barmless of KESC's negligence or willful misconduct in the operation of the Complex during such period, but only to the extent that such loss or damage is not covered by insurance.

Ku

# ARTICLE VII INTERCONNECTION

#### 7.1 Interconnection and Transmission Facilities.

KESC shall be responsible for the design, construction, installation (excluding the Metering System, as provided in Section 8.2), commissioning, operation and maintenance of the Interconnection Facilities and the Transmission Facilities in accordance with the terms of this Agreement, and KESC shall own all of such Interconnection Facilities and Transmission Facilities.

#### 7.2 Data Necessary for Construction of Interconnection and Transmission Facilities.

KESC acknowledges that the Company has provided KESC with the information required in Item 2 of Schedule 3. Based upon this information, KESC has determined that the Interconnection Facilities and the Transmission Facilities can be designed and constructed within the time required by Section 7.4 of this Agreement. Within ten (10) Days of a request by KESC, the Company shall provide all additional information reasonably requested by KESC in connection with its completion of the Interconnection Facilities and Transmission Facilities. KESC shall utilize such supplemented information in its final design of the Interconnection Facilities and Transmission Facilities. The timely provision by the Company of such supplemental or additional information shall not modify the obligation of KESC to construct the Interconnection Facilities and the Transmission Facilities as required herein.

# 7.3 Granting of Easements and Rights of Way.

HK\_

Mb

The Company shall grant to KESC permanent easements and rights-of-way (a)across the Site necessary to install, operate, maintain, replace and/or remove the Interconnection Facilities and the Transmission Facilities. The easements shall grant to KESC adequate and continuing rights for the purposes set forth in Section 7.1 to enter the Site subject only to KESC's giving prior notice to the Company. Upon request by KESC, the Company shall execute such easements, rights-ofway, licenses and other documents, each in recordable form, as KESC may reasonably require to record any and all of the above rights. Consideration for such rights shall be the execution of this Agreement and no other consideration shall be required. Insofar as it shall be consistent with the Laws of Pakistan, all easements, rights-of-way, licenses and other rights hereunder shall survive termination or expiration of this Agreement. Revocable licenses, if any, granted to KESC pursuant to this Section 7.3 shall include such reasonable further term, not to exceed ninety (90) Days beyond the Term, to allow KESC to remove the Interconnection Facilities and the Transmission Facilities. When on the Site, KESC shall comply with all reasonable instructions of the Company and its Contractors relating to the carrying out of any work on the Site and, notwithstanding any other provision in this Agreement to the contrary, shall indemnify and hold the Company and the Contractors harmless from any loss or damage sustained by virtue of KESC's negligence or willful misconduct in the exercise of rights pursuant to this Section 7.3, but only to the extent that such loss or damage is not covered by insurance. 1-

(b) Except as provided in Section 7.3(a), KESC, at its expense, shall be responsible for obtaining all rights-of-way, easements and other real or personal property interests

necessary to construct, operate and maintain the Transmission Facilities and the Interconnection Facilities during the term of this Agreement.

# 7.4 Construction of Interconnection and Transmission Facilities.

The Company shall give KESC thirteen (13) months prior written notice of the Scheduled Commercial Operations Date then anticipated by the Company. Following the receipt of such notice, KESC shall commence the final design of the Interconnection Facilities and Transmission Facilities. Thereafter, KESC shall give the Company reports on the progress of the design and construction of the Interconnection Facilities and the Transmission Facilities as appropriate until the same are completed. KESC shall, at its sole expense except as provided in Section 7.3(a), construct and test the Interconnection Facilities and Transmission Facilities no later than ninety (90) Days prior to the Scheduled Commercial Operations Date provided to KESC pursuant to the first sentence of this paragraph; provided, however, that such completion date shall be extended on a Day-for-Day basis for any changes in the Scheduled Commercial Operations Date and to the extent necessary because of the occurrence of any of the following:

(a) the failure by the Company to execute, in sufficient time for KESC to complete the Interconnection Facilities and the Transmission Facilities, such easements, rights-of-way, licenses and other documents, each in recordable form, as KESC may reasonably require to record the deeds, easements, rights-of-way and licenses granted pursuant to Section 7.3;

(b) the failure by the Company to provide KESC, on a timely basis, with anytechnical data not included in Schedule 3 available to the Company and requested by KESC relating to the Complex reasonably necessary for KESC to undertake the design, construction, installation, commissioning, maintenance and operation of the Interconnection Facilities or the Transmission Facilities;

(c) an Emergency;

Ho

(d) a Force Majeure Event under Section 13.1(c)(i)(E); or

(e) any other failure by the Company to perform in accordance with this Agreement that materially affects KESC's ability to perform its obligations in accordance with this Article VII;

provided, however, that no extension shall be granted to KESC to the extent that such failure or delay would have nevertheless been experienced by KESC. At any time after it has given KESC notice of the anticipated Scheduled Commercial Operations Date, the Company may request KESC in writing to accelerate the design and construction of the Interconnection Facilities and/or the Transmission Facilities at the Company's expense, and KESC shall comply with such a request if at all possible. Within fifteen (15) Days of receipt of such a request, the Parties shall meet and KESC shall advise the Company of whether such acceleration is possible and, if so, the number of Days to be gained, the incremental cost of such acceleration, any options that may be available, and KESC's suggested payment schedule therefor. Upon receiving the Company's commitment in writing to pay for a programme of acceleration identified at the meeting, KESC shall promptly undertake such programme.

### 7.5 Interconnection Equipment on Company's Side of Interconnection Point.

Subject to Sections 7.6 and 8.2, the Company shall be responsible, at Company's expense, for designing, constructing, installing and maintaining all auxiliary and interconnecting equipment on the Company's side of the Interconnection Point, and, except for the Metering System, the Company shall own all such auxiliary and interconnection equipment.

#### 7.6 Protective Devices.

(a) Subject to giving the Company reasonable notice, KESC may require the Company to modify or expand the requirements for protective devices. Following approval by KESC of the costs of such modification or expansion, the Company shall perform such modification or expansion and, for costs incurred for modifications or expansions requested after the Commercial Operations Date, KESC shall reimburse the Company for the reasonable costs of such modification or expansion.

(b) Together with an invoice for reimbursement, the Company shall provide reasonable documentation of the expenses incurred in modifying or expanding the protective devices. Payments shall be due thirty (30) Days after delivery of the invoice by the Company. In case of any Dispute, the provisions of Section 9.8 shall apply. Such work shall be completed within a reasonable time under the circumstances. KESC shall be notified in advance of, and shall have the right to observe, all work on the protective devices.

(c) Each Party shall notify the other Party in advance of any changes to either the Complex or the Grid System that may affect the proper coordination of protective devices between the two systems. Such changes shall not be made without KESC's approval.

7.7 <u>Testing</u>.

The Parties shall cooperate in testing the Interconnection Facilities and the Transmission Facilities in accordance with the schedule developed by the Operating Committee (but in no event later than the time provided in Section 7.4) and at such other times thereafter as either Party may reasonably require.

### ARTICLE VIII METERING & TELECOMMUNICATIONS

# 8.1 Metering System.

(a) KESC shall, at its expense, procure, own and maintain the Metering System. The Company, at its expense, shall procure, install, own and maintain the Back-Up Metering System.

(b) KESC shall provide and install a strip chart recorder and shall make a continuous recording of the Net Electrical Output of the Complex. Such Net Electrical Output shall be recorded on appropriate magnetic media or equivalent, which recording shall be used to compute penalties under Section 9.4(c).

# 8.2 Installation of Metering Systems.

Subject to Section 8.2(b), the Company, at its expense, shall install the (a) Metering System on the Site at a location to be agreed upon by the Parties, and upon completion, convey the Metering System to KESC. Prior to the installation by the Company or the Contractor of the Metering System equipment, the Company will deliver to KESC the protection scheme and the metering plan for the Complex for KESC's approval. KESC will provide written comments to the protection scheme and the metering plan within thirty (30) Days of their receipt. The Company will incorporate KESC's comments received during such thirty (30) Day period into the protection scheme and the metering plan and deliver final copies to KESC. KESC will approve the final scheme and plan within fifteen (15) Days or notify the Company that it does not approve the scheme and plan, giving its reasons therefor. If KESC does not give reasons for not approving the scheme and plan within such fifteen (15) Day period, KESC shall be deemed to have approved such scheme and plan. Upon approval by KESC, the Company will complete the design and commence construction of the Metering System. Such installation shall be completed not later than fifteen (15) Days prior to the scheduled date to begin initial testing of the Complex. The Company shall provide KESC with sixty (60) Days advance notice of, and KESC shall have the right to observe and inspect, the installation of the Metering System. KESC shall be notified not less than fifteen (15) Days prior to, and shall have the right to observe, the installation of the Back-up Metering System by the Company.

(b) If the Metering System is not provided to the Company by KESC at a reasonable time taking into account the construction schedule, then KESC shall reimburse the Company for all reasonable expenses incurred by the Company for the acquisition of the Metering System. Together with an invoice for reimbursement, the Company shall provide reasonable documentation of the expenses incurred for the purchase of the Metering System. Payment shall be due along with the first scheduled payment made pursuant to Section 9.7, provided, in case of any Dispute, the provisions of Section 9.8 shall apply.

8.3 <u>Testing of Metering System</u>.

Nb

(a) KESC shall initially test the Metering System for accuracy in accordance with Schedule 5 by the later of fifteen (15) Days after it is installed by the Company or the date scheduled

for initial testing of the Complex to begin, and thereafter at intervals of not less than one hundred and eighty (180) Days after giving the Company no less than forty-eight (48) hours advance notice. The Company may have a representative present during any such testing, as well as during any inspection of the Metering System or adjustment thereof.

(b) KESC shall also test the Metering System at any other time reasonably requested by the Company, such additional testing to be at the Company's expense unless the test indicates that the Metering System is inaccurate by more than one-half percent (0.5%), in which case KESC shall bear the cost of the additional test. The Company may have a representative present during any such testing, as well as during any inspection of the Metering System or adjustment thereof.

(c) When on the Site, KESC shall comply with all reasonable instructions of the Company and the Contractors and, notwithstanding any other provision in this Agreement to the contrary, shall indemnify and hold the Company and the Contractors harmless from any loss or damage sustained by virtue of KESC's negligence (except in the case of an error made in good faith) or willful misconduct in the performance of its obligations pursuant to Section 8.3(a) and (b), but only to the extent that such loss or damage is not covered by insurance.

8.4 <u>Reading of Meters</u>.

llb

Alber-

Procedures. The Metering System shall be read Monthly on the last Business (a) Day of each Month (or such other Day as may be mutually agreed upon by the Parties) for the purpose of determining the Net Electrical Output of the Complex since the preceding reading. The Company shall read the Metering System during normal business hours, and unless otherwise mutually agreed by the Parties, shall give KESC at least forty-eight (48) hours notice of the time the Company shall read the Metering System. KESC shall have the right to have a representative present during any such reading. In the event that a KESC representative is present at such reading of the Metering System for the purpose of measuring Net Electrical Output, then such reading shall be jointly taken and recorded. All measurements of Dependable Capacity shall be jointly taken and recorded. In the event that a KESC representative is not present at a reading of Net Electrical Output, then the Company representative shall take and record such reading and make a photographic record thereof. The Company shall maintain a log of all such meter readings. Measurements recorded shall be delivered by the recording Party to the non-recording Party through facsimile within forty-eight (48) hours after the readings are taken. In the event that the Metering System is not in service as a result of maintenance, repairs or testing, then the best available information, which may include the Back-Up Metering System, shall be used during the period that the Metering System is not in service and the foregoing provisions of this Section 8.4(a) shall apply to the reading of the Back-Up Metering System.

1

2

(b) <u>Inaccuracies in Metering System</u>. When, as a result of any test pursuant to Section 8.3, the Metering System is found to be inaccurate by more than one-half percent (0.5%) or is otherwise functioning improperly, then the correct amount of Net Electrical Output delivered to KESC for the actual period during which inaccurate measurements were made, if any, shall be determined as follows:

48

(i) First, the readings of the Back-Up Metering System, if any, may be utilized to calculate the correct amount of Net Electrical Output, unless a test of such Back-Up

Metering System, as required by either Party, reveals that the Back-Up Metering System is inaccurate by more than one-half percent (0.5%) or is otherwise functioning improperly;

(ii) If there is no Back-Up Metering System or if the Back-Up Metering System is found to be inaccurate by more than one-half percent (0.5%) or is otherwise functioning improperly, then the Company and KESC shall jointly prepare an estimate of the correct reading on the basis of all available information and such guidelines as may have been agreed to between the Company and KESC;

(iii) In the event that KESC and the Company fail to agree upon an estimate for the correct reading, KESC shall make any payments to the Company required as a result of its estimate of the correct reading and the matter may be referred by either Party for determination by an expert pursuant to Article XV; and

(iv) The difference between the previous payments by KESC for the period of inaccuracy and the recalculated amount shall be offset against or added to the next payment to the Company under this Agreement, as appropriate. If the period of inaccuracy cannot be accurately determined, it shall be deemed to have begun on the date which is midway between the date the meter was found to be inaccurate and the date of the last meter reading accepted by the Parties as accurate. In no event, however, shall any such adjustment be made for any period prior to the date on which the Metering System was last tested and found to be accurate within plus or minus one-half percent (0.5%) and not otherwise functioning improperly.

#### 8.5 Sealing of Metering System.

The Metering System and the Back-Up Metering System shall comply with the specifications in Schedule 5 and shall be jointly sealed. Such seals shall be broken only by KESC personnel. The Company shall be given at least twenty-four (24) hours advance notice of the breaking of seals on the Metering System or the Back-Up Metering System; <u>provided</u>, <u>however</u>, that no such notice will be necessary when the breaking of a seal is necessitated by the occurrence of an Emergency. Such notice will specify the time at which a meter seal will be broken by KESC personnel, and the Company will be given the opportunity to be present when such seals are broken.

#### 8.6 Repair, Replacement or Recalibration of Metering System.

When any component of the Metering System is found to be outside acceptable limits of accuracy or otherwise not functioning properly, KESC shall forthwith repair, recalibrate or replace such component of the Metering System at its expense. Similarly, when any component of the Back-Up Metering System is found to be outside acceptable limits of accuracy or otherwise not functioning properly, the Company shall forthwith repair, recalibrate or replace such component of the Back-Up Metering System at its expense. Upon the completion of any examination, maintenance, repair or recalibration of, or replacement of any component in, the Metering System or the Back-Up Metering System shall be jointly sealed.

8.7 <u>Telecommunications Circuit</u>.

The Company, at its sole cost and expense, shall provide:

(a) Telecommunications system for (i) telecommunication and teleprotection facilities at KESC's grid stations at Baldia and Hub Chowki and the Complex control centre; and (ii) telemetering and data interface for KESC's Supervisory Control and Data Acquisition Systems (SCADA);

(b) An extension of the Control Centre's PBX system in the control room of the Complex to permit voice communications between the Complex and the Control Centre; and

(c) Equipment in the Complex to transmit and receive facsimiles.

The selection and installation of items to be provided by the Company in accordance with this Section 8.7(a) shall be subject to the prior written approval of KESC.

ulu ar
## ARTICLE IX COMPENSATION, PAYMENT AND BILLING

## 9.1 Capacity Payments.

F

(a) From and after the Commercial Operations Date, KESC shall pay the Capacity Payment to the Company, in accordance with Section 9.7. For each Month, the Capacity Payment shall be equal to the product of the Capacity Purchase Price and the Dependable Capacity in effect for such Month; <u>provided</u>, that if the Commercial Operations Date occurs on a date which is not the first Day of a Month, the Capacity Payment, for the period from the Commercial Operations Date occurs, shall be multiplied by a fraction the numerator of which is the number of Days remaining in the Month from and including the Day immediately following the Commercial Operations Date and the denominator of which is the number of Days in such Month.

(b) If, for a period of eighteen (18) consecutive Days, the Company is unable to deliver Net Electrical Output in an amount exceeding fifty percent (50%) of the Despatch levels requested by KESC for a reason other than Force Majeure, which Despatch requests are within the Technical Limits of the Complex, other than because of a Maintenance Outage or a Scheduled Outage (or a breach or default by the GOP under the Implementation Agreement or the Guarantee), then KESC shall be entitled to suspend the payment of Capacity Payments until such time as the Company notifies KESC that it is again capable of meeting KESC's Despatch requests and delivers Net Electrical Output for not less than three (3) hours in an amount not less than fifty percent (50%) of the next Despatch request by KESC, which Despatch request shall occur on the Day immediately following the notice from the Company. Any Capacity Payments or portions thereof not paid by KESC pursuant to this Section 9.1(b) shall be credited towards the payment of any liquidated damages payable by the Company pursuant to Section 9.4(b).

(c) If, during the Term, the Dependable Capacity of the Complex as determined by two consecutive Annual Capacity Tests is reduced by an amount that exceeds six percent (6%) of the AIDC and if there are any further reductions in Dependable Capacity in excess of two percent (2%) of the AIDC thereafter, then in each case the escalable component of the Capacity Purchase Price for the present Agreement Year and each subsequent Agreement Year shall be adjusted so that, after such adjustment, the Capacity Purchase Price for the average Dependable Capacity provided for the Term, taking into account the reduction thereof and assuming that such reduction shall continue for the remainder of the Term, shall equal the average Capacity Purchase Price shown in Table I of Schedule 6. If the Parties cannot agree on the appropriate adjustment to the Capacity Purchase Price to be made in connection with the reduction in Dependable Capacity, the matter shall be referred to an expert for determination in accordance with Section 15.2. If, following any downward adjustment in the Capacity Purchase Price pursuant to this Section 9.1(c), the Dependable Capacity is increased by one percent (1%) or more of the AIDC, the Capacity Purchase Price shall be adjusted upward in the like manner used for the downward adjustment.

(d) As soon as practicable after the date of this Agreement but in any event not later than the date of the earlier to occur of either Financial Closing or Commercial Operations Date, the Company shall notify KESC and the GOP in writing that it elects Exchange Risk Insurance (as

defined in the Implementation Agreement), subject to proper application therefor with the National Bank of Pakistan, or indexation of the foreign loan portion of the Non-Escalable Component of the Capacity Purchase Price for changes in the value of the Pakistan Rupee against the Dollar under the terms and conditions of Schedule 6. If the Company elects Exchange Risk Insurance it will notify KESC not later than the date of the earlier to occur of either Financial Closing or Commercial Operation Date whether it elects to pay the premium for the Exchange Risk Insurance or, after the Commercial Operations Date, to have the premium paid directly by KESC with a reduction in the Capacity Purchase Price equal to the Exchange Risk Insurance Component thereof.

#### 9.2 <u>Energy Payments</u>.

(a) From and after the Commercial Operations Date, KESC shall pay to the Company, Monthly in arrears, the Energy Purchase Price for each kWh of Net Electrical Output, in accordance with the provisions of Section 9.7.

(b) During Commissioning, KESC shall pay to the Company, Monthly in accordance with Section 9.7, an amount equal to the Fuel Cost Component of the Energy Purchase Price as set forth in Schedule 6, for each kWh of Net Electrical Output delivered to KESC during testing performed pursuant to the Construction Contract and Commissioning.

(c) The Company may at its sole option nominate a revised Energy Purchase Price. This revised Energy Purchase Price will be used for the purpose of payment of Net Electrical Output and Despatch while in effect. The revised Energy Purchase Price will be effective until the Company notifies KESC that the revised Energy Purchase Price is no longer to be used for payment purposes; <u>provided</u>, <u>however</u>, that the revised Energy Purchase Price shall be effective for not less than one (1) week. The revised Energy Purchase price shall not in any event exceed the then-prevailing Energy Purchase Price calculated in accordance with this Article IX and Schedule 6.

3

9.3 Energy Purchase Price Premium.

In the event that the Commercial Operations Date occurs on or before December 31, 1997 (the "Premium Date"), in addition to the Energy Purchase Price, KESC shall pay to the Company a premium of Rupees equivalent to US Cent 0.25 per kWh of Net Electrical Output delivered to KESC from the Commercial Operations Date and continuing to the tenth (10th) anniversary of such date; provided, however, that the Dependable Capacity is maintained above 100 MW during such ten (10) year period. The failure to achieve at least 100 MW in a Dependable Capacity test carried out in accordance with Section 10.4 shall result in loss of the premium until the Dependable Capacity of the Complex is determined to be equal to or above 100 MW as demonstrated in a subsequent Dependable Capacity test. In no event shall the Premium Date or the period, if any, during which the premium shall be paid be extended by reason of a Force Majeure Event or for any other reason whatsoever, provided, however, that the Premium Date shall be extended on a Day-for-Day basis for (a) KESC's failure to complete the Interconnection and Transmission Facilities as required by Section 7.4 or (b) a delay caused by KESC of testing of the Complex by the Company pursuant to Section 10.1(b). The judgement as to whether, but for KESC's failure to complete the Interconnection and Transmission Facilities or to accommodate the Company's testing schedule, the Complex was or would have been ready for the scheduled programme of tests, as described more fully in Sections 3.6 and 10.1, respectively, shall be made by a committee (the "Interconnection" Committee) to be composed of two

(2) experts each from KESC and the Company and an independent expert acceptable to the experts of each Party. If either Party believes that it is reasonably likely that the Interconnection and Transmission Facilities will not be completed as required by Section 3.6 and 7.4, such Party may require the formation of the Interconnection Committee. If either Party requests the formation of the Interconnection Committee. If either Party requests the formation of the Interconnection Committee, the experts of each Party to serve on the Committee shall be named by the earlier of (i) one hundred fifty (150) Days prior to the Scheduled Commercial Operations Date or (ii) ninety (90) Days prior to the date KESC expects to complete the Interconnection and Transmission Facilities, as such date is conveyed by KESC to the Company. The independent expert shall be selected within thirty (30) Days thereafter. The Committee shall monitor, and keep itself fully informed of the progress on the construction of the Complex through periodic inspections.

#### 9.4 Liquidated Damages and Penalties.

(a) <u>Delay in Commissioning</u>. In the event that the Complex shall not have been Commissioned on or before the Required Commercial Operations Date, then the Company shall pay to KESC, Monthly in arrears, as liquidated damages for delays in the occurrence of the Commercial Operations Date \$2.50 per kW of the Estimated Dependable Capacity per Month prorated daily until the Commercial Operations Date.

#### (b) Forced Outages and Partial Derating.

(i) In the event that the sum of (A) the Equivalent Weighted Forced Outage Energy (as hereinafter defined) for the Complex plus (B) the Weighted Complex Partial Derating (as hereinafter defined) expressed in MWh plus (C) the Weighted Complex Maintenance Outages (as hereinafter defined) expressed in MWh, shall exceed  $\Omega$  MWh in any Agreement Year (where  $\Omega$  equals the Average Dependable Capacity multiplied by 500 hours), then the Company shall pay to KESC, as liquidated damages, the Capacity Damages Amount multiplied by ((A+B+C) -  $\Omega$ ) for any Agreement Year. In computing liquidated damages payable by the Company for the first Agreement Year, the sum of A+B+C in this equation shall be multiplied by 0.75. There shall be no adjustment in any subsequent Agreement Year.

(ii) For the purposes of this Section 9.4(b), the Equivalent Weighted Forced Outage Energy shall be the summation of the products of (A) each hour of Forced Outage, multiplied by (B) the Dependable Capacity of the Complex as last tested, expressed in MW, multiplied by (C) the applicable Weighting Factor.

(iii) For the purposes of this Section 9.4(b), the Weighted Complex Partial Derating shall be the summation of the products of (A) the Partial Derating during the period a Partial Forced Outage was in effect, multiplied by (B) the time in hours that such Partial Derating shall be in effect, multiplied by (C) the applicable Weighting Factor.

(iv) For the purposes of this Section 9.4(b), the Weighted Complex Maintenance Outages shall be the summation of the products of (A) the reduction in Dependable Capacity during the period a Maintenance Outage was in effect, multiplied by (B) the time in hours that such Maintenance Outage shall be in effect, multiplied by (C) the applicable Weighting Factor and multiplied by (D) 0.5.

(v) The "Weighting Factor" applicable to hours of Forced Outage, Partial Derating and Maintenance Outage shall be:

All Working Days Except Fridays			Fri	Fridays	
	12 Peak	12 Non-	12 Peak	12 Non-	
<u>Months</u>	<u>Hours</u>	Peak Hours	Hours	<u>Peak Hours</u>	24 Hours
January	0.5	0	-	-	0
February	0.5	0 .	-	_	0
March	1	0,5	-	ن <b>ـ</b>	0
April	2	1.25	-	_	1
May	2.6	1.5	2	1.5	<u>.</u>
June	2.6	1.5	2	1,5	-
July	2	1.5	-	-	1
August	1.5	1	-	-	1
September	t t	0.5	-	-	0
October	2	1.09	-	-	1
November	0,5	0	-	791	0
December	0,5	0	<b>.</b>	•	0

where for these purposes peak hours shall be the hours of 11:00 am to 11:00 pm on each Day, provided, however, that KESC reserves the right to designate new periods, peak hours (which shall be periods of twelve (12) consecutive hours, but which may vary from month to month) and Weighting Factors for the purposes of this Section 9.4(b) by giving notice in writing to the Company at least twelve (12) Months prior to the first Day of any Agreement Year, provided, further, that the maximum value of any Weighting Factor shall be 2.6, the total weighted hours per Agreement Year shall be 8,760, the hours in any period shall be consecutive, there shall be no more than three (3) periods in any one Day, and on Fridays, except during any two of KESC's peak Months, there shall be one period only; provided, further, that KESC shall only make such redesignations in response to long term load and generation changes.

(c) <u>Despatch Levels</u>. In the event that after two (2) identical Despatch requests separated by the greater of (A) a sufficient period of time for the Company to have complied with the first request based on ramp time schedules as provided in the Technical Limits, or (B) ten (10) minutes, the Company does not achieve the operating level requested by KESC pursuant to Section 6.2(b) within the time allowed by the Technical Limits, within a tolerance of plus or minus three percent (3%), then the Company shall pay to KESC, as liquidated damages, an amount equal to Rs.0.30 per kWh for each kWh outside the tolerance; <u>provided</u>, <u>however</u>, that KESC shall not be entitled to liquidated damages pursuant to this Section 9.4(c) if the requested operating level cannot be achieved within the Technical Limits or is above the Declared Available Capacity of the Complex (as adjusted by Forced Outages or Partial Deratings, whether such Forced Outages or Partial Deratings are declared prior or subsequent to such Despatch requests). If the Company is unable to meet the requested Despatch for a continuous period of three (3) hours, then such shortfall shall be treated as a Partial Derating.

(d) <u>Waiver of Defenses</u>. The Parties agree that KESC may be substantially damaged in amounts that may be difficult or impossible to determine in the event that the Complex (i)

54

31-

is not in service by the Required Commercial Operations Date, (ii) is not capable of achieving and maintaining the expected Dependable Capacity, (iii) cannot minimize the number of Forced Outage hours and Partial Deratings, or (iv) cannot achieve the designated operating levels. Therefore, to the limited extent set out in this Agreement, the Parties have agreed on sums that the Parties agree are reasonable as liquidated damages and it is further understood and agreed that the payment of liquidated damages is in lieu of actual damages for such occurrences and the collection of such damages, plus any interest that may be due thereon, is the sole remedy of KESC in the event of any such failure by the Company. The Company hereby waives, to the extent permitted by applicable law, any defense as to the validity of any liquidated damages in this Agreement on the grounds that such liquidated damages are void as penalties.

(e) <u>Adjustment</u>. The amounts of all of the liquidated damages set forth in Sections 9.4(b) and (c) shall be adjusted from time to time in accordance with Schedule 6.

(f) <u>Security</u>.

F

(i) <u>Company Security</u>. (A) On the date of the Construction Financial Closing, the Company shall provide to KESC the Company Letter of Credit in an amount equal to \$25.00/kW of Estimated Dependable Capacity. Upon delivery of the Letter of Credit required by Section 9.4(f)(i)(B), KESC shall release to the Company the Letter of Credit provided pursuant to this Section 9.4(f)(i)(A).

(B) On the date of the Financial Closing, the Company shall provide to KESC the Company Letter of Credit in an amount equal to \$25.00/kW of Estimated Dependable Capacity. Upon delivery of the Letter of Credit required by Section 9.4(f)(i)(C), KESC shall release to the Company the Letter of Credit provided pursuant to this Section 9.4(f)(i)(B).

(C) On the Commercial Operations Date, the Company shall provide to KESC the Company Letter of Credit, which shall have a term of twelve (12) months from the Commercial Operations Date, in an amount equal to an aggregate of two (2) Months of Capacity Payments (as adjusted in accordance with Schedule 6) computed on the basis of the AIDC or, with respect to any replacement Company Letter of Credit, on the basis of the thenprevailing Dependable Capacity (in each case without regard in any event to amounts being held in escrow pending the resolution of a Dispute concerning liquidated damages assessed against the Company). Not less than ten (10) Days prior to the expiration of the Company Letter of Credit and each replacement Company Letter of Credit, the Company shall provide a replacement Company Letter of Credit which shall have a term of twelve (12) months from the expiration of the immediately precedent Company Letter of Credit. The Letter of Credit provided pursuant to this Section 9.4(f)(i)(C) shall be maintained throughout the Term and shall be reinstated to the full required amount within thirty (30) days of any draw thereon by KESC.

(D) In the event that the Company shall be required to pay liquidated damages to KESC, and the Company fails to make any such payments of damages when due, then KESC shall be entitled to draw or collect such amounts, less any amounts disputed by the Company, from the Company Letters of Credit upon presentation of a certificate of an authorized officer of KESC stating that (1) amounts shown in the invoice accompanying the

55

14----

certificate are due and payable by the Company to KESC under this Agreement and (2) an invoice for such amount has been delivered to the Company at least twenty-five (25) Days prior to the presentation of the certificate and either (aa) no amounts shown in such invoice have been disputed by the Company or (bb) a portion of the amount shown in the invoice has been disputed by the Company, identifying such disputed amount. The certificate shall be accompanied by the relevant invoice delivered to the Company and any Invoice Dispute Notice delivered to KESC by the Company.

(ii) KESC Security. (A) On the Commercial Operations Date, KESC shall provide to the Company the KESC Letter of Credit, which shall have a term of twelve (12) months from the Commercial Operations Date. Not less than ten (10) Days prior to the expiration of the KESC Letter of Credit and any subsequent KESC Letter of Credit, KESC shall provide a replacement KESC Letter of Credit which shall have a term of twelve (12) months from the expiration of the immediately precedent KESC Letter of Credit. The KESC Letter of Credit shall provide for draws by the Company for the purposes of Section 9.7 in immediately available funds for any amounts due to the Company under Section 9.7, less any amounts disputed by KESC, upon presentation of a certificate of an authorized officer of the Company stating that (1) amounts shown in the invoice accompanying the certificate are due and payable by KESC to the Company under this Agreement and (2) an invoice for such amount has been delivered to KESC at least twenty-five (25) Days prior to the presentation of the certificate and either (aa) no amount shown in such invoice have been disputed by KESC or (bb) a portion of the amount shown in the invoice has been disputed by KESC, identifying such disputed amount. The certificate shall be accompanied by the relevant invoice delivered to KESC and any Invoice Dispute Notice delivered to the Company by KESC. The KESC Letter of Credit provided pursuant to this Section 9.4(f)(ii)(A) shall be maintained in the required amount throughout the Term. The KESC Letter of Credit shall be reinstated to the full required amount within thirty (30) Days of any draw thereon by the Company; provided, however, that if, after the draw by the Company, the amount remaining under the KESC Letter of Credit is less than the amount due to the Company under another invoice already delivered to KESC by the Company (the "Second Invoice"), then the KESC Letter of Credit shall be reinstated to the full required amount by the twenty-fifth (25th) Day following the delivery to KESC of the Second Invoice

(B) Notwithstanding the provisions of Section 9.4(f)(ii)(A) above, in lieu of delivering the KESC Letter of Credit or any replacement KESC Letter of Credit, KESC may in its sole discretion establish with the Escrow Bank an Escrow Account equal to the thenrequired amount of the KESC Letter of Credit from which only amounts due and payable by KESC to the Company under this Agreement may be withdrawn. The Escrow Agreement governing this account shall provide for withdrawals from the account upon presentation of a draft executed by the Company for any amount due and payable by KESC under this Agreement and shall be accompanied by (1) a certificate executed by an authorized officer of the Company stating that (aa) amounts due and payable by KESC to the Company under this Agreement have been invoiced to KESC in accordance with this Agreement at least twenty-five (25) Days prior to the presentation of the certificate and (bb) either (xx) no amount shown in such invoice has been disputed by KESC or (yy) a portion of the amount shown in the invoice has been disputed by KESC, identifying such disputed amount, (2) the relevant invoice delivered to KESC and (3) the Invoice Dispute Notice delivered to the Company by KESC, if

56

Ale

Mp

any. The Company shall not be entitled to draw any amounts shown in the invoice to KESC that have been disputed by KESC. If and for so long as KESC maintains the Escrow Account in lieu of the KESC Letter of Credit, the Escrow Account shall be reinstated to the full required amount (in addition to any amounts being held in a separate account in escrow pursuant to section 9.8 pending the resolution of a Dispute) within thirty (30) days of any draw thereon by the Company; provided, however, that if, after the draw by the Company under amount remaining in the Escrow Account is less than the amount due to the Company under another invoice already delivered to KESC by the Company (the "Second Invoice"), then the Escrow Account shall be reinstated to the full required amount by the twenty-fifth (25th) Day following the delivery to KESC of the Second Invoice. For so long as KESC maintains the Escrow Account in lieu of the KESC Letter of Credit, the Company shall draw or collect all amounts due and payable by KESC under this Agreement directly from the Escrow Account established by KESC.

(C) To the extent that KESC is required by the provisions of this Article IX to pay Capacity Payments, Energy Payments, Supplemental Tariff Payments or Supplemental Charges, the Energy Purchase Price premium referred to in Section 9.3, or the amounts referred to in Section 9.6, the Company shall draw or collect such amounts directly from the KESC Letter of Credit or, if KESC has established the Escrow Account pursuant to Section 9.4(f)(ii)(B), the Escrow Account.

(iii) <u>Improper Draws</u>. In the event that either Party draws against the letter of credit or Escrow Account provided by the other Party and it is subsequently determined that the Party making such draw was not entitled to do so, then the drawing Party shall repay such amount to the other Party, together with all costs and expenses incurred by such other Party in connection with such drawing, plus interest thereon from the date of the draw through the date of repayment at the Base Rate plus four (4) percent per annum, compounded semi-annually.

# 9.5 Payment of Liquidated Damages.

(a) Within fourteen (14) Days after the end of each Month and within thirty (30) Days of the end of each Agreement Year, subject to the Company's review, KESC shall compute and advise the Company in the Invoice for Liquidated Damages of the amount of liquidated damages due to KESC pursuant to this Agreement for the preceding Month or Agreement Year, as the case may be. Disputes regarding payments of any amount specified in an Invoice for Liquidated Damages shall be resolved pursuant to the terms of Section 9.8.

(b) The Company shall pay the invoice amount to KESC by no later than twentyfive (25) Days after delivery by KESC of the Invoice for Liquidated Damages. Late payments shall bear interest at a rate per annum equal to the Base Rate plus two percent per annum (or at the Base Rate in the event that the Company has notified KESC within such twenty-five (25) Day period that it disputes such payment and pays within such twenty-five (25) Day period all undisputed amounts) compounded semi-annually and shall be computed for the actual number of Days on the basis of a three hundred sixty-five (365) Day year.

NG AL

#### 9.6 <u>Bonus</u>.

In the event that the Net Electrical Output during any Agreement Year is in excess of 680,433,000 kWh, then KESC shall pay to the Company within sixty (60) days of the end of such Agreement Year, in addition to any other payments due to the Company pursuant to Section 9.2, an amount equal to Rs.0.08 for each kWh by which the Net Electrical Output of the Complex during the Agreement Year is greater than 680,433,000 kWh.

(b) The amount of the bonus payable pursuant to this Section 9.6 shall be adjusted from time to time in accordance with Schedule 6.

9.7 Billing.

(a) From and after the Commercial Operations Date, the Company shall invoice KESC for the Capacity Payment due for each Month at any time following the tenth (10th) Day of such Month, unless the eleventh (11th) Day of such Month is not a Business Day for KESC, then the Company may invoice KESC on the last preceding Day that is a Business Day for KESC. Each invoice shall set forth the then-prevailing Capacity Furchase Price, as determined in accordance with Schedule 6, and the then-prevailing Dependable Capacity. The Company shall draw on the KESC Letter of Credit, pursuant to Section 9.4(f)(ii)(A), or the Escrow Account, pursuant to Section 9.4(f)(ii)(B), for the amount of each invoice, as the case may be, less any disputed amounts, on or after the twenty-fifth (25th) Day following the later of the tenth (10th) Day of such Month or the Day the invoice is delivered to KESC.

(b) For each Month in which the Company delivers Net Electric Output to KESC, the Company shall read the Metering System in accordance with Section 8.4 and shall-prepare an invoice showing the amount of the Energy Payment payable to Company for such Month (provided that in the case of the Month in which the Commercial Operations Date occurs the Company shall prepare two invoices, the first for the period commencing on the first Day of the Month through and including the Commercial Operations Date and the second invoice for the period commencing on the Day after the Commercial Operations Date through the end of the Month). Such invoices shall show the reading of the Metering System taken on or near the end of the Month in accordance with Section 8.4, the reading of the Metering System taken on or near the end of the preceding month and such other information and calculations, in reasonable detail, to permit KESC to confirm the consistency of the invoice with the provisions of Schedule 6. The Company shall draw on the KESC Letter of Credit or the Escrow Account for the amount shown in the invoice as due to Company (less any amount disputed by KESC) at any time on or after the twenty-fifth (25th) Day following the Day the invoice is delivered to KESC.

(c) Unless specifically provided otherwise in Section 10 of Schedule 6, any amounts or portions of amounts that are Supplemental Charges may be invoiced by the Company on a Monthly basis at any time after the first Day of the Month following the Month in which any such Supplemental Charges are incurred by the Company. The invoice shall state that the due date for payment of such invoice by KESC shall be the date twenty-five (25) Days following the date of delivery of such invoice. With respect to invoices for Pass-Through Items, such invoices from the Company to KESC shall be accompanied by the invoice to the Company for which recovery from KESC is being sought. The Company shall draw on the KESC Letter of Credit or the Escrow

58

Als-

Account for the amount shown in the invoice as due to Company (less any amount disputed by KESC) at any time after the twenty-fifth (25th) Day following the Day the invoice is delivered to KESC.

(d) Either Party shall have the right to review an invoice or statement prepared by the other Party, and if it disagrees with the determination of the amount payable by or to such Party under such invoice or statement, may request clarification and substantiation of such invoice or statement. Neither Party shall waive the right to seek revision of an Invoice and payment of the corrected amount unless such Party fails to deliver an Invoice Dispute Notice within the period provided in Section 9.8.

(e) Late payments shall bear interest at a rate per annum equal to the Base Rate plus two percent (2%) per annum compounded semi-annually and shall be computed for the actual number of Days on the basis of a three hundred sixty-five (365) Day Year.

#### 9.8 Payment Disputes.

(a) At any time prior to one hundred and eighty (180) Days after the date that an invoice has been received by a Party, such receiving Party may serve notice (an "Invoice Dispute Notice") on the other Party that the amount of such invoice is in dispute. Each Invoice Dispute Notice shall specify the invoice concerned, the amount of the Dispute and the basis therefor.

A copy of any Invoice Dispute Notice delivered by KESC to the Company shall also be delivered to the KESC Letter of Credit issuing bank or the Escrow Bank, as appropriate. Until a copy of the Invoice Dispute Notice delivered by KESC to the Company is received by the KESC Letter of Credit issuing bank or the Escrow Bank, as appropriate, such bank shall be entitled to assume that no amount shown in the relevant invoice delivered by the Company for payment is disputed by KESC.

(b) In the event of such a Dispute either Party may seek resolution of the Dispute in accordance with Article XV hereof. Upon the determination by the expert appointed under Article XV that all or any portion of a disputed amount is owed by a Party, such Party shall pay such amount to the other Party in accordance with subsection (c) below, or notify the other Party of its intent to arbitrate the Dispute and deposit the amount determined by the expert to be owed into an interest earning escrow account established for that purpose with a scheduled Pakistan bank, which in the case of KESC shall be a separate subaccount with the Escrow Bank (or, if KESC has not established an Escrow Account, in the bank in which KESC has established the KESC Letter of Credit) and in the case of the Company shall be an account with the bank issuing the Company Letter of Credit then in effect. Until a final resolution of the Dispute, the funds held in escrow shall not be used for any other purpose or as a credit against any other obligation hereunder, except as a reserve against the contingent obligation of the matter under Dispute.

(c) Upon the resolution of the Dispute, any amounts determined to be owing which have not been paid shall be paid or released from escrow by the owing Party to the other Party, as the case may be, together with interest equal to the Base Rate from the date payment under the applicable invoice was due through the date of deposit into the escrow account and any interest earned thereon while held in escrow.

#### 9.9 Proration Resulting From KESC's Operation or Termination.

(a) If the Complex is operated by KESC pursuant to Section 6.11 or this Agreement is terminated in accordance with its terms, then for the purposes of calculating liquidated damages due to KESC under Section 9.4(b), such liquidated damages shall be determined by multiplying the applicable hourly threshold figure by the "Damages Proration Factor", as hereinafter defined, and substituting the product thereof for the applicable threshold figure.

(b) The "Damages Proration Factor" shall be one (1) minus a fraction, the numerator of which shall be the number of Days the Complex was operated by KESC or the number of Days remaining in the Agreement Year following termination, as the case may be, and the denominator of which is three hundred and sixty-five (365).

#### 9.10 Supplemental Tariff Payments

(a) If there shall have occurred a Pakistan Political Force Majeure Event or a Change in Law, then upon the termination of the Force Majeure Event or compliance with the Change in Law, as the case may be, and the completion of restoration or modification of the Complex, if any is necessary, and the resumption of normal operations at the Complex, the Company will be entitled to invoice KESC and KESC will pay to the Company, in addition to the Capacity Payments and the Energy Payments pursuant to Sections 9.1 and 9.2 and any other payments due hereunder, the Supplemental Tariff Payments. Invoices for Supplemental Tariff Payments shall be delivered by the Company and paid by KESC in the same manner and on the same schedule as invoices for Capacity Payments as provided in Section 9.7(a) (with respect to additional capital costs) and, if increased consumables are to be payable on a kWh basis, as agreed by the Company and the GOP, for Energy Payments as provided in Section 9.7(b) (with respect to the costs of additional consumables).

9.11 <u>Retained Payments Fund</u>

Al

(a) On or before the Commercial Operations Date, the Company shall establish and maintain, for the remaining Term, a separate Reserve Fund with a depository institution and under depository agreements reasonably satisfactory to KESC. On the termination of this Agreement, all amounts in the Reserve Fund shall be payable to Company.

(b) The Reserve Fund shall be funded by the Company out of retained earnings commencing on the date that the first Capacity Payment is made. On each Capacity Payment date, one twenty-fourth of the annual operating and maintenance budget for the Complex, less fuel expenses, will be deposited into the Reserve Fund until a reserve of six such payments is established. After the second Agreement Year and at any time thereafter, the Reserve Fund may be reestablished at such other level that the Parties agree is appropriate for a facility of this size and type, considering Prudent Utility Practices, the design, technology and operating history of the Complex and other pertinent information. Any investment income resulting from the depository arrangements of the Reserve Fund shall remain in the Reserve Fund; <u>provided</u>, <u>however</u>, that so long as no Company Event of Default exists, any monies in excess of the minimum investment required above may be paid to the Company upon its request.

60

(c) Monies in the Reserve Fund may be drawn on and used by the Company, (i) to pay expenses on Major Maintenance Items (as defined below) or (ii) only to the extent the Company lacks other available funds therefor, for the purpose of paying maintenance and associated operating expenses with respect to the Complex or to pay for alterations, repairs, improvements, renewals and replacements with respect to the Complex which are necessary to the proper operation of the Complex. As used herein "Major Maintenance Item" means an item of maintenance or repair of the Complex which will require a material expenditure and which is anticipated to be performed in accordance with manufacturers recommendations, sound engineering practices or Prudent Utility Practices but which is not expected to be performed on an annual or more frequent basis.

(d) (i) If after the withdrawal of any funds from the Reserve Fund for the payment of Major Maintenance Items as described in Section 9.11(c)(i), the amount in the Reserve Fund is less than the amount required pursuant to Section 9.11(b), the Company shall replenish the Reserve Fund by depositing funds therein in accordance with Section 9.11(b).

(ii) If, after the withdrawal of any funds from the Reserve Fund for the purpose described in Section 9.11(c)(ii) above, the amount remaining in the Reserve Fund is less than the amount required pursuant to Section 9.11(b) above, the Company shall replenish the Reserve Fund by depositing therein, within one Month after the end of such Month in which such withdrawal occurred, an amount sufficient to restore the amount required in Section 9.11(b). Such amount shall be paid out of fifty percent (50%) of the escalable component of the Capacity Payment, available during the Month; provided, however, if the Company's net cash flow is insufficient to fund the Reserve Fund at the required level, any shortfall shall be carried over and due the following Month(s).

(e) The Company shall keep accurate records with respect to the Reserve Fund and all disbursements therefrom and shall, upon KESC's request, supply a complete accounting or independent audit thereof to KESC.

(f) Separate accounts established at the request of the lenders pursuant to the Financing Documents that in all material respects comply with the provisions of this Section 9.11 shall satisfy the Company's obligation to maintain a Reserve Fund hereunder.

F

#### ARTICLE X TESTING AND CAPACITY RATINGS

# 10.1 Testing of the Complex Prior to Commercial Operations Date.

The Company will use its best efforts to provide KESC on an on-going basis with (a) relevant information regarding its programme for testing the Complex. The Company's preliminary programme and schedule for testing the Complex and KESC's obligations (including the availability of the Interconnection Facilities and Transmission Facilities) are set forth in Schedule 10 hereto. Not less than thirty (30) Days prior to the commencement of such test programme, the Company will deliver to KESC in writing the final programme for testing the Complex, including the expected duration of the Company's start-up testing programme and a tentative schedule for conducting all tests required by Sections 10.2 and 10.3. The Company shall advise KESC in writing of any changes in its final schedule for the testing programme not less than seven (7) Days prior to the commencement of the tests required by Section 10.2. Such final schedule may not materially increase or advance the date of KESC's obligations as set forth in Section 7.4 without the prior written consent of KESC. If the schedule for any test required by Section 10.2 or 10.3 is adjusted after the Company has provided KESC with the final testing programme schedule, then the Company shall advise KESC not less than forty-eight (48) hours prior to the commencement of any such test. On each Day beginning with the Day on which testing commences, the Company shall provide KESC with a schedule of the tests to be conducted on the following Day or Days (if such test will continue for more than one (1) Day). All testing of the Complex shall satisfy the Commissioning and test criteria provided in Sections 10.2 and 10.3 and Schedule 4.

If KESC is unable to accommodate the schedule for such test or tests as provided by (b) the Company in the final schedule for the programme of tests pursuant to Section 10.1(a), KESC will give the Company notice within forty-eight (48) hours of its receipt of the final schedule for testing of its requirements regarding deferral of any Commissioning test or tests for the Complex and the Parties will mutually agree on a date for any deferral test or programme of tests; provided, however, that should KESC defer any Commissioning tests beyond ten (10) Days from the date on which the tests were originally scheduled and such deferral causes the Scheduled Commercial Operations Date of the Complex, as agreed to by the Parties under Section 7.4, to be delayed or deferred, then KESC shall pay to the Company Monthly, in arrears, (and prorated for any portion of a Month) an amount equal to the carrying cost on the debt related to the Complex under the Financing Documents plus fifty percent (50%) of the escalable component of the Capacity Purchase Price. Such payments shall continue until the date that the delay or deferral caused by KESC ceases; provided, however, that KESC shall notify the Company at the end of any such delay or deferral. In addition, the Required Commercial Operations Date and the Premium Date shall be extended on a Day-for-Day basis by the number of Days the testing is delayed due to KESC; provided, however, that the extension of the Required Commercial Operations Date and the Premium Date shall be subject to a certification by the Interconnection Committee (as defined in Section 9.3) that the Complex was ready for Commissioning tests and that the delay or deferral caused by the then-scheduled Commissioning of the Complex to be delayed or deferred.

مرلا

Me

# 10.2 Tests Prior to Synchronization of the Complex.

The Company shall carry out, or shall cause the Contractors to carry out, the following tests:

- (a) automatic voltage regulator setting and adjusting in stand-still condition and with the generator running at no load;
- (b) engine governor control checks, including a governor overspeed test;
- (c) open and short circuit tests on the generator; and
- (d) functional testing and timing of high voltage switchgear in the sub-station of the Complex.
- (e) The Company and KESC shall verify that the protection level settings for the following are as agreed by the Operating Committee:
  - (i) stator earth fault;
  - (ii) negative phase sequence;
  - (iii) generator transformer overcurrent and earth fault; and
  - (iv) high voltage busbar protection.
- (f) Voltage phasing checks will be carried out between the sub-station of the Complex and the Grid System.

All intertripping circuits between the Complex and KESC equipment will be proved.

10.3 Tests After Synchronization of the Complex and Commercial Operations Test.

(a) After first synchronizing the Complex, initial operational testing of the Complex shall be conducted by the Company or its Contractors. Once the Company is satisfied that the Complex is capable of continued reliable operation, the Company shall so notify KESC and carry out or cause its Contractors to carry out, the Commercial Operations Test, which will include:

- (i) initial Dependable Capacity test;
- (ii) reliability run test;
- (iii) automatic voltage regulator droop;

(iv) engine governor operation;

(v) reactive capability;

UB Als

63

- (vi) minimum load capability;
- (vii) response of Complex to step load changes; and
- (viii) full load rejection;

Minimum performance criteria for the above test or tests for the Complex to become Commissioned are included in Schedule 4.

(b) <u>Reliability Run and Initial Dependable Capacity</u>. Upon completion of the reliability run test prerequisites as included in Schedule 4, the Company shall declare to KESC the commencement of the reliability run test. During the period of the reliability run test, the initial Dependable Capacity of the Complex will be determined in the following manner:

(i) The Complex shall be in operation with normal auxiliaries and full colony load, if connected directly to the Complex, in service.

(ii) The Company will declare to KESC the commencement of the test and will record the reading of the Metering System.

(iii) The test duration will be 6 hours and at the end of this period the Company will record the new reading of the Metering System. The initial Dependable Capacity as determined by such test shall be the difference between the reading taken at the end of the six (6) hour period and the reading taken at the beginning of such period, divided by six (6); <u>provided</u>, that the initial Dependable Capacity shall not be considered to have been established unless the result of such determination is equal to or greater than the minimum criteria for such test set forth in Schedule 4; <u>provided</u>, <u>further</u>, that such initial Dependable Capacity may not exceed 105% of the Estimated Dependable Capacity.

(c) Additional Commercial Operations Tests.

(i) The Company shall be entitled to attempt as many Commercial Operations Tests of the Complex as are necessary to satisfy the minimum criteria for achieving the Commercial Operations Date. The Company shall give KESC not less than three (3) Days notice of each additional Commercial Operations Test it desires to attempt.

(ii) If the results of a Commercial Operations Test satisfy the minimum performance criteria to achieve the Commercial Operations Date, but the Company is not satisfied with the results of such Commercial Operations Test, the Company may request two additional tests to establish the Commercial Operations Date with at least two (2) Days notice provided to KESC prior to a subsequent test; <u>provided</u>, <u>however</u>, that the Company will not be paid for capacity until it has notified KESC that the Company has designated the test as the Commercial Operations Test.

(iii) When the Company is satisfied with a test to establish the Commercial Operations Date, the Company shall notify KESC that the Company has designated such test as the Commercial Operations Test and shall set the Actual Initial

Dependable Capacity at any level up to the tested level; <u>provided</u>, <u>however</u>, such AIDC may not exceed one hundred and five percent (105%) of the Estimated Dependable Capacity.

(iv) The Commercial Operations Date shall occur and Capacity Payments shall commence as of the Day following the date on which the Complex is Commissioned as certified in writing by the Engineer.

# 10,4 Testing of Dependable Capacity of the Complex after Commercial Operations Date.

(a) During each Demonstration Period a test shall be conducted to determine the Dependable Capacity of the Complex. The Company may at any time during a Demonstration Period declare a period to be a test period; <u>provided</u>, that KESC has scheduled in excess of ninety-five (95) percent of the Dependable Capacity of the Complex for a continuous period of six (6) hours or more in its notification of requirements to be given to the Company pursuant to Section 6.2(a)(iv). Upon the Company declaring a test period the Despatch level for the test period shall be deemed to be, and KESC shall provide load for, the maximum capability of the Complex. During such test period the Complex shall not be controlled by AGC.

(b) The test period shall be for six (6) continuous hours. The test shall be run using the Metering System and plant instrumentation for measurements. The Dependable Capacity shall be the Net Electrical Output during those six (6) hours divided by six (6), but may not exceed 105% of the Estimated Dependable Capacity. For purposes of determining Capacity Payments payable pursuant to Section 9.1, the Dependable Capacity Level shall be adjusted to reflect the tested Dependable Capacity, effective the Day after such testing is complete.

(c) In the event that the Company has not declared a test period within forty-two (42) Days of the start of a Demonstration Period, then if:

(i) KESC has not scheduled the required level and duration of generation pursuant to Section 10.4(a) during the forty-two (42) Day period, such that the Company was able to declare a test period, then the Company may request a test; or

(ii) KESC did schedule the required level and duration of generation pursuant to Section 10.4(a) during the forty-two (42) Day period, such that the Company was able to declare a test but then elected not to declare a test, then KESC may request a test.

(d) If either Party requests a test pursuant to Section 10.4(c), then such test shall be performed in accordance with the provisions of Section 10.4(b) within seven (7) Days following such request, <u>provided</u> that such request is made at least fourteen (14) Days prior to the end of the Demonstration Period. The Company shall give KESC not less than twenty-four (24) hours notice of its intention to perform the test.

(e) The Company may, within twenty-four (24) hours of completion of any test, reject the test and may conduct a retest, <u>provided</u>, <u>however</u>, that the Company cannot conduct more than two retests. The Company shall give KESC at least twenty-four (24) hours notice of the retest and the retest shall be conducted within six (6) Days of the completion of the rejected test.

65

Kle\_

(f) Either Party shall be entitled to request one test of Dependable Capacity of the Complex between any two consecutive Demonstration Periods. The Company shall be entitled to one re-test of any such test provided that it rejects the test within twelve (12) hours of completing the test. The test and, as appropriate, the re-test shall be conducted in accordance with Section 10.4(b), within six (6) Days of its request or, as the case may be, the rejection and the Company shall give KESC not less than twenty-four (24) hours notice of its intention to perform the test.

#### 10.5 Notice of and Compliance With Testing Procedures.

The Company shall carry out Commissioning of the Complex, testing the Dependable Capacity of the Complex at the Commercial Operations Date and testing of Dependable Capacity of the Complex thereafter in accordance with Sections 10.2 through 10.4. KESC shall use its reasonable efforts to comply promptly with all reasonable requests by the Company for assistance in carrying out such testing and Commissioning. KESC shall be given prior written notice of any testing or Commissioning procedure in accordance with Sections 10.2 through 10.4 and shall be entitled to be present and observe any such testing and Commissioning.

# 10.6 Copies of Test Results.

The Company shall provide KESC with copies of the test results of all tests performed pursuant to Sections 10.2 through 10.4 above and after every general overhaul of a generating unit at the Complex. KESC shall not use or disclose such results other than in connection with the administration and enforcement of this Agreement.

10.7 <u>Testing Disputes</u>.

In the event that a Dispute arises between the Company and KESC regarding the testing of Dependable Capacity or the protection tests described in Sections 10.2 through 10.4, such Dispute shall be resolved pursuant to Article XV.

#### 10.8 Estimated Dependable Capacity.

The Company's Estimated Dependable Capacity of the Complex, at the Reference Conditions specified in Schedule 1, is 119.5 MW net.

## ARTICLE XI **INSURANCE**

#### 11.1 Maintenance of Insurance Policies.

(a) The Company, at its sole cost and expense (subject to the provisions of Section 11.1(b), below), shall obtain and maintain, or cause to be obtained and maintained, during the Term the policies of insurance set forth on Schedule 8 in the amounts set forth therein and during the periods mentioned therein; provided, however, that such amounts may be changed from time to time with the prior written consent of KESC; provided, further, that the Company shall not be in breach of its obligations hereunder if and to the extent that any particular insurance is unavailable to it under commercially reasonable terms for reasons other than any negligence or default by, or the condition of, the Complex or the Company.

(b) Following a Pakistan Political Force Majeure Event, (or an event or occurrence that, had it affected the Company directly, would have been a Pakistan Political Force Majeure Event) to the extent that the insurance required by Section 11.1(a), above, is not available to the Company at commercially reasonable rates due to the occurrence of the Pakistan Political Force Majeure Event, (or the other event described above) upon notice to KESC by the Company, the additional cost of such insurance attributable to the occurrence of the Pakistan Political Force Majeure Event (or the other event described above) as determined by an expert in conformity with the provisions of Section 15.2, shall be recoverable by the Company from KESC and treated as a Pass-Through Item. In such an event, in lieu of making such payment to the Company, KESC in its sole discretion may elect to procure the insurance required by Section 11.1(a) on behalf of the Company and deduct five percent (5%) of the escalable component of the then-prevailing Capacity Payments as full compensation therefor. The Company shall be named as the loss payee on any such insurance procured by KESC pursuant to this Section 11.1(b). The additional compensation provided under this Section 11.1(b) and any such deduction shall cease as soon as the Company's insurance rates are no longer affected by the Pakistan Political Force Majeure Event (or the other event described above). From time to time, at the request of KESC or the Company, the expert will determine the extent to which the Company's insurance rates are then affected by the Pakistan Political Force Majeure Event (or the other event described above).

#### Maintenance of "Occurrence" Form Policies. 11.2

The coverage requested in Section 11.1 and any "umbrella" or excess coverage shall be "occurrence" form policies. In the event the Company has "claims made" form coverage, the Company must obtain prior approval of all "claims-made" policies from KESC.

#### 11.3 Policy Endorsements.

 $|\rangle$ 

The Company shall cause the insurers to provide the following endorsement items in the comprehensive or commercial general liability and, if applicable, umbrella or excess liability policies relating to the ownership, construction, operation and maintenance of the Complex provided pursuant to Section 11.1: Als\_\_\_\_

(a) KESC, its directors, officers and employees shall be additional insureds under such policies with respect to claims arising out of or in connection with this Agreement;

(b) The insurance shall be primary with respect to the interest of KESC, its directors, officers, and employees and any other insurance maintained by them is excess and not contributory with such policies;

(c) The following cross liability clause shall be made a part of the policy:

"In the event of claims being made by reason of (i) personal and/or bodily injuries suffered by any employee or employees of one insured hereunder for which another insured hereunder is or may be liable, or (ii) damage to property belonging to any insured hereunder for which another insured is or may be liable, then this policy shall cover such insured against whom a claim is made or may be made in the same manner as if separate policies have been issued to each insured hereunder, except with respect to the limits of insurance.";

(d) The insurer shall waive all rights of subrogation against KESC, its officers, directors and employees; and

(e) Notwithstanding any provision of the policy, the policy may not be canceled, non-renewed or materially changed by the insurer without giving thirty (30) Days, except in the case of non-payment, in which case it will be ten (10) Days, prior written notice to KESC. All other terms and conditions of the policy shall remain unchanged.

 $\frac{1}{2}$ 

4

#### 11.4 Endorsements to Fire and Perils and Machinery Breakdown Policies.

The Company shall cause the insurers to provide the endorsements referred to in Section 11.3(a), (b), (d) and (e) in the fire and perils and machinery breakdown policies covering the Complex as required by Section 11.1.

## 11.5 Certificates of Insurance.

The Company shall cause its insurers or agents to provide KESC with certificates of insurance evidencing the policies and endorsements listed above. Failure by the Company to obtain the insurance coverage or certificates of insurance required by this Article XI shall not in any way relieve or limit the Company's obligations and liabilities under any provision of this Agreement. If the Company shall fail to procure or maintain any insurance required pursuant to this Article XI, then KESC shall have the right to procure such insurance in accordance with the requirements of Schedule 8 and shall be entitled to offset the premiums paid for such insurance against any amounts owed to the Company pursuant to the terms of this Agreement. The Company shall be named as the loss payee on any such insurance procured by KESC pursuant to this Section 11.5.

10

# 11.6 Insurance Reports.

Ē

The Company shall provide KESC with copies of any underwriters' reports or other reports received by the Company from any insurer; <u>provided</u>, that KESC shall not disclose such reports to any other person except as necessary in connection with administration and enforcement of this Agreement or as may be required by any Public Sector Entity having jurisdiction over KESC and shall use and internally distribute such reports only as necessary in connection with the administration and enforcement of this Agreement.

116 Alen

## ARTICLE XII LIABILITY AND INDEMNIFICATION

#### 12.1 Limitation of Liability.

Except as required by Section 12.2, neither Party shall be liable to the other Party in contract, tort, warranty, strict liability or any other legal theory for any indirect, consequential, incidental, punitive or exemplary damages. Neither Party shall have any liability to the other Party except pursuant to, or for breach of, this Agreement; <u>provided</u>, <u>however</u>, that this provision is not intended to constitute a waiver of any rights of one Party against the other with regard to matters unrelated to this Agreement or any activity not contemplated by this Agreement.

## 12.2 Indemnification.

(a) <u>KESC</u>. Except as specifically provided elsewhere in this Agreement, KESC shall indemnify and defend the Company its officers, directors and employees against, and hold the Company its officers, directors and employees harmless from, at all times after the date hereof, any and all Losses incurred, suffered, sustained or required to be paid, directly or indirectly, by, or sought to be imposed upon, the Company its officers, directors and employees, for personal injury or death to persons or damage to property arising out of any negligent or intentional act or omission by KESC in connection with this Agreement. Notwithstanding anything to the contrary contained in the preceding sentence, nothing in this Section 12.2(a) shall apply to any Loss in respect of which the Company receives indemnification in full pursuant to the terms of the Implementation Agreement.

(b) <u>The Company</u>. The Company shall indemnify and defend KESC its officers, directors and employees against, and hold KESC its officers, directors and employees harmless from, at all times after the date hereof, any and all Loss, incurred, suffered, sustained or required to be paid, directly or indirectly, by, or sought to be imposed upon, KESC its officers, directors and employees, for personal injury or death to persons or damage to property arising out of any negligent or intentional act or omission by the Company in connection with this Agreement.

(c) <u>Joint Negligence</u>. In the event injury or damage results from the joint or concurrent negligent or intentional acts or onussions of the Parties, each Party shall be liable under this indemnification in proportion to its relative degree of fault.

(d) <u>Survival</u>. The provisions of this Section 12.2 shall survive for a period of five (5) years following the termination of this Agreement (or such later date as the Company actually vacates the Site where the Complex has been or is to be transferred to the GOP or KESC).

# 12.3 Assertion of Claims to Exceed Minimum Indemnification Amount.

Each Party shall be solely liable, and shall not be entitled to assert any claim for indemnification under this Agreement, for any Loss that would otherwise be the subject of indemnification under this Agreement until all Losses of such Party, in the aggregate, during the then-current Year exceed the Minimum Indemnification Amount. For the purposes of this Section 12.3, a Loss (or claim for indemnification) shall be deemed to arise in the Year the event giving rise to such Loss (or claim for

hlp

116\_

70

indemnification) occurred, or if the event is continuing in more than one Year, in the Year such event ends.

#### 12.4 Indemnification for Fines and Penalties.

Any fines or other penalties incurred by a Party for non-compliance with Laws of Pakistan shall not be reimbursed by the other Party but shall be the sole responsibility of the non-complying Party.

## 12.5 Defense of Claims.

:

(a) The indemnifying Party shall be entitled, at its option and expense and with counsel of its selection, to assume and control the defense of any claim, action, suit or proceeding resulting from any matter for which it is obligated to indemnify the other Party hereunder, subject to the prior approval of the indemnified Party, <u>provided</u> it gives prompt notice of its intention to do so to the indemnified Party and reinburses the indemnified Party for the reasonable costs and expenses incurred by the indemnified Party prior to the assumption by the indemnifying Party of such defense.

(b) Unless and until the indemnifying Party acknowledges in writing its obligation to indemnify the indemnified Party and assumes control of the defense of a claim, suit, action or proceeding in accordance with Section 12.5(a), the indemnified Party shall have the right, but not the obligation, to contest, defend and litigate, with counsel of its own selection, any claim, action, suit or proceeding by any third party alleged or asserted against such Party in respect of, resulting from, related to or arising out of any matter for which it is entitled to be indemnified hereunder, and the reasonable costs and expense thereof shall be subject to the indemnification obligations of the indemnifying Party hereunder.

(c) Upon assumption by the indemnifying Party of the control of the defense of a claim, suit, action or proceeding, the indemnifying Party shall reimburse the indemnified Party for the reasonable costs and expenses of the indemnified Party in the defense of the claim, suit, action or proceeding prior to the indemnifying Party's acknowledgement of the indemnification and assumption of the defense.

(d) Neither Party shall be entitled to settle or compromise any such claim, action, suit or proceeding without the prior written consent of the other Party, <u>provided</u>, <u>however</u>, that after agreeing in writing to indemnify the indemnified Party, the indemnifying Party may settle or compromise any claim without the approval of the indemnified Party.

(e) Following the acknowledgement of the indemnification and the assumption of the defense by the indemnifying Party, the indemnified Party shall have the right to employ its own counsel and such counsel may participate in such action, but the fees and expenses of such counsel shall be at the expense of such indemnified Party, when and as incurred, unless (i) the employment of counsel by such indemnified Party has been authorized in writing by the indemnifying Party, (ii) the indemnified Party shall have reasonably concluded that there may be a conflict of interest between the indemnifying Party and the indemnified Party in the conduct of the defense of such action, (iii) the indemnifying Party shall not in fact have employed independent counsel reasonably satisfactory to the indemnified Party to assume the defense of such action and shall have been so notified by the

Ak\_

indemnified Party, or (iv) the indemnified Party shall have reasonably concluded and specifically notified the indemnifying Party either that there may be specific defenses available to it which are different from or additional to those available to the indemnifying Party or that such claim, action, suit or proceeding involves or could have a material adverse effect upon it beyond the scope of this Agreement. If clause (ii), (iii) or (iv) of the preceding sentence shall be applicable, then counsel for the indemnified Party shall have the right to direct the defense of such claim, action, suit or proceeding on behalf of the indemnified Party and the reasonable fees and disbursements of such counsel shall constitute legal or other expenses hereunder.

M 16

## ARTICLE XIII FORCE MAJEURE

## 13.1 Definition of Force Majeure,

A "Force Majeure Event" shall mean any event or circumstance or combination of events or circumstances that is beyond the reasonable control of a Party and that on or after the date of the earlier to occur of Construction Financial Closing or Financial Closing, materially and adversely affects the performance by such affected Party of its obligations under or pursuant to this Agreement (including a Party's ability to deliver or receive energy from the Complex); provided, however, that any such event or circumstance or combination of events or cicumstances shall not constitute a "Force Majeure Event" within the meaning of this Section 13.1 to the extent that such material and adverse effect could have been prevented, overcome or remedied in whole or in part by the affected Party through the exercise of diligence and reasonable care, it being understood and agreed that reasonable care includes acts and activities to protect the Complex from a casualty event, which are reasonable in light of the likelihood of such event, the probable effect of such event if it should occur and the likely efficacy of the protection measures. "Force Majeure Events" hereunder shall include each of the following events and circumstances, but only to the extent that each satisfies the above requirements:

(a) political events that occur inside or directly involve Pakistan ("Pakistan Political Force Majeure Events"), including, but not limited to:

(i) any act of war (whether declared or undeclared), invasion, armed conflict or act of foreign enemy, blockade, embargo, revolution, riot, insurrection, civil commotion, act or campaign of terrorism, or sabotage;

(ii) any Lapse of Consent that (i) shall itself have existed for twenty-six (26) Days or more, (ii) together with any and all other Lapses of Consents that have occurred in the same Agreement Year, shall have existed in the aggregate for thirty (30) Days or more in such Agreement Year, or (iii) together with any and all other Lapses of Consents that have occurred in the same and in the two immediately preceding Agreement Years, shall have existed, in the aggregate, for thirty-five (35) Days or more;

(iii) radioactive contamination or ionizing radiation originating from a source in Pakistan or resulting from another Pakistan Political Force Majeure Event;

(iv) strikes, works to rule or go-slows that extend beyond the Complex, are widespread or nationwide, or that are of a political nature, such as, by way of example and not limitation, labour actions associated with or directed against a Pakistan political party, or those that are directed against the Company (or its contractors) as part of a broader pattern of labour actions against companies or facilities with foreign ownership or management; or

(v) a nationwide shortage of fuel oil that prevents the Fuel Supplier from providing adequate deliveries of fuel oil to the Complex for more than twenty-one (21) Days as determined by the Director-General Oil, Ministry of Petroleum and Natural Resources;

73

11/2-

(b) Changes in Law;

(c) other events beyond the reasonable control of the affected Party ("Other Force Majeure Events"), including, but not limited to:

(i) uncontrollable events including, but not limited to:

(A) lightning, fire, earthquake, tsunami, flood, storm, cyclone, typhoon, or tornado;

(B) fire, explosion or chemical contamination (other than resulting from an act referred to in Section 13.1(a)(i), in which case it shall be a Pakistan Political Force Majeure Event;

(C) epidemic or plague;

(D) a Lapse of Consent unless such Lapse is a Pakistan Political Force Majeure Event;

(E) prior to the Commercial Operations Date, a delay beyond the thirtieth (30th) Day after the scheduled receipt date of the receipt of a major piece of equipment that has been timely ordered and must be manufactured expressly for a Party to perform its obligations under this Agreement, when such delay is caused solely by an accident in transportation or a strike.

(ii) political events that occur outside Pakistan and do not directly involve Pakistan, including, but not limited to:

(A) any act of war (whether declared or undeclared), invasion, armed conflict or act of foreign enemy, blockade, embargo, revolution, riot, insurrection, civil commotion, or act of terrorism;

(B) radioactive contamination or ionizing radiation originating from a source outside Pakistan and not falling within Section 13.1(a)(iii); and

(C) strikes, works to rule or go-slows that are widespread or nationwide.

(d) Force Majeure Events shall expressly not include the following conditions, except and to the extent that they result directly from a Force Majeure Event:

74

May

(i) Except as provided in Section 13.1(c)(i)(E), late delivery of machinery, equipment materials, spare parts or consumables (including fuel);

(ii) a delay in the performance of any Contractor; or

(iii) normal wear and tear or random flaws in materials and equipment or breakdowns in equipment.

## 13.2 Notification Obligations.

If by reason of a Force Majeure Event a Party is wholly or partially unable to (a) carry out its obligations under this Agreement, the affected Party shall (i) give the other Party notice of the Force Majeure Event(s) as soon as practicable, but in any event, not later than the later of fortyeight (48) hours after the affected Party becomes aware of the occurrence of the Force Majeure Event(s) or six (6) hours after the resumption of any means of providing notice between the Company and KESC and (ii) give the other Party a second notice, describing the Force Majeure Event(s) in reasonable detail and, to the extent which can be reasonably determined at the time of such notice, providing a preliminary evaluation of the obligations affected, a preliminary estimate of the period of time that the affected Party will be unable to perform such obligations and other relevant matters as soon as practicable, but in any event, not later than seven (7) Days after the initial notice of the occurrence of the Force Majeure Event(s) is given by the affected Party. When appropriate or when reasonably requested to do so by the other Party, the affected Party shall provide further notices to the other Party more fully describing the Force Majeure Event(s) and its cause(s) and providing or updating information relating to the efforts of the affected Party to avoid and/or to mitigate the effect(s) thereof and estimates, to the extent practicable, of the time that the affected Party reasonably expects it will be unable to carry out any of its affected obligations due to the Force Majeure Event(s).

(b) The affected Party shall provide notice to the other Party of (i) with respect to an ongoing Force Majeure Event, the cessation of the Force Majeure Event, and (ii) its ability to recommence performance of its obligations under this Agreement as soon as possible and in any event not later than seven (7) Days after the occurrence of each of clause (i) and (ii) above.

(c) Failure by the affected Party to give written notice of a Force Majeure Event to the other Party within the forty-eight (48) hour period or six (6) hour period required by Section 13.2(a) shall not prevent the affected Party from giving such notice at a later time; provided, however, that in such case the affected Party shall not be excused pursuant to Section 13.4 for any failure or delay in complying with its obligations under or pursuant to this Agreement until such notice has been given. If said notice is given within the forty-eight (48) hour period or six hour period required by Section 13.2(a), the affected Party shall be excused for such failure or delay pursuant to Section 13.4 for memory of the relevant Force Majeure Event.

## 13.3 Duty to Mitigate.

The affected Party shall use all reasonable efforts to mitigate the effects of a Force Majeure Event, including, but not limited to, the payment of reasonable sums of money by or on behalf of the affected Party, which sums are reasonable in light of the likely efficacy of the mitigation measures.

AL Mb

#### 13.4 Delay Caused by Force Majeure.

So long as the affected Party has at all times since the occurrence of the Force Majeure Event complied with the obligations of Section 13.3 and continues to so comply, then (i) the affected Party shall not be liable for any failure or delay in performing its obligations (other than an obligation to make a payment) under or pursuant to this Agreement during the existence of a Force Majeure Event and (ii) any performance deadline that the affected Party is obligated to meet under this Agreement shall be extended; provided, however, that no relief, including the extension of performance deadlines, shall be granted to the affected Party pursuant to this Section 13.4 to the extent that such failure or delay would have nevertheless been experienced by the affected Party had the Force Majeure Event not occurred; provided, further, that, in the case of a Force Majeure Event which damages the Complex, in no event shall the obligations of the affected Party under this Agreement to meet performance deadlines be extended beyond the end of the Restoration Period (as defined in the Implementation Agreement) determined in accordance with Article XVII of the Implementation Agreement. Other than for breaches of this Agreement by the other Party, and without prejudice to the affected Party's right to indemnification pursuant to Article XII or for payment pursuant to Article IX, the other Party shall not bear any liability for any loss or expense suffered by the affected Party as a result of a Force Majeure Event.

13.5 Payments During Force Majeure Event.

(a) Upon the occurrence of any Force Majeure Event after the Commercial Operations Date, then during the pendency of a Force Majeure Event, KESC shall pay to the Company Energy Payments for Net Electrical Output delivered during the pendency of such Force Majeure Event plus Capacity Payments in an amount equal to the amount due to the Company pursuant to Section 9.1 immediately prior to such Force Majeure Event multiplied by the FM Ratio. For the purposes of this Section 13.5, the FM Ratio shall be:

(i) in the case of a Force Majeure Event declared by the Company where the Complex is capable of partial operation, a fraction, the numerator of which is the Dependable Capacity as determined by testing in accordance with Article X as soon as practicable after the declaration of such Force Majeure Event, and the denominator of which is the Dependable Capacity determined by the most recent test prior to the Force Majeure Event conducted pursuant to Article X;

(ii) in case of Force Majeure Event declared by the Company where the Complex is not capable of operation, the FM Ratio shall be zero; and

(iii) in the case of a Force Majeure Event declared by KESC the FM Ratio shall be one (1).

(b) During the pendency of a Force Majeure Event, either Party may request that a Dependable Capacity or gross capacity test, as the case may be, be performed in order to determine the FM Ratio applicable in Section 13.5(a)(i), and thereafter such new FM Ratio shall be used to compute the payments to be made by KESC to the Company; <u>provided</u>, <u>however</u>, that no more than two (2) tests may be requested by a Party within any thirty (30) Day period during the pendency of the

Who the

Force Majeure Event. Except as provided in this Section 13.5, the Company shall not be entitled to any payments from KESC during any Force Majeure Event.

#### 13.6 Supplemental Tariff for Pakistan Political Force Majeure Events.

In the event that a Pakistan Political Force Majeure Event results in damage to, or other adverse effects on, the Complex, which is repaired or otherwise remedied by the Company, and the Implementation Agreement is not terminated pursuant to Section 17.8(a) or Section 17.8(b) thereof, the Company shall be entitled to receive Supplemental Tariff Payments from KESC in accordance with the procedures set forth in Schedule 6 to recover the difference between the restoration costs approved in accordance with Section 17.7(d) of the Implementation Agreement and any insurance proceeds received by the Company as a result of the occurrence of the Pakistan Political Force Majeure Event.

## 13.7 Supplemental Tariff for Change in Law.

In the event of the occurrence of a Change in Law (including a Change in Law that becomes applicable to the Company because of damage to and the restoration of the Complex as described in Section 13.6)) that requires a material modification or a material capital addition to the Complex, which is completed by the Company, or in lieu thereof or in addition thereto, an increase or decrease in the use or quality of consumables by the Complex, in each case, as approved by the GOP, in the manner provided by the Implementation Agreement, and the Implementation Agreement is not terminated pursuant to Section 17.8(a) or Section 17.8(b) thereof, the Company will be entitled to receive Supplemental Tariff Payments from KESC in accordance with the procedures set forth in Schedule 6. Such Supplemental Tariff Payments will allow the Company to recover the costs of complying with the Change in Law, including (i) the cost of any material modifications or material capital additions to the Complex that are necessary for the Company to come into compliance with the Change in Law and are approved in accordance with Section 17.7(d) of the Implementation Agreement and (ii) the cost of additional quantities or higher quality of consumables that can be directly attributed to compliance by the Company with the Change in Law. Any reduction in cost due to a decrease in the use or quality of consumables by the Company with the Change in Law.

12h Mu

# ARTICLE XIV TAXES

# 14.1 Taxes Applicable to the Company.

Subject to the terms of the Implementation Agreement, all present and future federal, provincial, municipal or other lawful taxes, duties, levies, or other impositions applicable to the Company, the Complex, the Project and the Company's other assets shall be paid by the Company in a timely fashion. Nothing herein, however, shall in any way limit or override any provisions of this Agreement, including Schedule 6, that allow certain taxes and charges to be treated as "Pass-Through Items."

#### 14.2 <u>Taxes Applicable to KESC</u>.

All present and future federal, provincial, municipal or other lawful taxes, dutics, levies, or other impositions applicable to KESC arising from or in connection with its rights and obligations under this Agreement shall be paid by KESC in a timely fashion.

Ale les

## <u>ARTICLE XV</u> RESOLUTION OF DISPUTES

# 15.1 Resolution by Parties.

(a) In the event that a Dispute arises, the Parties shall attempt in good faith to settle such Dispute by mutual discussions within thirty (30) Days after the date that the disputing Party gives written notice of the Dispute to the other Party, which may include referring the Dispute to the Operating Committee for a specified time period; provided, however, that if the Dispute involves the amount of an invoice and after ten (10) Business Days of mutual discussion either Party believes in good faith that further discussion will not resolve the Dispute to its satisfaction, such Party may immediately refer the matter to the expert for consideration pursuant to Section 15.2;

(b) In the event that the Dispute is not resolved in accordance with Section 15.1(a), either Party may refer the Dispute to the chief executive officer or chief operating officer of the Company and the General Manager - System Operations of KESC (or such other official authorized by KESC) for further consideration. In the event that such individuals are unable to reach agreement within fifteen (15) Days, or such longer period as they may agree, then either Party may refer the matter to an expert in accordance with Section 15.2 hereof or, if the Dispute is not of a type required to be referred to an expert under Section 15.2, commence arbitration of the Dispute in accordance with Section 15.3.

15.2 Mediation by Expert.

(a) In the event that the Parties are unable to resolve a Dispute in accordance with Section 15.1, then either Party, in accordance with this Section 15.2, may refer the Dispute to an expert for consideration of the Dispute and to obtain a recommendation from the expert as to the resolution of the Dispute; <u>provided</u>, <u>however</u>, that with respect to Disputes that involve the amount of an invoice, either Party may require that an expert be appointed in accordance with the provisions of Section 15.2(b) to consider a Dispute in advance of the occurrence of such a Dispute, and shall nominate a person it proposes to be the expert.

(b) The Party initiating submission of the Dispute to the expert shall provide the other Party with a notice stating that it is submitting the Dispute to an expert and nominating the person it proposes to be the expert. The other Party shall, within fifteen (15) Days of receiving such notice, notify the initiating Party whether such person is acceptable. If the Party receiving such notice fails to respond or notifies the initiating Party that the person is not acceptable, the Parties shall meet and discuss in good faith for a period of ten (10) Days to agree upon a person to be the expert. If the Parties are unable to agree, the responding Party shall by the end of such ten (10) Day period nominate a person to be an expert, whereupon the two nominated experts shall meet and agree upon a third person who shall be the expert.

(c) (i) Consideration of the Dispute by an expert shall be initiated by the Party who is seeking consideration of the Dispute by the expert submitting to both the expert and the other Party written materials setting forth:

79

Ale

- (A) a description of the Dispute;
- (B) a statement of the Party's position; and
- (C) copies of records supporting the Party's position.

(ii) Within ten (10) Days of the date that a Party has submitted the materials described in Section 15.2(c)(i), the other Party may submit to the expert:

- (A) a description of the Dispute;
- (B) a statement of the Party's position; and
- (C) copies of any records supporting the Party's position.

The expert shall consider any such information submitted by the responding Party within the period provided in Section 15.2(c)(ii) and, in the expert's discretion, may consider any additional information submitted by either Party at a later date.

(d) The Parties shall not be entitled to apply for discovery of documents, but shall be entitled to have access to the other Party's relevant records and to receive copies of the records submitted by the other Party.

(e) Each Party shall designate one person knowledgeable about the issues in Dispute who shall be available to the expert to answer questions and provide any additional information requested by the expert. Except for such person, a Party shall not be required to, but may, provide oral statements or presentations to the expert or make any particular individuals available to the expert.

(f) Except as provided in Section 15.2(h) with respect to the payment of costs, the proceedings shall be without prejudice to any Party and any evidence given or statements made in the course of this process may not be used against a Party in any other proceedings. The process shall not be regarded as an arbitration and the laws relating to commercial arbitration shall not apply. Unless the Parties agree in a writing signed by both Parties at the time the expert is selected stating that the decision of the expert will be binding, the determination of the expert shall not be binding.

(g) When consideration of the Dispute by an expert is initiated, the expert shall be requested to provide a recommendation within fifteen (15) Days after the ten (10) Day response period provided in Section 15.2(c)(i) above has run. If the expert's recommendation is given within such fifteen (15) Day period, or if the expert's recommendation is given at a later time and neither Party has at such time initiated any other proceeding concerning the Dispute, the Parties shall review and discuss the recommendation with each other in good faith for a period of ten (10) Days following delivery of the recommendation before proceeding with any other actions.

(h) If a Party does not accept the recommendation of the expert with respect to the Dispute, it may initiate arbitration proceedings in accordance with Section 15.3, <u>provided</u>, <u>however</u>, that prior to initiating the arbitration proceedings it shall have paid all costs of the expert (including the reimbursement of any costs paid to the expert by the other Party) and all out-of-pocket

80

costs of the other Party. Similarly if the expert has not submitted its recommendation within the time period provided in Section 15.2(g), a Party may initiate arbitration proceedings in accordance with Section 15.3, provided that prior to initiating the arbitration proceedings it shall have paid all costs of the expert (including the reimbursement of any costs paid to the expert by the other Party),

Except as provided in Section 15.2(h), the costs of engaging an expert shall (i) be borne equally by the Parties and each Party shall bear its own costs in preparing materials for, and making presentations to, the expert.

#### 15.3 Arbitration.

- (a) Any Dispute arising out of or in connection with this Agreement and not resolved following the procedures described in Sections 15.1 and 15.2 shall, except as hereinafter provided, be settled by arbitration in accordance with the Rules of Procedure for Arbitration Proceedings (the "ICSID Rules") of the International Centre for the Settlement of Investment Disputes (the "Centre") established by the 1965 Convention on the Settlement of Investment Disputes between States and Nationals of other States (the "Convention") and the Parties hereby consent to Arbitration thereunder. For the purposes of consenting to the jurisdiction of the Convention, the Parties agree that the Company is a foreign controlled entity unless the amount of the voting stock in the Company held by Foreign Investors should decrease to less than thirty-five percent (35%) of the outstanding voting stock of the Company other than due to the purchase of shares by any of the Federally Controlled Entities, or the Provincial Government or any entity controlled by or under the influence of the Provincial Government.
- (b) Notwithstanding the foregoing, unless and until the GOP has implemented the Convention by an Act or an Ordinance confirmed by an Act, or if for any reason the Dispute cannot be settled in accordance with the ICSID Rules, whether because the Company should not be agreed to be a foreign controlled entity, or the request for arbitration proceedings is not registered by the Centre, or the Centre fails or otherwise, such Dispute shall be finally settled by arbitration under the Rules of Arbitration of the International Chamber of Commerce (the "ICC Rules") by one or more arbitrators appointed in accordance with the ICC Rules,
- Any arbitration shall be conducted in Karachi, Pakistan, and unless otherwise agreed (c) by the Parties, the number of arbitrators shall be one; provided, however, that if the Company desired that arbitration be conducted outside of Pakistan, the arbitration shall be conducted in Singapore and the Company shall pay all costs of the arbitration as and when incurred by KESC, including the out of pocket costs of the arbitration of both Parties (excluding any award made by the arbitrator) in excess of the costs that would have been otherwise incurred by KESC had the arbitration been conducted in Karachi, Pakistan. The arbitrator shall resolve any Disputes as to whether a cost would have been incurred in connection with the arbitration in Karachi, Pakistan (the "Base Costs") or was associated with the removal to Singapore (the "Incremental Costs"). The arbitrator may order that KESC bear its own Incremental Costs in part or in full if he finds that KESC's claim or defence in the Arbitration was spurious and without any llb

Ale

81

merit whatsoever, and KESC shall pay the amount ordered; <u>provided</u>, <u>however</u>, that if a matter in Dispute involves a sum of seven million Dollars (\$7,000,000) or more, or the validity or enforceability of this Agreement, or the termination of this Agreement, arbitration shall, unless otherwise agreed by the Parties, be conducted in Singapore, and, in such case, each Party shall pay its own costs of arbitration as and when incurred, unless such costs are ordered by the Arbitrator to be paid by one Party, in which case they shall be paid by such Party.

#### 15.4 Commercial Acts: Sovereign Immunity.

KESC unconditionally and irrevocably agrees that the execution, delivery and performance by it of this Agreement and those agreements included in the Security Package to which it is a party constitute private and commercial acts. In furtherance of the foregoing, KESC hereby irrevocably and unconditionally agrees that: (i) should any proceedings be brought against KESC or its assets, other than the Grid System, electric generation assets and equipment, electric distribution assets, other assets necessary for the fulfilment of its duties and responsibilities under the Pakistan Water and Power Development Authority Act of 1958 and assets protected by the diplomatic and consular privileges under the 1978 Immunity Act of the United Kingdom or the 1976 Sovereign Immunities Act of the United States or any analogous legislation (collectively, the "Protected Assets") in any jurisdiction in connection with this Agreement or any of the transactions contemplated by this Agreement, no claim of immunity from such proceedings will be claimed by or on behalf of KESC on behalf of itself or any of its assets (other than the Protected Assets); (ii) it waives any right of immunity which it or any of its assets (other than the Protected Assets) now has or may in the future have in any jurisdiction in connection with any such proceedings; and (iii) consents generally in respect of the enforcement of any judgment against it in any such proceedings in any jurisdiction to the giving of any relief or the issue of any process in connection with such proceedings (including without limitation, the making, enforcement or execution against or in respect of any of its assets other than the Protected Assets). The Company irrevocably waives any and all rights it may have to enforce any judgement or claim against the Protected Assets.

UB AL

#### ARTICLE XVI NOTICES

#### 16.1 Addresses and Addressees.

Except as otherwise expressly provided in this Agreement, all notices or other communications to be given or made hereunder shall be in writing, shall be addressed for the attention of the persons indicated below and shall either be delivered personally or sent by courier, registered or certified mail or facsimile. The addresses for service of the Parties and their respective facsimile numbers shall be:

If to the Company:

Attention:

Chief Executive Tapal Energy Limited Ameejee Chambers Campbell Street P.O. Box No. 51 Karachi 74200

Facsinile No.: (92-21) 262-7817

# If to KESC:

Attention:

General Manager Karachi Electric Supply Corporation Ltd., 7th Floor, State Life Building # 11 Abdullah Haroon Road Karachi, Pakistan

All notices shall be deemed delivered (a) when presented personally, (b) if received on a business Day for the receiving Party, when transmitted by facsimile to the receiving Party's facsimile number specified above and, if received on a Day that is not a Business Day for the receiving Party, on the first Business Day following the date transmitted by facsimile to the receiving Party's facsimile number specified above, (c) one (1) Day after being delivered to a courier for overnight delivery, addressed to the receiving Party, at the address indicated above (or such other address as such Party may have specified by written notice delivered to the delivering Party at its address or facsimile number specified above in accordance herewith) or (d) five (5) Days after being deposited in a regularly maintained receptacle for the Postal Service in Pakistan, postage prepaid, registered or certified, return receipt requested, addressed to the receiving Party, at the address indicated above (or such other address as the receiving Party may have specified above in accordance herewith). Any notice given by facsimile shall be confirmed in writing delivered personally or sent by registered or certified mail, but the failure to so confirm shall not void or invalidate the original notice if it is in fact received by the Party to which it is addressed.

# 16.2 Changes of Address.

.

,

ц.,

ŧ.

:

Any Party may by notice change the addressees and/or addresses to which such notices and communications to it are to be delivered or mailed.

ph the

<u>~</u>~

 $\sim$ 

,

## ARTICLE XYII MISCELLANEOUS PROVISIONS

## 17.1 Amendment.

This Agreement can be amended only by agreement between the Parties in writing. No amendment of this Agreement will be effective without the prior written consent of the GOP if such amendment increases or potentially increases materially the liability of the GOP under the Implementation Agreement or the Guarantee.

17.2 <u>Headings</u>.

The headings contained in this Agreement are used solely for convenience and do not constitute a part of this Agreement nor shall such headings be used in any manner to aid in the construction or interpretation of this Agreement.

17.3 Third Parties.

This Agreement is intended solely for the benefit of the Parties hereto and, except for rights expressly granted to Lenders, nothing in this Agreement shall be construed to create any duty to, standard of care with reference to, or any liability to, any person not a Party to this Agreement.

17.4 <u>No Waiver</u>.

(a) No waiver by either Party of any default or defaults by the other Party in the performance of any of the provisions of this Agreement:

(i) shall operate or be construed as a waiver of any other or further default or defaults whether of a like or different character; or

(ii) shall be effective unless in writing duly executed by a duly authorized representative of such Party.

(b) Neither the failure by either Party to insist on any occasion upon the performance of the terms, conditions and provisions of this Agreement nor time or other indulgence granted by one Party to the other shall act as a waiver of such breach or acceptance of any variation or the relinquishment of any such right or any other right hereunder, which shall remain in full force and effect.

(c) Settlement or waiver of any Dispute or breach related to Articles IV, XI, and XIV, and Sections 3.1 and 3.6 shall be effective only if agreed to, in writing, by both KESC and the GOP.

b Ale

## 17.5 Relationship of the Parties.

This Agreement shall not be interpreted or construed to create an association, joint venture, or partnership between the Parties or to impose any partnership obligation or liability upon either Party. Neither Party shall have any right, power or authority to enter into any agreement or undertaking for, or act on behalf of, or to act as or be an agent or representative of, or to otherwise bind, the other Party.

## 17.6 <u>Survival</u>.

Cancellation, expiration or earlier termination of this Agreement shall not relieve the Parties of obligations that by their nature should survive such cancellation, expiration or termination, including, without limitation, warranties, remedies, promises of indemnity and confidentiality.

17.7 Language.

The language for the purpose of administering this Agreement shall be English.

17.8 Choice of Law.

This Agreement and the rights and obligations hereunder shall be interpreted, construed and governed by the laws of England (without regard to conflicts of law rules).

#### 17.9 Entirety.

This Agreement and the Schedules attached hereto are intended by the Parties as the final expression of their agreement on the matters contained herein and are intended also as a complete and exclusive statement of the terms of their agreement with respect to the Net Electrical Output and Dependable Capacity sold and purchased hereunder. All prior written or oral representations, understandings, offers or other communications of every kind pertaining to the sale or purchase of Net Electrical Output and Dependable Capacity hereunder to KESC by the Company are hereby abrogated and withdrawn.

17.10 Assignment.

(a) This Agreement may not be assigned by either Party other than by mutual agreement between the Parties in writing.

(b) Notwithstanding the foregoing, for the purpose of financing the Project, the Company may assign to, or grant a security interest in favour of, the Lenders in its rights and interests under or pursuant to (i) this Agreement, (ii) any agreement or document included within or contemplated by the Security Package, (iii) the Complex, (iv) the Site, (v) the movable, immovable and intellectual property of the Company, or (vi) the revenues or any of the rights or assets of the Company. The Company shall not create any security over its rights and interests under this Agreement without the prior written consent of KESC except as already provided above.
(c)The Lenders shall have no obligation to KESC under this Agreement until such time as the Lenders or their designees succeed to the Company's interest under this Agreement, whether by exercise of their rights or remedies under the Financing Documents or otherwise, in which case the Lenders or their designees shall assume liability for all of the Company's obligations under this Agreement, including payment of any amounts due and owing to KESC for payment defaults by the Company under this Agreement (other than, so long as the liability insurance required by Section 11.1 has been and is in effect, damages or penalties incurred by the Company under Section 12.2(b), arising during the period prior to the Lenders's or such designees' succession to the Company's interest in and under this Agreement; provided that any liability of the Lenders or their designees shall be strictly limited to the Lenders' interest in the Complex. Notwithstanding the foregoing, KESC shall not be prevented from terminating this Agreement in respect of any liability of the Company arising under Section 12.2(b) that is not assumed by the Lenders. Except as otherwise set forth in the immediately preceding sentence, none of the Lenders or their designees shall be liable for the performance or observance of any of the obligations or duties of the Company under this Agreement, nor shall the assignment by the Company of this Agreement to the Lenders give rise to any duties or obligations whatsoever on the part of any of the Lenders owing to KESC.

Upon notification by the Lenders or the Agent to KESC of the occurrence and continuance of an event of default under the Financing Documents and the succession of the Lenders to the Company's interests in and under this Agreement, the Lenders shall have the right, among others, to (i) take possession of the Complex and, prior to the Commercial Operations Date, complete construction of the Complex and operate the same and, after the Commercial Operations Date, operate the same, and (ii) cure any continuing Company Event of Default under this Agreement as provided in Section 4.5. Notwithstanding the foregoing, the Lenders shall have no obligation to cure any Company Event of Default that is not capable of being cured including, but not limited to, a default under Section 4.2 (f), (i), (j), (k) or (m) and no right will exist for KESC to terminate this Agreement based upon such Company Events of Default occurring prior to the delivery of the Lenders's notice. Without the requirement of obtaining any further consent from KESC, upon the exercise by the Lenders or their designees of any of the remedies set forth in the Financing Documents, the Lenders may assign their rights and interests and the rights of the Company under this Agreement to a Transferee acceptable to GOP (hereinafter defined) so long as such Transferee shall assume all of the obligations of the Company under this Agreement.

Upon such assignment and assumption, the Lenders shall be relieved of all obligations under this Agreement arising after such assignment and assumptions.

(d) As used herein, a "Transferee" shall be a person who (i) either is an experienced and qualified power plant operator or who shall have agreed to engage the services of a person who is an experienced and qualified power plant operator, (ii) shall have paid all amounts, if any, then due and payable to KESC under this Agreement, and (iii) shall have expressly assumed in writing for the benefit of KESC the ongoing obligations of the Company under this Agreement (including the obligation of the Company to maintain and operate the Complex in accordance with the requirements of this Agreement).

(e) At the request of the Company, delivered to KESC not less than thirty (30) Days in advance KESC shall execute and deliver at the Financial Closing, all such acknowledgements

87

No Alt

to the Lenders or their designees of any security created in accordance with this Section 17.10 as are reasonably requested by the Company and the Lenders to give effect to the foregoing.

(f) Notwithstanding the above, KESC shall have the right to assign this Agreement to any entity or entitles assuming all or part of KESC's rights and obligations in connection with the distribution of Net Electrical Output; <u>provided</u>, that the GOP without interruption guarantees the performance of the succeeding entity or entities on the same terms and conditions as the Guarantee or such other commercial security is provided for the obligations of the succeeding entity or entities that in the reasonable business judgment of the Company provides an adequate alternative to the Guarantee and all of KESC's obligations under this Agreement are assigned pursuant to law to or contractually assumed, through a novation, by one or more entities, each of which has the legal capacity and appropriate commercial function to perform such obligations.

### 17.11 Confidentiality.

(a) Each of the Parties and their contractors, consultants and agents shall hold in confidence all documents and other information whether technical or commercial supplied to it by or on behalf of the other Party relating to the design, construction, insurance, operation, maintenance, management and financing of the Complex and all information and documents obtained by it in the course of any inspection performed in accordance with the terms of this Agreement, and shall not, save as required by law or appropriate regulatory authorities, prospective lenders to, or investors in, the Company and their professional advisers, publish or otherwise disclose or use the same for its own purposes otherwise than as may be required to perform its obligations under this Agreement. Notwithstanding the above, nothing herein contained shall preclude the use of provisions similar to those contained in this Agreement and the other Agreements referred to herein and in agreements prepared and issued in connection with other projects.

(b) The provisions of paragraph (a) above shall not apply to:

(i) any information in the public domain otherwise than by breach of this Agreement;

(ii) information in the possession of the receiving Party thereof before divulgence as aforesaid, and which was not obtained under any obligation of confidentiality; and

(iii) information obtained from a third party who is free to divulge the same, and which is not obtained under any obligation of confidentiality.

# 17.12 Successors and Assigns.

This Agreement shall be binding upon, and inure to the benefit of, the Parties hereto and their respective successors and permitted assigns.

. Hb

Aler.

#### 17.13 No Liability for Review.

No review and approval by KESC of any agreement, document, instrument, drawing, specifications or design proposed by the Company shall relieve the Company from any liability that it would otherwise have had for its negligence in the preparation of such agreement, document, instrument, drawing, specification or design or failure to comply with the applicable Laws of Pakistan with respect thereto, or to satisfy the Company's obligations under this Agreement, the Implementation Agreement and the other documents comprising the Security Package, nor shall KESC be liable to the Company or any other person by reason of its review and approval of an agreement, document, instrument, drawing, specification, or design,

#### No Third Party Beneficiaries. 17.14

This Agreement shall not confer any right of suit or action whatsoever on any third party except for the rights granted to the Lenders.

#### Affirmation. 17,15

The Company and KESC declare and affirm that neither Party has paid nor has it undertaken to pay and that it shall in the future not pay any bribe, pay-offs, kick-backs or unlawful commission and that it has not in any other way or manner paid any sums, whether in Pakistani currency or foreign currency and whether in Pakistan or abroad, or in any other manner given or offered to give any gifts and presents in Pakistan or abroad to any person or company to procure this Agreement, and the Company and KESC undertake not to engage in any of the said or similar acts during the term of and relative to this Agreement.

#### Approval Not to be Unreasonably Withheld or Delayed. 17.16

Unless otherwise provided herein with respect to a particular provision, whenever the acceptance, consent or approval of a Party is required herein, such acceptance, consent or approval shall not be unreasonably withheld or delayed by such Party.

#### 17.17 Double Jeopardy.

A final, non-appealable order issued in a proceeding initiated by the GOP and based upon a claim of breach of the Implementation Agreement shall be with prejudice to any proceedings against the Company that KESC could otherwise bring for breach by the Company of substantially the same obligations under this Agreement. Nothing in this Section shall prevent KESC and the GOP from separately initiating proceedings to terminate this Agreement and the Implementation Agreement,

89

respectively, pursuant to Sections 4.2 and 4.4 of this Agreement and Sections 19.1 and 19.3 of the Implementation Agreement.

IN WITNESS WHEREOF the Parties have executed and delivered this Agreement as of the date first above written.

	THE KARACHI ELECTRIC SUPPLY	
	CORPORATION LIMITED	
	By:	
	General Melleyer (GIV & C)	
	Name: K.S.S.C. Lts. Karsoul.	
	Title:	
	AXIILAPULLAL	
	$\frac{C^{m+2}}{C^{n+1}} \rightarrow \frac{1}{2} \mathcal{L}$	
	TAPAL ENERGY LIMITED	
	e and a second	
	By:	
131		
	Name: MOIZ TAPAL	
College	Title: Chif Escenture	

Ľ

### MINIMUM FUNCTIONAL SPECIFICATIONS

The Complex consists of one (1) land-based powerplant comprised of twelve (12) residual fuel oil ('RFO') fired diesel generating sets (each a 'Unit') with the following design ratings:

1.	Gross Capacity under ISO 3046 (Latest Version as of the date of this Agreement)		
	a. Diesel generating unit at alternator terminals	10,500 kWe per Unit	
	b. Multiplied by twelve (12) Units	$126,000  kW_e$ for the Complex	
2,	Derating from ISO standard conditions to Reference Co	nditions 2,500 kWe for the Complex	
3.	Gross Capacity after derating	123,500 kWe for the Complex	
4.	Estimated auxiliary consumption for the Complex	4,000 kWe for the Complex	
5.	Estimated Dependably Capacity of the Complex at		
	the 132kV busbar under Reference Conditions	119,500 MWe	

The Site is located off Hub River Road, South of Hub Chowki, in the District of Karachi West, Sindh Province.

The Site area is characterized as tropical. Average rainfall is 700 mm per year, but this amount falls in short durations. Reference Conditions at the Site are:

AMBIENT AIR TEMPERATURE	$= T_{m}$	= 30  deg. C = 303  K
CHARGE AIR COOLANT TEMPERATURE	= T <sub>cra</sub>	= 40  deg. C = 313  K
TOTAL BAROMETRIC PRESSURE	<i>≕p</i> r	= 1031.2 mbar
RELATIVE HUMIDITY		= 80 %

Jub Ale

A maximum earthquake design factor of 0.1 g will be utilized for the design of the plant buildings and structures and the design wind speed will be 40m/sec.

The Site will be at elevation <65 m with respect to mean sea level (MSL). All structures will have a ground floor elevation of at least 20 m above MSL. Access to the Site will be provided by a road. The main power block consists of twelve (12) RFO fired diesel generating sets. Other plant buildings and structures outside of the main power block include, but are not limited to, the following:

91

Fuel Storage Area; Combined Fuel Treatment / Boiler (for heating RFO only) / Water Treatment House; Fire Fighting Basin; Workshop / Stores; Office Building; Switchyard; Guardhouse.

The Complex will use a simple cycle diesel generating sets with waste heat recovery to heat the RFO systems.

Cooling will be supplied via a radiator cooling system. The primary fuel will be RFO, with Light Fuel Oil ('LFO') being used for flushing the engines and for LFO fired boilers.

Each generator will be nominally rated at 13,125 kVA, 0.8 lagging and 0.9 leading power factor, 11,000 V, 3 phase, 50 cycle, not less than 0.55 short circuit ratio. Each generator will be air cooled.

The Complex will be capable of operation within a voltage range of  $\pm 10\%$  on the 132 kV system.

The Complex will have three main step-up transformers, each rated not less than 58.6 MVA, 132/11kV, ONAF.

Interconnection with the KESC system will be via a 132 kV, switchyard located on the Complex.

A common control room is provided to monitor and control the Complex. Operator interfaces for control of the Complex will be via control panel and PC operating stations and each Unit will be capable of being started and stopped locally. The plant control systems include a data acquisition system.

Fuel supply to the Complex will be via road tanker delivery. The Site will have sufficient RFO storage capacity to support operation of the Complex for the equivalent of 100 % of full load for 30 Days.

All material, plant, equipment and machinery incorporated in the construction of the Complex shall be new and unused.

# TECHNICAL LIMITS

# l. <u>Design Limits</u>

F

# 1.1 Unit Starts

(a) The notice required by the Company to start-up the Complex or Unit(s) and synchronize to the KESC Grid System will vary according to the length of time the Complex or Unit(s) has been shutdown. Table I below shows the length of notice required against various periods of shutdown.

# <u>Table 1</u>

	Length of Shutdown	Notice requi	ired to synchronize
(i)	Not more than 2 hours		20 minutes
(ii)	More than 2 hours but . not more than 8 hours		60 minutes
(iii)	More than 8 hours but not more than 32 hours		2 hours
(iv)	More than 32 hours but not more than 150 hours	•	4 hours
(v)	More than 150 hours		7 hours

(b) For the purposes of this Schedule start up of the Complex or Unit(s) is classified as follows:

"Hot Start" - A start following a shutdown period as per (i), (ii), (iii) or (iv) above.

"Cold Start" -

A start following a slutdown period as per (v) above.

No Ale

and the reference to starts, starting or started means the process of starting the Complex or Unit(s) and synchronizing it with the KESC Grid System.

(c) The notice required to synchronize under item (i) above shall apply provided the previous shutdown was not the result of a trip.

(d) Starting of the Complex shall be subject to the following limits for each Unit:

# Table 2

Type of Starts	Total agg maximum of star	number	Maximum number of starts per <u>year</u>
Hot Starts	j	No coobiad limi	tationa
Cold Starts	ſ	No technical limitations	

# 1.2 <u>Complex loading</u>

(a) The Complex load ramping rate is the steady rate at which the load can be raised. The maximum load ramping rates are shown below in Table 3A:

# Table 3A

	pl <b>ex L</b> oad <u>Range</u>	Cold Start <u>% per minute</u>	Hot Start <u>% per minute</u>
(i)	0 ≤ 25	5	6
(ii)	> 25 ≤ 50	5	6
(iii)	> 50 ≤ 100	4	5

Complex load percentages in this Table 3A refer to the sum of the load(s) at the generator terminals of the Units already sychronized to the KESC Grid System as a percentage of gross capacity of the Complex specified in Section 3 of Schedule 1.

Us Ate

1 ~

(b) The Unit load ramping rate is the steady rate at which the load can be raised. The maximum Unit load ramping rates are shown below in Table 3B:

Ľ

# <u>Table 3B</u>

Unit le	oad range, %	Cold Start (MW / min.)	Hot Start (MW / min.)
(i)	0 ≤ <b>2</b> 5	0.8	ł
(ii)	$> 25 \le 50$	D,8	1
(iii)	$> 50 \le 100$	0,8	1

Unit load percentage in the Table 3B refers to the load at the generator terminals of the Unit as a percentage of the gross capacity of the Unit specified in Section 1.a. of Schedule 1.

- (c) Step changes in Despatched load of up to 20% are allowable provided that Complex load is greater than 30% of gross capacity of the Complex specified in Section 3 of Schedule 1. After such step change the new Complex load must be held constant for 5 minutes for stabilization purposes, or for a prorata period for lesser step changes.
- (d) The Complex can withstand a full load rejection and remain in a safe condition. Provided the Complex auxiliaries are operated continuously, the Complex can be re-synchronized within one hour provided that the reason for the load rejection has been removed.
- (e) The Complex minimum continuous loading shall be 6% of the gross capacity of the Complex specified in Section 3 of Schedule 1.
- 1.3 Frequency, Power Factor, Voltage Limits and Droop Settings
  - (a) The Complex will operate at 100% load with a power factor in the range 0.8 lagging to 0.9 leading which range shall not be exceeded. At 0% load the Complex has a Reactive Power capability of 144 MVAR lagging and 91.2 MVAR leading. At 100% load the Complex has a Reactive Power capability of 104.4 MVAR lagging and 72 MVAR leading.
  - (b) The Complex can operate within the range  $\pm$  10% on the 132 kV high voltage system which range shall not be exceeded.

Jub Ales

- (c) The Complex can operate within the frequency range 47 Hertz to 53 Hertz which range shall not be exceeded.
- (d) The Complex will be subject to tripping if frequency and/or voltage fluctuations outside the ranges stated in 1.3(b) and 1.3(c) above occur.
- (e) The Unit governor droop is adjustable in the range 0% to 10%. The automatic voltage regulator droop setting is adjustable in the range  $\pm$  10% of rated voltage with a droop characteristic of  $\pm$  0.5%.
- 1,4 General
  - (a) The Company shall advise KESC of any temporary operating constraints and limits which may from time to time apply to the Complex.

## 2. Design Maintenance Limits

.....

The cycle of Scheduled Outages is set out in Table 4 below together with a manufacturer's recommended durations for such inspections.

### TABLE 4

Major items and nature of maintenance per 12 Month periods	Operating hours interval for each maintenance activity	Estimated duration of each activity <u>per unit</u>
Change of Oil in TurboCharger	1000	2 hours
Change of Fuel Filters	1000	4 hours
Change Lube Oil Filters	1000	4 hours
Check of Valve Clearance	1000	5 hours
Check of Alarm and Stop/Safety D	evices 1000	15 hours
Inspection of Injectors	1000	10 hours
Change Oil in Governor	2000	1 hour
Cleaning of Air Coolers	2000	10 hours
Inspection of Crankcase	4000	2 hours
Check of Crankshaft Alignment	4000	4 hours
Inspection of TurboCharger	4000	5 hours
Inspection of Pistons	8000	2 hours(per piston)
Inspection of Engine Driven Pump	s 4000	10 hours

Mb Ale

Inspection of Thermostatic Valves

F

4000

8 hours

720 hours

Required time per year (including other activites)

Scheduled Outages thereafter continue on a one year cycle which must be maintained.

The scheduling of maintenance inspections will be compatible with manufacturer's requirements. All inspections required in accordance with manufacturer's specifications will be carried out during Scheduled Outages. In addition to the above annual Scheduled Outages, 720 hours of Scheduled Outages shall be allowed for every fifth (5th) year for overhaul items of work as shown in Table 5.

# <u> TABLE 5</u>

		Estimated duration of each activity <u>per unit</u>	
Piston-overhaul	≈-40,000		
Cylinder liner overhaul	≈ 40,000	274	
TurboCharger overhaul	≈ 40,000	120	
Fuel Injection Pump overhaul	≈ 40,000	18	
Inspection / overhaul of Main Bearings	≈ 40,000	108	
Check of Damper	≈ 40,000	5	

## 3. Prudent Utility Practice

Notwithstanding anything to the contrary, the Company shall operate and maintain the Complex in accordance with Prudent Utility Practices.

4. While the Complex is under the control of AGC, KESC shall not remotely control the Start-up, synchronization to the Grid System or shutdown of any of the individual diesel generating units at the Complex.

Ub Pk

# INTERCONNECTION FACILITIES AND TRANSMISSION FACILITIES

- 1. Interconnection and Transmission Facilities
  - The connection between the Complex and KESC's network shall be through (a) 132 kV double circuit transmission lines, one originating from the Baldia grid station and the other from Hub Chowki grid station. The transmission line each will terminate in the substation of the Complex; the location of which is shown on the Site plan attached as Figure 1. The circuits of the transmission lines will connect at bushings provided by the Company as shown on the single line diagram of the Complex substation attached as Figure 2. The boundary of responsibility between the Company and KESC will be at the top of the bushings (the "Interconnection Point"). The Company will provide KESC with an earth connection from the earthing system of the Complex. KESC will install the Metering System (on the 132 kV side of the Complex's transformers) which together with the transmission lines referred to above within the Site boundary shall comprise the "Interconnection Facilities." This equipment will remain the property of KESC and shall be commissioned and maintained thereafter by KESC.
  - (b) <u>Protection</u>. A carrier intertripping circuit for each transmission line shall be provided between the line circuit breakers at the Complex owned by the Company and the line circuit breakers at Baldia and Hub Chowki grid stations owned by KESC.

### 2. <u>Design Data</u>

The following design data has been provided by the Company to KESC to enable completion of KESC of the design of the Interconnection Facilities and the Transmission Facilities.

- 2.1. Generator Design Data
- (a) <u>Rating</u>

Rating Power factor

Number of phases Number of poles

plo Ale

98

13,125 kVA 0.8 lagging 0.9 leading 3 10

Frequency Rated speed Terminal voltage Short circuit ratio at rated MVA Excitation system		50 Hz ± 6 % 600 rpm 11 kV ± 10 % not < 0.55 brushless
(b) <u>Generator Reactances</u> (at the rated MVA & kV base)		
Unsaturated direct axis synchronous reactance Saturated direct axis sub-transient reactance Saturated direct axis transient reactance Negative phases sequence reactance Zero phase sequence reactance Leakage reactance		190 % 22.7 % 33.8 % 24.7 % 12.8 % 0.01 per unit
(c) Generator time constants		
Direct axis open circuit time constant Direct axis open circuit sub-transient time constant Direct axis short-circuit trans-transient time constant Direct axis short circuit sub-transient time constant		4.500 seconds 0.037 seconds 1.000 seconds 0.035 seconds
(d) Inertia constant		
Diesel engine generator	J = H =	5,050 kgm <sup>2</sup> * 0.69 sec
* Definition of J:	$\mathbf{l} =$	$0.25 \text{ GD}^2$
Note: The above design values will have tolerances	as specified i	n the relevant IEC standards.

# 2.2 Excitation System

Ē

Excitation of the main generator is provided by a brushless system using booster transformers. The excitation control system comprises of one automatic voltage regulator (AVR) and one manual excitation control (MEC). During normal operation, the whole excitation system is subject to automatic control by means of AVR.

**Technical Characteristics** 

(i) Voltage setting range for AVR operation:  $\pm 10\%$ 

Alle Ale

- (ii) Voltage adjusting range for MEC operation: 90-110 %
- (iii) Generator terminal voltage is held within  $\pm$  10% from no load to full load at rated frequency
- (iv) Under the maximum direct current supplied from the excitation system for a specified time, the ceiling voltage to the generator field voltage is 2.0 per unit
- (v) The transfer function diagram of the AVR to manual excitation system as mentioned above is to be provided.

# 2.3 <u>Generator Transformer</u>

MVA rating ONAF Rated voltage On-Load Tap changers Frequency Connection Group B.I.L. 11 kV side B.I.L. 132 kV side Insulation levels for impulse Insulation levels for AC withstand Type of cooling Temperature rise wdg./oil Maximum system short circuit power Impedance and Losses at reference power at nominal tap	58.6 MVA 132 kV/11 kV ± 10 % ± 13 Taps 50 Hz ± 6 % Ynd5 75 kV 650 kV 145 kV 275 kV ONAN / ONAF 55 / 50 °C 10 GVA 58.6 MVA 12 % UV to LV
•	275 kV
• •	55/50 °C
, —	10 GVA
• •	58,6 MVA
Impedance	12 % HV to LV
No Load Losses	30 kW
Load Losses	230 kW
Auxilary Losses	4 kW
Positive Reactance	$U_{\rm X} = 13 \%$
Zero Sequence Reactances	$Z_0 = 9.6 \%$

\* i HE-L (Leakage)

- \* ii. HE-T (Leakage)
- \* iii L-T (Leakage)
- Magnetizing reactance at rated voltage (from H.V. terminal)
- \* X air core (from H.V. terminal)
- \* X air core (from L.V. terminal)
- Saturation curve at no load V (rms)
- \* Versus I (ms)

\* Data to be supplied by \_\_\_\_\_.

~ Alls Ale

- 2.4 The governor droop will be adjustable from 0 % to 10 % and is designed to operate over the frequency range 47 to 53 Hz.
- 2.5 The AVR droop setting is  $\pm 0$  to 10 % of rated voltage, with a droop characteristic of  $\pm 0.5$  %.

LU1 Ale -

1

1

.

:----





. .

. .

I a support a construction definition of a construction

:

### COMMISSIONING AND TESTING

## (a) <u>Delivered Capacity</u>

During Commissioning the Actual Initial Dependable Capacity (AIDC) of the Complex will be demonstrated by testing in accordance with Article X of the Power Purchase Agreement. For purposes of determining AIDC, the tested capacity of the Complex shall be adjusted to Reference Conditions in accordance with Table 1 of this Schedule 4. In the event of a shortfall in tested capacity of the Complex exceeding ten percent (10%) of the Estimated Dependable Capacity, KESC shall have the right to reject the Complex.

# (b) Reliability Run

A reliability run will be carried out as part of the Commissioning tests and must be satisfied prior to the Complex being certified Commissioned by the Engineer. The run will be for a period of 7 days (168 hours) and will include seventy-two (72) continuous hours at maximum continuous rating ('MCR' which shall be 10,300\_kWe as measured at the generator terminals as specified by the manufacturer) except during TurboCharger washing of each unit for about fifteen (15) minutes when the Unit load will be allowed to be reduced to twenty percent (20%) of MCR when required but not earlier than 68 operating hours. The output during the remaining 96 hours of the test will be as requested by KESC. The test shall have been satisfactorily completed only if it continues without interruption for not less than 168 hours.

# (c) <u>Automatic Voltage Regulator (AVR) Droop</u>

The AVR will be demonstrated to control the generator voltage over the range of  $\pm 10$  percent of rated voltage with a droop characteristic of  $\pm 0.5$  percent.

# (d) Engine Governor Operation

The operation of the engine speed governor will be demonstrated over its range, the droop being adjusted from 1.0 percent to 10.0 percent.

11- Ale

# (e) <u>Reactive Capacity</u>

Tests will demonstrate the capability of the Complex to operate at rated voltage and frequency at power factors and under reactive conditions as follows:

,	Lagging	Leading
100% output	104 40 MVAR	72,00 MVAR
0% output	144.00 MVAR	91.20 MVAR

(f) Minimum Load Capability

Tests will demonstrate the capability of the Complex to be operated at six percent (6%) percent of the gross capacity of the Complex specified in Section 3 of Schedule 1 while the Units and auxiliaries remain in a stable and controlled condition.

(g) Response of Complex to Step Load Changes

The Complex shall be demonstrated to be capable of a step increase in Despatched load of 20 % provided the Complex load is greater than 30% of gross capacity of the Complex specified in Section 3 of Schedule 1. It shall also be demonstrated to be capable of withstanding a sudden loss of demand of 10 percent of such gross capacity from any load in the range 30 percent to 100 percent of such gross capacity. The Complex must not trip and must otherwise remain in a safe condition.

### (h) <u>Full Load Rejection</u>

Tests shall demonstrate the ability of the Complex and its auxiliaries to withstand full and part load rejection and remain in a safe condition.

The Company's obligation to conduct such test is subject to the provision by KESC of the required load.

# TABLE 1

# ADJUSTMENT OF POWER OUTPUT

When the diesel engine generating sets (each a 'Unit') are operated under conditions different from the Reference Conditions, the output of the Units shall be adjusted in accordance with the following Adjustment Formulae as per ISO 3046/1;

Eqn. 1	ujuo - cito	$P_x = \alpha P_r$	
•			
Eqn. 2		$\alpha = k - 0.7 (1 - k) [(1/\eta_m) - 1]$	
Egn. 4		$k = (p_x / p_{ra})^m (T_{ra} / T_x)^n (T_{cra} / T_{cx})^q$	
Eqn. 5		$p_{\rm ra} = p_{\rm r} \left( \pi_{\rm r} / \pi_{\rm max} \right)$	
Symbols:	$\mathbf{p}_{\mathbf{x}}$	<ul> <li>power output under Reference Conditions</li> <li>power output under the conditions being considered</li> <li>power adjustment factor</li> </ul>	
	α k	<ul> <li>power adjustment factor</li> <li>ratio of indicated power</li> </ul>	
	ո ղ <sub>ու</sub>	= mechanical efficiency, equal to 0.9 for the Units	
	$p_{\rm X}$	<ul> <li>total barometric pressure condition being considered</li> </ul>	
	$p_{\rm ra}$	= substitute reference total barometric pressure given by Eqn. 5	
	T <sub>ra</sub>	= reference absolute air temperature	
. •	$T_x$	≈ absolute air temperature being considered	
	$T_{cra}$	<ul> <li>reference absolute charge air coolant temperature</li> </ul>	
	Т <sub>сх</sub>	<ul> <li>absolute charge air coolant temperature at the charge air cooler being considered</li> </ul>	inlet
	Pг	<ul> <li>reference total barometric pressure</li> </ul>	
	$\pi_r$	the boost pressure ratio at declared power under standard reference conditions stated by the manufacturer	rence
	$\pi_{max}$	= the maximum available boost pressure ratio stated by the manufactur	er
Note 1:		$p_{\rm ra} = p_{\rm r}  {\rm as}  \pi_{\rm r}  {\rm is}  {\rm equal to}  \pi_{\rm max}$	
Note 2:		In the above formulae $T_{ra} \leq T_x$ and $p_x \leq p_{ra}$ and $T_{cra} \leq T_{cx}$ and $k \leq 1$	
Note 3:	Note 3: Exponents: $m = 0.7$ ; $n = 1.2$ ; $q = 1.0$		
Note 4:		Subscripts: $r = Reference$ Conditions ; $x = conditions$ at the time of the	e test
Note 5:		The Reference Conditions are as stated in Schedule 1	
	/	- Mb Mb ~ 1	

### METERING STANDARDS AND TESTING

# 1. <u>Provision of Tariff Metering</u>

The metering points to record the MWh and MVARh exchange between the Complex and the KESC Grid System shall be as shown in an appropriate diagram to be provided by the Company. The current and voltage transformers will measure current and voltage on the outgoing busbar of the 132 kV switchyard of the Complex. The meters owned by KESC will be located at the Complex in a location mutually agreeable to the Parties. Any photographic facilities will be provided by the Company as part of the verification process for monthly meter readings.

The Metering System and the Back-up Metering System shall be to a mutually agreed international standard providing a measured accuracy of  $\pm 0.5\%$ .

- 2. <u>Testing</u>
  - 2.1 The calibration of meters will be checked to ensure that the accuracy remains within the specified limits.

The method of calibration and frequency of tests will be agreed between the Company and KESC based on knowledge of the performance and the design of the installed meters and the manufacturers' recommendations.

2.2 Compensation will be made for the errors of current and voltage transformers in the meter calibration or during the computation of records. Current and voltage transformers will be tested for ratio and phase angle errors following manufacture at an accredited testing station in the presence of representatives from the Company and KESC. Test certificates issued by the testing station will be issued independently to both parties.

2.3 Testing and calibration of the Metering System shall be carried out by KBSC after giving appropriate notice to the Company in line with the agreed frequency of testing or in the event of either party having reasonable cause to believe the meters are outside specified limits. During such tests and calibration the Company shall have the right to have a representative present at all times.

Ub Als

2.4 Testing and calibration of the Back-up Meter System shall be carried out by the Company as in paragraph 2.3 above. During such tests and calibration KESC shall have the right to have a representative present at all times.

106

e

Mb Ale

Γ

H.

r--

.

# TARIFF, INDEXATION AND ADJUSTMENT

# Contents

PART I: THE REFERENCE TARHF. 109					
A ESTABLISHMENT OF THE REFERENCE TARIFF. 109					
1. 2.	Introduction and Definitions				
B DESCRIPTION OF REFERENCE TARIFF					
3. 4. 5. 6.	Capacity Purchase Price       110         Energy Purchase Price       111         Supplemental Charges and Supplemental Tariff       111         Assumptions       112				
C CAPACITY PURCHASE PRICE					
7. 8	Payment of Capacity Purchase Price       115         Capacity Purchase Price Components       115				
D ENERGY PURCHASE PRICE					
9.	Energy Purchase Price				
E S	UPPLEMENTAL CHARGES				
10.	Supplemental Charges       117         10.1       Pass-Through Items       117				

PART II: INDEXATION/ADJUSTMENT PROVISIONS						
11.	Indexation Factors	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,				
	the Me	107	1			

12.	Metho	Methodology to be used for Indexation		
	12.1	Fuel Price Factor	118	
	12.2	Inflation and Exchange Rate Indexation Factor	119	
	12.3	Exchange Rate Adjustment Factor		
	12.4	Exchange Rates	. 120	
	12.5	Inflation Factors	120	
	12.6	Replacement of an Index	,120	
13,			121	
	13 1	Energy Purchase Price	121	
	13,2	Capacity Purchase Price	123	
	13.3	Bonus and Liquidated Damages	124	
		· · ·		
PART III: PASS-THROUGH ITEMS			127	
14.	Items	Payable on the Basis of Actual Cost	127	
PART	r <b>IV:</b> Al	DJUSTMENT FORMULAE	128	
15.	Calculation of Inflation and Exchange Rate Factors 12			

# TABLES

ļ

1

I Schedule of Reference Tariff and its Components for each Agreement Year

# ANNEXES

I Pro-forma monthly invoice for Capacity Purchase Price

II Pro-forma invoice for Energy Purchase Price

III Pro-forma monthly invoice for Supplemental Charges

ple Me

# PART I: THE REFERENCE TARIFF

# A -- ESTABLISHMENT OF THE REFERENCE TARIFF

# 1. Introduction and Definitions

Ē

- 1.1 This Schedule 6 shall be read in conjunction with, and is subject to, the provisions of Article IX of the Power Purchase Agreement of which this Schedule 6 is a part. To the extent that any provision of this Schedule 6 is inconsistent with any provision of Article IX, the provision of Article IX shall prevail. References to Articles and Sections are to Articles and Sections of this Schedule 6 unless indicated otherwise. References to Tables and Annexes are to the Tables and Annexes to this Schedule 6.
- 1.2 Payments to be made to the Company under the Power Purchase Agreement shall be calculated in accordance with this Schedule 6, and adjusted from time to time as provided herein.
- 1.3 The procedures for the presentation and payment of invoices as set out in Article IX of the Power Purchase Agreement shall apply to all invoices referred to in this Schedule 6.
- 1.4 Definitions

Capitalized terms used and not defined herein are used herein as defined in the Power Purchase Agreement. Without prejudice to the generality of Section 1.1, for the purposes of this Schedule 6 the following words and phrases shall bear the meanings ascribed thereto:

"Customs Duties" - has the meaning ascribed thereto in the Implementation Agreement;

"E," - The Exchange Rate Indexation Factor, as calculated pursuant to Section 15.2;

"Exchange Risk Insurance" - has the meaning ascribed to that term in the Implementation Agreement;

<u>"Fuel Supply Agreement"</u> - The fuel supply agreement dated 199 between the Company and the Pakistan State Oil Company Limited, as in effect on the date hereof or as amended from time to time with the prior written consent of KESC.

pilo Me

::

"<u>Foreign Lenders</u>" - the lenders providing finance to the Company under the Loan Agreements that are not incorporated under the Laws of Pakistan;

" $\mathbb{E}_{\underline{i}}$ " - The Inflation and Exchange Rate Indexation Factor, as calculated pursuant to Section 15.1;

"Loans" - means the loans made by any of the Lenders to the Company;

"Period" - the period from the Commercial Operation Date through 1 January or 1 July, whichever is earlier, and each period of six Months beginning on 1 January and 1 July thereafter.

"Reference Date" - 1 January 1994; and

"<u>Reference Tariff</u>" - the Capacity Purchase Price and the Energy Purchase Price for each Agreement Year during the Term as each would be computed as of the Reference Date and according to the assumptions stated herein.

"RFO" - Residual Fuel Oil

- 2. <u>Establishment</u>
- 2.1 The reference tariff (the "Reference Tariff"), and each component thereof, computed on the basis of the assumptions stated herein, for each Agreement Year during the Term is shown in Table I.
- 2.2 The actual tariff (the "Tariff") payable in any Period shall be determined by applying the provisions of this Schedule 6 to the Reference Tariff.
- 2.3 All Adjustments to the Reference Tariff that are identified to occur at Financial Closing shall occur at the earlier to occur of the Financial Closing or the Commercial Operations Date.

### **B** -- DESCRIPTION OF REFERENCE TARIFF

3. <u>Capacity Purchase Price</u>

The Capacity Purchase Price shall comprise the following components:

3.1 The Escalable Component as set out in Section 8.1; and When Miles M

- 3.2 The Non-Escalable Component as set out in Section 8.2;
- 4. Energy Purchase Price

The Energy Purchase Price shall comprise the following components:

- 4.1 The Fuel Cost Component as set out in Section 9.1(a); and
- 4.2 The Variable O&M Costs Component as set out in Section 9.1(b).
- 5. Supplemental Charges and Supplemental Tariff
- 5.1 Supplemental Charges

In addition to the Capacity Purchase Price and the Energy Purchase Price, the Company shall be entitled to submit each Month an invoice to KESC and receive payment for any Pass-Through Items identified in Section 14.1 that have been incurred in the immediately preceding Month.

- 5.2 Supplemental Tariff
  - 5.2(a) If, due to a Pakistan Political Force Majeure Event or a Change in Law, a Supplemental Tariff is due and payable to the Company from KESC as provided in Sections 13.6 and 13.7 of the Power Purchase Agreement, the Company shall invoice KESC for payment of the Supplemental Tariff. When any necessary restoration or modification to the Complex is complete and the Complex has returned to operation, the Supplemental Tariff will be payable to the Company by KESC.
  - 5.2(b) Supplemental Tariff payments will be structured to permit the Company to recover costs (including financial costs) of restoration or modification of the Complex over a reasonable term approved by the GOP, which term shall not be greater than the term of the debt secured by the Company to effect the restoration or modification of the Complex, and for the Company (or KESC) to recover in each Agreement Year the cost of any increases (or decreases) in consumables directly attributable to a Change in Law. Any Supplemental Tariff Payment amounts to recover debt incurred in connection with the restoration or modification. Any Supplemental Tariff Payment to recover the return on additional equity contributions made in connection with the restoration of the Complex shall be escalable in accordance with

the file 111

Section 13.2(a) <u>provided however</u>, that in that event the Reference Date for the Inflation and Exchange Rate Indexation Factor shall be the weighted average date on which the additional equity is invested into the Company for the restoration of the Complex. Supplement Tariff Payments to recover the cost of increased (or decreased) consumables shall be fixed prior to the date the Complex return to operation (or if the Complex does not cease operation, at a date agreed by the Company and the GOP) and shall thereafter be subject only to indexation in conformity with Section 13.1(a) or Section 13.1(b) as applicable, <u>provided however</u>, that the Reference Date shall be the date the Complex does not cease operation, at a date agreed by the Complex shall be the date the Complex returns to operation (or if the Complex does not cease operation, at a date agreed by the Complex shall be the date the Complex returns to operation (or if the Complex does not cease operation, at a date agreed by the Complex does not cease operation, at a date agreed by the Complex does not cease operation, at a date agreed by the Complex does not cease operation, at a date agreed by the Complex does not cease operation, at a date agreed by the Complex does not cease operation, at a date agreed by the Company and the GOP).

5.2(c) Invoices for Supplemental Tariff Payments shall be submitted by the Company to KESC at any time after the first Day of the Month in which such payments are due and shall show the due date to be thirty (30) Days after the delivery of the invoices.

# 6. Assumptions

The Capacity Purchase Price and the Energy Purchase Price of the Reference Tariff are based upon the following assumptions set out in paragraphs 6.1 to 6.11 below, established as at the Reference Date (the "Assumptions"):

- 6.1 The Complex will burn RFO supplied by the Fuel Supplier under the Fuel Supply Agreement and the price of such RFO shall be the price of RFO delivered at the Complex as established by the Director General (Oil) Ministry of Petroleum and Natural resources, GOP, on the Reference Date.
- 6.2 Withholding tax on dividends on shares of the Company shall be payable at the rate of 7.5 percent and the Government of Pakistan shall exempt shareholders in the Company from any withholding tax in excess of 7.5 percent.
- 6.3 The Company shall be exempt from the payment of Iqra and Import License Fees on all plant and equipment imported into Pakistan for incorporation into the Complex through the Commercial Operations Date. All Octroi charges before and after the Commercial Operations Date shall be payable by the Company without adjustment of the Tariff.
- 6.4 The Company shall be liable to withhold and pay to the GOP as full and final tax liability of the Contractors (and sub-contractors) @ 4% of the relevant payments made by the Company to the Contractors plus 4% of the relevant payments made by the Contractors to their direct sub-contractors so that (i) the Company shall deduct income tax of @4% from its relevant

Hb Mk

payments to the Direct Contractors; (ii) the Direct Contractors will deduct 4% from the relevant payments to the direct sub-contractors.

- 6.5 The Company, provided it shall remain incorporated in Pakistan, for the sole purpose of power generation, shall be exempt from the payment of income tax on its income from the Complex for the term of the Power Purchase Agreement.
- 6.6 Interest payments, mark-up and fees payable to Foreign Lenders under the Financing Documents shall be exempt from withholding tax.
- 6.7 Exemptions shall be granted from the payment of Customs Duties, sales tax and surcharges as specified in Section 18.1(b) of the Implementation Agreement, including spare parts imported prior to the Commercial Operations Date, which shall be imported free of all duties/taxes. After the Commercial Operations Date, imported spare parts not manufactured in Pakistan, as reasonably determined by the Private Power and Infrastructure Board, will be subject to duty of 20%, and if manufactured in Pakistan but nonetheless imported by the Company, will be subject to duty at the then-prevailing rates.
- 6.8 Zakat

- 6.8(a) For so long as the majority of the shareholders of the Company are non-Muslims or non-residents of Pakistan, the GOP shall exempt the Company from the payment of *Zakat*.
- 6.8(b) Non-Muslim and non-Pakistan resident shareholders shall be exempt from the payment of *Zakat* on dividends paid by the Company.
- 6.9 If the O & M Contractor is a Pakistan subsidiary of a foreign company.
  - 6.9(a) income tax will be payable by such subsidiary on the income derived by the parent company from the provision of technical services to the O & M Contractor at the rate of 20 percent of the gross revenue derived from the provision of such services; and
  - 6.9(b) the subsidiary shall be exempt from income tax on its profits in Pakistan.
- 6.10 If the O & M Contractor is a Pakistan branch of a foreign company, it shall pay income tax on its profits derived from the O&M Agreement in accordance with the Laws of Pakistan.

Who Alk

### 6.11 Change in Assumptions

If and to the extent that any of the Assumptions shall be or shall become invalid or inapplicable or shall vary, adjustments to the Reference Tariff shall be calculated in accordance with the following provisions:

6.11(a)

To the extent that changes in the assumptions set out in paragraphs 6.2 to 6.10 give rise to variations in expenditure, or, in the case of paragraph 6.2 or 6.4, to any variation in withholding tax payable by the Company in accordance with paragraph 6.2 or 6.4, such variations in expenditure or such amounts paid by way of withholding tax (net of any tax refunds or rebates received by the Company) shall be Pass-Through Items and, on and after the Commercial Operations Date, shall be reimbursed by an exact payment equal to the relevant variation in expenditure or withholding tax from KESC to the Company and, in the case of changes in the assumptions that reduce an applicable expenditure or withholding tax below the amount or percentage specified in the assumptions, from the Company to KESC. These changes shall be effected as described in Section 10.1. Changes in such assumptions occurring prior to the Commercial Operations Date shall earn interest at the average rate of interest accruing under the Financing Documents from the date paid by the Company through the Commercial Operations Date, and the Supplemental Tariff Payments shall be structured to allow the recovery of such expenditures through the first five (5) Agreement Year's of the Power Purchase Agreement or, in the case of reductions in expenditures or withholding taxes, paid by the Company to KESC over the same term.

6.11(b)

Changes in the assumption set out in Section 6.1 shall not fall within the provisions of Section 6.11(a). Changes in the assumptions referred to in Section 6.1 shall be dealt with as provided in Section 13.1(a).

US Ale

### C -- CAPACITY PURCHASE PRICE

# 7. Payment of Capacity Purchase Price

The Capacity Purchase Price shall be established on the basis of a sixty percent (60%) plant factor as provided in the Policy March 1994. The Capacity Purchase Price will be payable by KESC to the Company each Month for each kW of Dependable Capacity and each Monthly invoice delivered pursuant to Section 9.7(a) of the Power Purchase Agreement shall include the Monthly amounts for the components set out in Section 8 below and all adjustments made thereto from the Reference Tariff in sufficient detail to allow KESC to confirm the accuracy of such adjustments. A proforma monthly invoice is included at Annex I.

8. <u>Capacity Purchase Price Components</u>

8.1 Escalable Component.

This component includes the fixed operations and maintenance cost, the insurance cost, the administrative cost and the return on equity, etc. The escalable component of the Capacity <u>Purchase Price as set forth in Table 1 is only adjustable against change, if any, proposed in the</u> Energy Price by the Company. Once determined at the earlier of Financial Closing or the Commercial Operations Date, the base figure will remain unchanged during the term of the Power Purchase Agreement and only indexation will be provided in accordance with the procedure set forth in Section 13.

# 8.2 Non-Escalable Component.

(a) The Non-Escalable Component as set forth in Table 1 includes the debt servicing charges, including payments of principal, interest and other fees to the Company's lenders. This component will decline with the passage of time as the loans related to the Project are repaid. Once adjusted at the earlier of Financial Closing or the Commercial Operations Date, pursuant to Section 13.2(b) this component will be indexed as provided in Section 8.2 (b) below.

(b) The Company has elected Indexation of the Non-Escalable Component pursuant to Section 9.1 (d) of the Power Purchase Agreement and, therefore, the non-esclabale component (less the local currency loan portion thereof, if any) will be indexed for changes in the value of the Rupee against the Dollar from the earlier of the date of Financial Closing or the Commercial Operations Date.

4

### D -- ENERGY PURCHASE PRICE

### 9. <u>Energy Purchase Price</u>

9.1 The Energy Purchase Price as set forth in Table 1 is adjustable against change, if any, proposed by the Company in the Escalable Component of the Capacity Purchase Price. Once determined at the earlier of the date of Financial Closing or the Commercial Operations Date, the components of the Energy Purchase price shall only be adjusted as specifically provided herein. The Energy Purchase Price comprises two components, each relating to the energy output of the Complex, as follows:

9.1(a) Fuel Cost Component.

Escalation in Fuel Cost Component will be provided over the term of the Power Purchase Agreement on the basis of changes in the fuel prices, maintaining the Guaranteed Heat Rate (thermal efficiency) as constant, and with respect to RFO, an assumed calorific value of 9,700 (net) kCal/Kg.

9.1(b) Variable O&M Costs Component.

This component includes the costs of operation and maintenance that are attributable to the operation of the Complex and are recoverable on a per kWh basis. Escalation of the Variable O&M Costs component will be provided over the term of the Power Purchase Agreement as provided in Section 13.1(b)

9.2 Each invoice submitted to KESC pursuant to Article IX of the Power Purchase Agreement in respect of the Energy Purchase Price shall include the amount per kWh for the Energy Purchase Price components multiplied by the number of kWh's delivered by the Company in the Month to which the relevant invoice relates. A pro-forma invoice for the Energy Purchase Price is set out in Annex II.

No Ale

### E -- SUPPLEMENTAL CHARGES

### 10. <u>Supplemental Charges</u>

F

The Company may submit to KESC, on a Monthly basis, invoices for such of the following Supplemental Charges as the Company shall have incurred in the immediately preceding Month. With respect to the Supplemental Charges set out in Section 10.1, invoices may be submitted by the Company to KESC at any time after the first Day of the Month following the Month in which any such-Supplemental Charges are incurred and shall show the due date to be thirty (30) Days after the delivery of the invoices. A pro-forma Monthly invoice for the Supplemental Charges referred to in Sections 10.1 is set out in Annex III.

### 10.1 Pass-Through Items

"Pass-Through Items" pursuant to Part III of this Schedule 6 shall be treated as Supplemental Charges for the purpose of invoicing and payment on the basis of the actual cost or charge incurred by the Company therefor as agreed between the Parties or, failing agreement, as determined by an expert pursuant to Sections 15.2 and 15.3 of the Power Purchase Agreement. Unless otherwise provided in Section 14, the Company shall present an invoice to KESC for a Pass-Through Item on the first Day of the Month following the Month in which the cost, charge, or other liability was incurred by the Company and shall show the due date of the invoice to be thirty (30) Days after its delivery.

### PART II: INDEXATION/ADJUSTIMENT PROVISIONS

# 11 Indexation Factors

- 11.1 Pursuant to this Schedule 6, certain items as specified herein shall be indexed in accordance with the provisions of this Part II of this Schedule 6.
- 11.2 If, at any time, an index is applied under this Part II to any component of the Reference Tariff, either Party may seek to verify the application of the relevant index and may require the other Party, where appropriate, to provide to it copies of its calculations, with reasonable supporting information for the application of the relevant index, to enable that Party to verify the result of applying the relevant index to the relevant component of the Reference Tariff.
- 11.3 The escalable components of the Reference Tariff shall be adjusted upward or downward, as appropriate, to reflect changes in the indexation or adjustment formula, as the case may be, according to which each such component is to be indexed or adjusted, as specified below.
- 11.4 The Non-Escalable Component, less any local debt portion thereof, as established on the earlier of the Date of Financial Closing or the Commercial Operations Date, shall be adjusted-upward or downward by application of the Exchange Rate variation of the Rupees to Dollars as specified in Section 13.2(b)(ii).
- 12. Methodology to be used for Indexation

Except as provided in this Schedule 6, the base date from which all values are to be indexed pursuant to this Schedule 6 is the Reference Date.

The methodology applicable to the calculation of each of the indices listed below shall be in accordance with this Article 12.

### 12.1 Fuel Price Factor

The Fuel Price Factor is the index to be used to calculate variations in those elements of the Fuel Cost Component or the Supplemental Charges with respect to variations in the price per unit of fuel (per Tonne of RFO delivered at the Complex) as established pursuant to the Fuel Supply Agreement at any date "p", and is calculated as follows:

AK

### Fuel Price Factor = Fuel Price<sub>p</sub> + Fuel Price<sub>REF</sub>

where:

Fuel  $Price_p =$  the price per unit of finel as established pursuant to the Fuel Supply Agreement (after correction for standard temperature and expressed in Rupees), at date "p", and

Fuel  $Price_{RBP}$  = the price per unit of fuel as established by the GOP (after correction for standard temperature and expressed in Rupees), at the Reference Date.

In calculating the price of RFO, the accounting convention of first-in/first out shall be employed.

Subject to the immediately preceding sentence, the Fuel Price Factor shall be applied to those elements stated to be subject to Indexation for Fuel Price from the date when a change in the price of fuel as established pursuant to the Fuel Supply Agreement is effective.

# 12.2 Inflation and Exchange Rate Indexation Factor

The components of the Reference Tariff stated to be subject to indexation for inflation and exchange rate movements in Section 13 below shall be adjusted with effect from the Reference Date in accordance with the Inflation and Exchange Rate Indexation Factor (as defined in Section 15.1) on the Commercial Operations Date and on each 1 January and 1 July thereafter using the most current exchange rate and inflation rate information available on that date. No retroactive adjustments will be permitted in the event that an index figure is subsequently revised. The adjusted Tariff payable to the Company in any Period, p, will be calculated by multiplying the Reference Tariff components subject to indexation for inflation and exchange rate movements by the Inflation and Exchange Rate Factor from the Reference Date through the calculation date, t, immediately preceding Period, p.

# 12.3 Exchange Rate Adjustment Factor

山此

The Non-Escalable Component of the Capacity Purchase Price less any local loan portion thereof (as shown at Financial Closing) is subject to indexation for exchange rate movements pursuant to Section 13.4(b), in each Agreement Year and shall be adjusted with effect from the earlier to occur of the date of the Financial Closing or the Commercial Operations Date, in

119

**7**.

accordance with Exchange Rate Indexation Factor (as defined in Section 15.2) on the Commercial Operations Date and on each 1 January, 1 April, 1 July and 1 October thereafter using the most current exchange rate and inflation rate information available on that date. No retroactive adjustments will be permitted in the event that an index figure is subsequently revised. The adjusted Tariff payable to the Company in any Period, p, will be calculated by multiplying the Non-Escalable Component, less any local loan portion thereof, of the Reference Tariff by the Exchange Rate Indexation Factor from the earlier to occur of the date of the Financial Closing or the Commercial Operations Date, through the calculation date, t, immediately preceding Period p.

## 12.4 Exchange Rates

The source for exchange rates for Dollars against the Rupee shall be the quotation by the State Bank of Pakistan in the most recently published Exchange Rates Bulletin issued by the Foreign Exchange Rates Committee of the Authorised Dealers' Spot T.T. & O.D. Selling Rate (as defined in the SBP Exchange Control Manual).

# 12.5 Inflation Factors

The source indices for indexation against inflation shall be the Consumer Price Index (IFS CPI), reported in the publication of the International Monetary Fund entitled "International Financial Statistics" for the United States and Pakistan.

### 12.6 <u>Replacement of an Index</u>

In this Schedule 6, where an element is stated to be subject to indexation by reference to a specified index or factor and at any time such index is withdrawn, becomes unavailable for any reason or becomes, in the reasonable opinion of KESC or the Company, inappropriate as a basis for indexation pursuant to this Schedule, then upon written notice from either Party, KESC and the Company shall use their best endeavors to identify a mutually acceptable alternative index.

Jul Ale
If after fourteen (14) Days the Parties are unable to agree on an alternative index to be substituted for the index that requires replacement pursuant to this Section 12.6, the Parties shall appoint an expert pursuant to Section 15.2 of the Power Purchase Agreement who shall nominate an index that in his sole opinion most adequately replaces the withdrawn, unavailable, or inappropriate index within fourteen (14) Days after his appointment. The Parties shall be bound by the determination of the expert as to the alternative index.

In the event that (1) the GOP no longer establishes the delivered price of RFO or (2) an index or factor, used to compute or index the same element in the Fuel Supply Agreement and in this Schedule 6, is no longer available and an alternate therefor is not established or approved by the GOP, then (1) an index to establish the price of RFO and changes thereof or (2) an alternate index or factor, as the case may be, acceptable to KESC, the Company and the Fuel Supplier shall be used under this Schedule 6. If KESC, the Company and the Fuel Supplier cannot agree to an alternate index or factor, as the case may be, within sixty (60) Days, an expert acceptable to KESC, the Company and the Fuel Supplier will nominate an alternate index or factor, as the case may be, that shall be binding on the Parties. KESC, the Company and the Fuel Supplier may refer the matter to the National Electric Power Regulatory Authority (NEPRA) for final resolution in lieu of referral to the expert. The expert or NEPRA, as the case may be, will be requested to make a final determination within sixty (60) Days after such referral.

Pending the substitution of an alternative index, no indexation adjustment shall be made with respect to the relevant index. Upon the substitution of an alternative index, the Parties shall make the indexation adjustment with the alternative index retrospectively to the date when the relevant adjustment would otherwise have been made.

## 13. Application of Indices to Variables

The escalable components of the Tariff shall be indexed in accordance with the formulae set out in this Article 13.

13.1 Energy Purchase Price

Ľ

13.1(a) The Fuel Cost Component ("FCC") of the Energy Purchase Price,

Variations to the FCC shall be calculated as provided in the following formula and at the times indicated in this Section 13.1(a):

 $FCC_{\circ} = FCC_{(RUF)} * Fuel Price Factor,$ 

S.

. .

where:

 $FCC_{e} =$  the Fuel Cost Component on the relevant calculation date, c.

 $FCC_{(REID)}$  = the value for FCC for the relevant Agreement Year given in Table I.

Fuel Price Factor has the meaning set out in paragraph 12.1.

13.1(b) The Variable Operations & Maintenance Costs Component.

This component is escalable against exchange rate variations of the Rupee to the Dollar and United States inflation rate. For determining the Exchange Rate, State Bank of Pakistan's TT&OD selling rate will apply. For determing the inflation rate the Consumer Price Index (CPI) of the United States as published by the International Monitory Fund (IMF) in the International Financial Statistics (IFS) will be applied.

The base date for application of indexation will be the Reference Date. The indexation will apply prospectively on the Commercial Operations Date and on each 1 January and 1 July thereafter.

The variable operations & maintenance costs element shall be indexed as follows:

 $O\&M_{\circ} = O\&M_{(REF)} * IE_{1}$ 

where:

 $O\&M_{o}$  = the value of the Variable Operations and Maintenance Costs component of the Tariff per kWh, expressed in Rupees, as adjusted at the relevant calculation date, c,

 $O\&M_{(REF)}$  = the value of the Variable Operations and Maintenance Costs component of the Reference Tariff per kWh, expressed in Rupees, in the relevant Agreement Year, as shown in Table I, and

 $IE_t$  = the value of the Indexation and Exchange Rate Factor ("IE<sub>t</sub>") as calculated in Section 15.1 at the Indexation date, t, immediately preceding the calculation date.

Ul Me

## 13.2 <u>Capacity Purchase Price</u>

## 13.2(a) Escalable Component:

The escalable component ("ESC-C") of the Capacity Purchase Price is directly escalable against exchange rate variations of the Rupee to the Dollar and the United States inflation rate. For determining the Exchange Rate, State Bank of Pakistan's TT&OD selling rate will apply. For determining the United States inflation rate the Consumer Price Index (CPI) of the United States as published by the International Monetary Fund (IMF) in the International Financial Statistics (IFS) will be applied.

The base date for application of indexation will be the Reference Date. The indexation will apply prospectively on the Commercial Operations Date and on each 1 January and 1 July thereafter.

The Escalable Component of the Capacity Purchase Price shall be indexed as follows:

 $ESC_{*}C_{e} = ESC_{(RBP)} * IE_{(....)}$ 

where:

 $ESC-C_{c}$  = the value of the Escalable Component of the Tariff, expressed in Rupees per kW per Month, as adjusted at the relevant calculation date, c,

 $ESC-C_{(RBP)}$  = the value of the Escalable Component of the Reference Tariff in the relevant Agreement Year, expressed in Rupees per kW per Month, as shown in Table I, and

 $IE_t$  = the value of the Indexation and Exchange Rate Factor ("IE<sub>t</sub>"), as calculated in accordance with Section 15.1 at the Indexation date, t, immediately preceding the calculation date.

## 13.2(b) Non-Escalable Component:

All All

(i) The Non-Escalable Component of the Capacity Purchase Price will be adjusted at earlier of the Financial Closing or the Commercial Operations Date, for the variation in exchange rates of Rupee to Dollar from the Reference Date to the date of adjustment. The adjustment factor for this component will be equal to 0.7% for each 1% variation in Dollar to Rupee exchange rate. The Non-Escalable Component (less

123

any local loan portion thereof) will be further indexed during the Term in accordance with clause (ii) below.

(ii) The base date for the application of such indexation will be the earlier of the date of the Financial Closing or the Commercial Operations Date. The indexation of the Non-Escalable Component (less any local loan portion thereof) will apply prospectively on the Commercial Operation Date and on each 1 January, 1 April, 1 July and 1 October thereafter.

The Non-Escalable Component shall be indexed as follows:

 $N-ESC_{c} = N-ESC_{(REF)} * E_{t}$ 

where:

N-ESC<sub>c</sub> = the value of the Non-Escalable Component of the Tariff (less any local loan portion thereof), as adjusted at the relevant calculation date, c.

N-ESC<sub>(REF)</sub> = the value of the Non-Escalable Component of the Reference Tariff (less any local loan portion thereof) in the relevant Agreement Year, as adjusted at the earlier of the date of Financial Closing or the Commercial Operations Date pursuant to Section 13.2(b)(i), and

 $E_t$  = the value of the Exchange Rate Factor (" $E_t$ ") as calculated in Section 15.2 at the indexation date, t, immediately preceding the calculation date.

## 13.3 Bonus and Liquidated Damages

The bonus payments payable to the Company pursuant to Section 9.6 of the Power Purchase Agreement and the liquidated damages payable to KESC by the Company pursuant to Section 9.5 of the Power Purchase Agreement shall be indexed for inflation and exchange rates as follows:

13.3(a) Bonus

(i) The bonus payments described in Section 9.6 of the Power Purchase Agreement shall be indexed for exchange rate variations of the Rupee to the Dollar and for United States inflation. The base date for application of indexation will be the Reference Date. The indexation will apply prospectively on the Commercial Operations Date and on each 1 January and 1 July thereafter.

Ub Mes 124

 $B_1 = B_{(IMID)} * IE_1 * K$ 

where:

Ľ

 $B_1$  = the value of the relevant bonus payment as adjusted at the relevant payment date,

 $B_{0NTD}$  = the value of the relevant bonus payment per kWh as shown in Section 9.6 of the Power Purchase Agreement,

 $\mathbb{E}_t$  = the value of  $\mathbb{E}_t$  at the indexation date, t, immediately preceding the calculation date.

K = the number of kWh's for which a bonus payment is applicable during the relevant Agreement Year pursuant to the terms of Section 9.6 of the Power Purchase Agreement.

(ii) The Energy Purchase Price Premium provided in Section 9.3 of the Power Purchase Agreement is payable in Rupees at the Rupee to Dollar Exchange Rate in effect immediately prior to the date of calculation, c, and is not subject to any escalation or indexation.

13.3(b) Liquidated Damages

The liquidated damages payable under Section 9.4(b) and 9.4(c) of the Power Purchase Agreement shall be indexed on the dates as set out in Section 15.1 as follows:

 $LD = LD_{(INID)} * IE_{t}$ 

where:

LD = the value of the relevant liquidated damages payment as adjusted at the relevant calculation date,

 $LD_{(INIT)}$  = the initial value of the relevant liquidated damages payment as computed in accordance with Section 9.4(b)&(c) of the Power Purchase Agreement.

 $IE_1 =$ the value of  $IE_1$  at the indexation date, t, immediately preceding the calculation date.

The indexation will apply prospectively on the Commercial Operations Date and on each 1 January and 1 July thereafter.

The liquidated damages payable under Section 9.4(a) shall be indexed on the date set out in Section 15.2 as follows:

 $LD = LD_{(INIT)} * I_t$ 

 $I_1 = P(US)t / P(RD)$ 

where:

 $LD_{(NTT)}$  = the initial value of the relevant liquidated damages payment as computed in accordance with Section 9.4(a) of the Power Purchase Agreement.

 $P_{(US)}$  = the average value of the end of Month values for the IFS CPI index for the United States over the six Months immediately prior to the date of calculation; <u>provided</u>, that if values for any of the Months are not available, then the average of the values of the most recent six Months for which figures are available shall be used.

 $P_{(RD)}$  = the value of the IFS CPI index for the United States at the Reference Date.

Illo Ales

## PART III: PASS-TBROUGH ITEMS

## 14. <u>Items Payable on the Basis of Actual Cost</u>

14.1 The items contained in this Part III of this Schedule 6 to the extent attributable to the Project ("Pass-Through Items") shall be payable by KESC to the Company on the basis of the actual cost incurred by the Company (or, in the case of Sections 14.2(d), the Contactors) and shall be invoiced by the Company paid by KESC as supplemental charges. Unless otherwise provided in this Schedule 6, the Company shall invoice KESC for obligations incurred by the Company with respect to such Pass-Through Items at such time as the Company incurs the liability for such Pass-Through Item, and any invoice presented by the Company to KESC for these purposes shall show the due date of such invoice to be twenty-five (25) Days after the delivery of the invoice to KESC. KESC shall pay any such invoice in accordance with Article IX of the Power Purchase Agreement.

## 14.2 Pass-Through Items

July Ale ~

The costs and charges constituting Pass-Through Items hereunder shall be:

- 14.2(a) payments into the Workers' Welfare Fund and the Workers' Profit Participation Fund ;
- 14.2(b) reasonable costs approved by KESC incurred by the Company after Commercial Operations Date for modifications or expansion of the requirements for protective devices required by KESC; and
- 14.2(c) Customs Duties on permanent equipment which are levied on the Company prior to the Commercial Operations Date and which are not refunded to the Company by Central Board of Revenue ("CBR") after the issuance of the consent of the CBR specified in Schedule 2 of the Implementation Agreement.

14.2(d) The tariff does not account for the 0.5% Turnover Tax on revenues and if it is to be paid by the Company during operations, the Tariff will be increased accordingly.

14.2(e) Taxes referred to in Section 18.4 of the Implementation Agreement.

14.2(f) Any penalty incurred by the Company for changes in fuel deliveries under Section 6.3(c) of the Fuel Supply Agreement as a result of changes in KESC's

Despatch levels that are recoverable by the Company pursuant to Section 6.2(d) of the Agreement.

14.2(g) Transportation Costs payable to the Fuel Supplier under Section 7.5 of the Fuel Supply Agreement in the event of a change in the Freight Pool Rules as determined pursuant to Section 12.6.

#### PART IV: ADJUSTMENT FORMULAE

## 15 <u>Calculation of Inflation and Exchange Rate Factors</u>

15.1 The Inflation and Exchange Rate Indexation Factor for adjustment as at the Commercial Operations Date and every 1 January and 1 July thereafter is as follows:

(i) 
$$IE_t = \frac{P_{(US)t}}{P_{(RD)}} * \frac{FX_{(US/P)t}}{FX_{(RD)}}$$

where:

All Mb

 $IE_1$  = the Inflation and Exchange Rate Indexation Factor applicable for the Period following date of calculation, t.

 $FX_{(US(P))}$  = the average exchange rate of the Dollar against the Rupee over the six Months prior to the date of calculation. This average will be calculated as the sum of the exchange rates (as described in Section 12.4 of this Schedule 6) for each of the last days of the six Months prior to date of calculation divided by six.

 $FX_{(RD)}$  = the exchange rate of the Dollar against the Rupee at Reference Date which shall be deemed for all purposes herein to be 30.03 Rupees to one Dollar.

 $P_{\text{(US)t}}$  = the average value of the end of Month values for the IFS CPI index for the United States over the six Months immediately prior to the date of calculation; <u>provided</u>, that if values for any of the Months are not available, then the average of the values of the most recent six Months for which figures are available shall be used.

 $P_{(RD)}$  = the value of the IFS CPI index for the United States at the Reference Date.

128

15.2 The Exchange Rate Adjustment Factor for adjustment as at the Commercial Operations Date and every 1 January, 1 April, 1 July and 1 October thereafter are as follows:

(i) 
$$E_{I} = \frac{FX_{(US/P)I}}{FX_{(RD)}}$$

where:

11L

715

 $E_t$  = the Exchange Rate Adjustment Factor applicable for the Period following date of calculation, t.

 $FX_{(USP)}$  = the average exchange rate of the Dollar against the Rupee over the three Months prior to the date of calculation. The average will be calculated as the sum of the exchange rates (as described in Section 12.5) for each of the last days of the three Months prior to the date of calculation divided by three.

 $FX_{(RD)}$  = the exchange rate of the Dollar against the Rupee at the earlier to occur of the date of the Financial Closing or the Commercial Operations Date.

Sheetf

۵۰۰۰۹ | |\_\_\_\_

]

( ) ....)

: : .

[ [...]

UNDICATIVE UNLY

# TAPAL ENERGY LIMITED

.

119.50 MW

Livits Sold at 60% Plant Factor.

Estimated Initial Dependable Capadity

		ENERG	ENERGY PURCHASE PRICE	FRICE		8	CAPACITY CHARGE	Ĩ		TOTAL		Dravo	CAPACITY PURCHASE PRICE	PRICE			
	1	<u>دد</u>	Rupees per kluh	ų.		ы.	Rupees per kWh	£	**	LARIFF Rs/kmh		Kupe	Rupecs per KW / MONIH	HING		20203	L
-	Torrestor					Non-Fis	Non-Fscalable		Totaí			Non-Escaiable	culable		Tetal	Completion	Enargy
		e z	Vartable	Total	Escalable	Compartent	pherit	FERI	Capacity		Escalatic	Component	опелт	FERI	Capacity	Boláte	Erroading 03%
	-		O&M		Component	LDCal	Farehm		Charge		Component	Local	Forelan		Charge	Dae. 39, 1297	Copacity Foctor
	ļ	1020	0.454	0.780	0.253	0.000	0,270	0.000	1,133	1.913	115.200	03970	361.100	0,000	196.300	0.075	0.080
_	- F		175	0.780	0.263	000.0	0.235	0.000	67.1 1	1.905	115,200	0,000	375.300	0.000	484.500	0.075	0,080
_	46	1630	0.156	3.7ED	2362	0,000	0.843	0.008	1,105	1,825	115,200	0,000	368.200	0.000	101-127	0.075	0.080
_	, <b>x</b>		0.135	0.760	0.2153	0.000	0.625	0,000	1.05.1	1.871	115.200	0:000	362.700	0.000	008.772	0.075	0.080
	r u	100	0.156	0.780	0.263	0.000	0,811	0,000	1-074	1.854	115.200	0.00.0	355.200	0.000	\$79,400	0.075	0.080
-	ьĸ	0.874	0.156	0.780	6.263	0,000	0.783	000	1.049	1.823	115.200	0.000	344.300	0,000	458.500	0.073	030'0
	> r	1624	0.156	0.780	0,263	0.000	0.769	0.000	1.012	1.812	115,200	0.000	042-302	0.00.0	452,600	0.075	080.0
-	- 03	0.624	0.156	0.780	0.263	0.000	0,767	0,000	1,030	013.1	115,200	0,000	335,300	0.880	451.160	0.075	0.080
2	, <del>о</del>	7690	0.156	0.760	0,283	0.000	127,0	0.000	1,020	1.800	115,200	0.000	121.000	0.000	446,800	570-0	0.050
et,	, <del>¢</del>	1674	9156	0.780	0.263	0:000	0,565	0.000	0.228	1.608	115.200	0,000	247.500	0.000	362.700	0.075	0,080
	; ;	100	1410	0.782	1220	0.000	0.495	0.000	0.745	1.534	109-300	0,386	216.200	0.000	226,700	54070	0,030
	ţ	164 U	0.164	0.763	0.251	0.000	0,480	0,000	0,731	1.513	006"601	0,000	210.200	0,000	320.100	0.000	0.030
i	1 Č	Tar	1154	0.732	0.251	0.000	0,210	0.008	0,567	1.349	109-800	0,000	135.890	00010	245,700	0.000	0.030
	2.2	PC9 0	0.164	0.783	1520	0,000	0,102	0,600	0.353	1.141	109.300	0.000	122	0.000	154,800	0.000	0.030
	ų	- 7230	0.164	0.738	1221	0:00	0.079	0,001	0,230	1.115	109-300	0,000	34,600	0,000	144,500	0000	0.020
	ų į	0.624	0161	0.788	1.251	0.000	0,079	0.000	0.230	1.118	102.300	0,000	34,630	0.000	144.500	0.000	02010
	1	0.624	0,164	0.738	13210	0.000	670-0	0,001	0.330	1.138	103.900	0.000	109772	0,000	144.500	00079	0.030
	с С	0.624	0.16¢	0.7\$\$	527	0.000	0.079	0.000	0.330 -	1.178	109,900	0.000	00976	0.000	144.500	0.000	0.20.0
	5	0.624	0.164	0.788	527	0:000	620°0.	0,000	002710	1.118	109.200	0,000	34.600	0.000	144.500	0.000	0.050
	1	0.624	0,154	0,785	0.251	0,000	0.079	0.000	0.330	1,113	103,200	0.000		00010	144,500	0000	0207B
	- 	0.624	0.164	0.736	1,25,1	0.000	6L0-0	0,000	0.530	1.113	102.200	0.000	24.800	0.000	14.500	0000	1989.B
	ន	0.624	0.164	0.73.6	0.251	0.000	0.078	0.000	0.330	1,118	103.500	0000	34,500	0.000	005 **1	00070	0.000
, Y						CONCO IN TH	I POWER SILPCHASE AGBERT	420 HA ANUC	CHARLES .								1
が	년 상 (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	NAUZZUAN															J
		Atoma Tado				2 USE	Xoe Cametry F	Mide Mide									
	Rs/kub		US Cents/kwh			Peritow Months	Perkyw Months US\$	US\$/kw/Manth									and the second
-A	1.829		80.9 20.¥			459.5 133.6		15.3 6,11									•

130 µb

Page 1

Tariff for Project Ute: Rupses 1.668/kwth (US DENTS 5.55/kwth)

## ANNEX I

# Pro forma monthly invoice for Capacity Purchase Price

NAME AND ADDRESS OF COMPANY]

KESC; Aimai House, Abdullah Haroon Road Karachi, Pakistan]

## INVOICE FOR [INSERT MONTH AND YEAR] CAPACITY PAYMENT

Invoice No: .....

Invoice Date:

Capacity Purchase Price for the Month commencing \_\_\_\_\_\_ is due as follows:

(1) Non-Escalable Component	Rs
(2) Escalable Component <sup>1</sup>	Rs
(3) Total Capacity Purchase Price	Rs
(4) Current Dependable Capacity (in MegaWatts)	MW
(5) Total Capacity Payment	Rs
(6) Less Liquidated Damages (net of Bonus) as per Bonus and Liquidated Damages Notice No dated [or]	Rs ()
(6) Plus Bonus (net of Liquidated Damages) as per Bonus and Liquidated Damages Notice no dated	R\$ ()
(7) Total Adjusted Capacity Payment	Rs

This invoice is payable in full on or before the twenty-fifth (25th) day following the date of this invoice.

1 Calculation of the escalation of Escalable Component of the Reference Capacity Purchase Price shall be shown in reasonable detail. Jus Ala

131

## ANNEX II

Pro forma Invoice for Energy Purchase Price

[NAME AND ADDRESS OF COMPANY]

[KESC; Aimai House, Abdullah Haroon Road Karachi, Pakistan]

## INVOICE FOR [INSERT MONTH AND YEAR] ENERGY PAYMENT

Invoice No...... Invoice Date \_\_\_\_\_

Energy Purchase Price due for the Period from \_\_\_\_\_\_ to \_\_\_\_\_\_to

(a) Fuel Costs Component<sup>2</sup>
(b) Variable Operations and

Maintenance Cost Component<sup>3</sup>

(c) Total Energy Purchase Price

(e) Total Energy Payment

(d) Total Net Electrical Output Delivered in kWh

Date of Initial Meter Reading
 Initial Meter Reading
 Date of Final Meter Reading
 Final Meter Reading
 Wh

Final Meter Reading

Rs.....

Rs.....

Rs.....

Rs.....

....kWh

Payment of this amount is to be made in full on or before the twenty-fifth (25th) day following the date of this invoice.

 $^2$  Calculation of adjustment of Fuel Cost Component of the Reference Energy Purchase Price shall be shown in reasonable detail. Include with this invoice the relevant invoices from the Fuel Supplier.

 $^3$  Calculation of escalation of Variable O&M component of the Reference Energy Purchase Price shall be shown in reasonable detail.

Who Ale

## ANNEX III Pro forma monthly invoice for Supplemental Charges

## [NAME AND ADDRESS OF COMPANY]

[KESC; Aimai House, Abdullah Haroon Road Karachi, Pākistan]

## INVOICE FOR SUPPLEMENTAL CHARGES

Invoice N	o.
1111 (100 11	VI

Invoice Date: .....

Rs .....

Rs .....

Supplemental-Charges for the month ending ...... are due as follows:-

Pass-Through Items:

..... (description of item(s))

**Total Supplemental Charges** 

Payment of this amount is to be made in full within 30 days of this invoice. (payment and account details to be inserted)

immediately preceding the calculation date.

flb Alle

## CONSTRUCTION REPORTS

[Format of Construction Reports to be supplied later by the Company]

pht

1-

1

i

.

## **INSURANCE**

## PART I: CONSTRUCTION PERIOD

## 1. <u>Marine and Air Cargo</u>:

<u>Cover:</u> All materials, equipment, machinery, spares and other items for incorporation in the Complex against all risks of physical loss or damage while in transit by sea or air from country of origin anywhere in the world to the Site in Pakistan, or <u>vice versa</u>, from the time of the insured items leaving warehouse or factory for shipment to the Site. Cover to institute Cargo Clauses (Air), institute War Clauses (Air), (Sendings By Post), institute Strikes Clause (Cargo, Air Cargo) or equivalent.

Sum insured:	An amount equal to cost and freight of any shipment
Deductible:	US\$ 10,000 each loss.
Insured:	The Company, the Contractors and suppliers to the Company and to the Contractors

## 2. Loss of Revenue Profits (following Marine incident);

<u>Cover</u>: Against loss of revenue following delay in start of commercial operations as a direct result or physical loss or damage to the materials, equipment, machinery and other items in transit by sea or air to the Site, to the extent covered under the Marine Cargo insurance.

Sum insured: An amount equal to the estimated Capacity Purchase Price which will be received following the Commissioned Date of the Complex.

Indemnity Period: 12 months.

Deductible: 30 days.

Insured:

The Company and the Lenders.

pllo Ale

## . <u>Contractors' All Risks</u>:

<u>Cover</u>: The contract Works executed and in the course of execution, materials and temporary works, while on the Site, against all risks of physical loss or damage <u>other than</u> war and kindred risks, nuclear risks, unexplained shortage, cost of replacing or repairing items which are defective in workmanship, material or design; penalties; consequential losses; cash; vehicles; vessels; aircraft. Cover shall provide the equivalent terms, conditions and perils/causes of loss provided under an Erection All Risks insurance policy.

Sum insured:

Deductibles;

In relation to Contract Works, Materials etc.

The Contract Price.

(a) arising during the Construction and Testing period:

- (i) from Storm, Tempest, US\$ 10,000
   Flood, Water Damage,
   Earthquake, Tsunami,
   Subsidence and Collapse
- (ii) from any other cause US\$ 5,000 [other than in (a)(i) above]
- (b) arising out of operational testing or commissioning: US\$ 35,000

<u>Period of Cover:</u> Actual construction, testing and commissioning until expiry of the warranty period.

Insured: The Company, the Contractors and all suppliers and consultants, GOP, KESC, PSO, and the Lenders.

<u>General:</u>

During the warranty period, cover shall be limited to the loss or damage for which the Construction Contractor is liable under the warranties of the Construction Contract. Cover shall include transit within Pakistan of locally procured materials. Cover shall cease, and be transferred to Operating Period insurance, on the day following the Commercial Operations Date.

Hile Mk

3.

## 4. Loss of Revenue (following C.A.R.):

Ľ

<u>Cover</u>: Against loss of revenue following delay in start of commercial operations as a direct result of physical loss of, or damage to, the Works during construction or operational testing to the extent that such loss or damage is covered under the Contractors' All Risks policy.

<u>Sum insured:</u>	An amount equal to the estimated Capacity Purchase Price which will be received following the Commissioned Date of the Complex.
Indemnity Period:	12 months.
Insured:	The Company, Lenders and O&M Contractor.
Deductible:	30 days.
Period of Cover:	Actual Construction, testing and commissioning periods of the Project from mobilization of the Contractors until the day following Commercial Operations Date.

## 5. <u>Public Liability:</u>

<u>Cover</u>: Against legal liability to third parties for bodily injury or damage to property arising out of the construction, testing and commissioning of the Complex in Pakistan.

Sum insured:	For any one claim: US\$ 5,000,000.
Deductible:	Not to exceed US\$ 25,000 for each claim for damage to property. None for injury to persons
Insured:	The Company, Contractors, all suppliers and consultants, GOP, KESC, and PSO
<u>Period of Cover:</u>	The actual construction, testing and commissioning of the Complex from mobilization of the Contractors until the day following Commercial Operations Date.

t

~

## 6. <u>Miscellaneous:</u>

Other insurance as is customary, desirable or necessary to comply with local or other requirements, such as Workmen Compensation Insurance in relation to all workmen employed in the construction of the Project and Motor Insurance on any vehicle.

plo Ale

#### PART II. OPERATING PERIOD

## 1. <u>All Risks Insurance - Fixed Assets:</u>

Ē

<u>Cover</u>: All building contents, machinery, stock, fixtures, fittings and all other personal property forming part of the Complex against "All Risks" of physical loss or damage, including (but not limited to) those resulting from fire, lightning, explosion, spontaneous combustion, storm, wind, tempest, flood, hurricane, water damage, riot, strikes, malicious damage, earthquake, tsunami, collapse and/or loss of contents of tanks.

 Sum insured:
 Full replacement value of the Complex.

 Deductible:
 Not to exceed US\$ 50,000 each loss.

Insured: The Company, the O&M Contractor, GOP, KESC, PSO and the Lenders.

2. Consequential Loss Following All Risks:

<u>Cover:</u> Loss of revenue due to loss of capacity and/or loss of output as a direct consequence of loss of or damage to the Complex and caused by a peril insured under paragraph 1 above.

Sum insured: An amount equal to the estimated Capacity Purchase Price which will be received during the Indemnity Period following the current period of the insurance policy.

Indemnity Period: 15 months.

Deductible: First 15 days of any interruption.

Insured: The Company, the O&M Contractor and the Lenders.

<u>General:</u> Rights of recourse against KESC, PSO, the Lenders and O&M Contractor shall be waived.

## 3. Machinery Breakdown:

<u>Cover:</u> All machinery, plant, boilers (if any) and ancillary equipment forming part of the Complex against sudden and unforeseen physical loss or damage resulting from mechanical and electrical breakdown or derangement, explosion or collapse of boilers (if any) and pressure

Nb Atts

vessels(if any), electrical short circuits, vibration, misalignment, excessive current or voltage, abnormal stresses, centrifugal forces, failure of protective or regulating devices, overheating, entry of foreign bodies, impact, collision and other similar causes.

Sum insured:	Full replacement value of all machinery, plant, boilers(if any), etc.
Deductible:	US\$ 10,000 each loss.
Insured:	The Company, the Lenders, KESC and the O&M Contractor.

4. Consequential Loss following Machinery Breakdown:

<u>Cover:</u> Loss of revenue due to loss of capacity and/or loss of output as a direct consequence of loss or damage to the Complex caused by a peril insured under paragraph 3 above.

Sum insured:An amount equal to the estimated Capacity Purchase Price which will<br/>be received during the Indemnity Period following the current period of<br/>the insurance policy.Indemnity Period:15 months.Deductible:First 30 days of any interruption.Insured:The Company, the O&M Contractor and the Lenders.General:Rights of recourse against KESC, PSO, the Lenders and O&M<br/>Contractor shall be waived.

## 5. Public Liability:

ę

<u>Cover</u>: Legal liability of the insured for damage to property of third parties or bodily injury to third parties arising out of the ownership, operation and maintenance of the Complex.

Sum insured: US\$ 5,000,000 for any occurrence.

Deductible:

Not to exceed US\$ 25,000 each claim for property. None for injury to persons.

Insured:

The Company, the O&M Contractor, the Lenders, GOP, KESC and PSO.

~ 1

~ Jub Me

## 6. <u>Miscellaneous:</u>

Ľ

,\_\_\_] . Other insurance as are customary, desirable or necessary to comply with local or other requirements, such as Workmen Compensation Insurance in relation to all workmen employed in the Complex or in connection with its operation, and Motor Insurance on any vehicle.

7. Indexing of Limits;

The coverage provided under Section 5 in Part I and Part II will be indexed in accordance with the Schedule 6.

Ule Me

## FORM OF LETTER OF CREDIT

## [ISSUED ON ISSUING BANK LETTERHEAD SHOWING FULL NAME AND ADDRESS]

Date and Place of Issue:

Applicant
Name The Power Company, Limited
Address \_\_\_\_\_\_, Pakistan

Advising and Negotiating Bank [name and address] \_\_\_\_\_, Pakistan

<u>Beneficiary</u> KESC [address]

\_\_\_\_\_, Pakistan

Attention:

We hereby issue our documentary credit as follows: <u>Type of Credit</u>: Irrevocable

Letter of Credit Number:

Date and Place of Expiry:

Date Place

~~~

[Advising and Negotiating Bank name and address]

file the

<u>Amount</u> [figures] [words]

Who Ale

Credit Available with: [Advising and Negotiating Bank], by negotiation against presentation of the documents detailed herein and of your draft(s) at sight drawn on Issuing Bank accompanied by a certificate signed on your behalf by a person describing himself therein as your duly authorized officer-stating that:

A. "This drawing in the amount of [currency and amount] is being made pursuant to the Power Purchase Agreement (Agreement) between The \_\_\_\_\_\_ Power Company, Limited ("Company") and KESC as a result of Company's failure to perform in accordance with Article/Section \_\_\_\_\_\_ of the Agreement."

## OR.

B. "KESC is making a drawing in the full available amount of [Issuing Bank] Letter of Credit No
 because the term of the Letter of Credit will expire within ten (10) business days of the date of this certificate and The \_\_\_\_\_\_ Power Company, Limited ("Company") has failed to deliver a replacement or renewal Letter of Credit acceptable to KESC, and security is still required under the terms of Article/Section \_\_\_\_\_\_ of the Power Purchase Agreement between the Company and KESC, dated \_\_\_\_\_\_199\_."

Presentation of either of the above certificates and all communications in writing with respect to this Letter of Credit shall be addressed to us at [Issuing Bank name and address] referencing Letter of Credit No. \_\_\_\_\_, Attention: \_\_\_\_\_, or at [Advising and Negotiating Bank name and address] referencing Letter of Credit No. \_\_\_\_\_, Attention: \_\_\_\_\_\_,

This Letter of Credit sets forth in full the terms of our undertaking and this undertaking shall not in any way be modified, amended, limited, or amplified by reference to any document, instrument, or agreement referred to herein, except only the certificates and draft referred to herein; and any such reference shall not be deemed to incorporate herein by reference any document, instrument, or agreement except for such certificates.

This Letter of Credit is transferable. Transfer may be effected only by Issuing Bank upon our receipt of an acceptable application for transfer accompanied by the original Letter of Credit and payment of our transfer commission in effect at the time of transfer.

Partial drawings are allowed.

Tested telex reimbursement is allowed.

Drafts drawn under this Letter of Credit must bear the clause:

"Drawn under [Issuing Bank] Letter of Credit No. \_\_\_\_\_, dated \_\_\_\_\_\_, 199\_,"

It is a condition of this Letter of Credit that it shall be automatically extended for an additional period of one year from the present and each future expiration date, unless, thirty (30) days prior to the thencurrent expiration date, we notify you by registered mail that this Letter of Credit will not be renewed for an additional period.

We hereby engage with you that drafts drawn strictly in compliance with the terms of this credit and amendments shall meet with due honor upon presentation. This credit is subject to "Uniform Customs and Practice for Documentary Credits" (1983 Revision), International Chamber of Commerce, Publication No. 400.

Mbs My

Authorised Signature

Authorised Signature

# TESTING SCHEDULE

|    | Stage                                | Month |
|----|--------------------------------------|-------|
| 1. | Construction Start                   | 0     |
| 2. | Target Completion of Interconnection | 10    |
| 3. | Scheduled Commercial Operations Date | 13    |
|    | Mb Alk                               | 1     |

E.

....

 $\int$ 

e.,

<u>~</u>~~

Amendment No.1 dated May 9, 1996

to the

Power Purchase Agreement

#### between

The Karachi Electric Supply Corporation Limited

and Tapal Energy Limited

dated September 26, 1995,

This Amendment No.1 dated May 9, 1996 to the above-referenced Power Purchase Agreement (the "Agreement") is made at Karachi between The Karachi Electric Supply Corporation Limited and Tapal Energy Limited.

## Whereas

- (1) On September 26, 1995, the parties entered into the Agreement.
- (2) The parties now desire to amend and clarify certain provisions of the Agreement pursuant to this Amendment No.1 to the Agreement.

NOW IT IS HEREBY AGREED as follows:

A. Definitions. Capitalized terms used and not defined herein shall have the same meaning as ascribed thereto in the Agreement. The deletion of any section and/or subsection shall not affect the numbering of subsequent sections and/or subsections, unless specified herein.

B. Amendments to the Agreement.

- 1 -

 Section 1.21 of the Agreement is hereby deleted and replaced in its entirety by the following:

"Company" means Tapal Energy Limited, a public limited company; or a private unlimited company into which Tapal Energy Limited may be converted or re-formed, incorporated under the Laws of Pakistan with its principal office located at Karachi and its permitted successors and assigns."

- Section 1.28 of the Agreement, "Construction Financial Closing" is hereby deleted in its entirety.
- Section 1.51 of the Agreement, "Financial Closing" is hereby deleted and replaced in its entirety by the following:

"1.51 "Financial Closing" means (a) the execution and delivery of one or more loan agreements that together evidence the financing for the construction, testing and completion and the permanent financing of the Complex, in each case, following the resolution of any objection raised by the GOP to a term sheet or schedule in accordance with Section 15.3 of the Implementation Agreement that sets out the principal repayment schedule and the other principal terms of the transaction between the Company and the Lenders or following the date on which the GOP shall be deemed to have not objected to such term sheets and principal repayment schedule pursuant to Section 15.3 of the Implementation Agreement and (b) the receipt of commitments for such equity as is required by the Company in order to satisfy the requirements of the Lenders, and the Letter of Support".

1 Ab

Ľ

- Section 3.3 (a) (i) of the Agreement is hereby amended by deleting the words "earlier to occur of either Construction Financial Closing or" in the first and second lines of that Section.
- Section 3.3 (a) (ii) of the Agreement is hereby amended by deleting the words "earlier to occur of either Construction Financial Closing or" in the first and second lines of that Section.
- Section 3.3 (a) (xi) of the Agreement is hereby amended by deleting the words
  - "earlier to occur of either Construction Financial Closing or" in the second line of that Section and
  - (ii) "earlier to occur of either Construction Financial Closing or" in the third and fourth lines of that Section.
- Section 3.3 (a) (xii) of the Agreement is hereby amended by deleting the words "earlier to occur of either construction Financial Closing or" in the first and second lines of that Section.
- Section 3.3 (a) (xiii) of the Agreement is hereby amended by deleting the words "Construction Financial Closing and again at" in the first line of that Section.
- Section 3.3 (a) (xv) of the Agreement is hereby amended by deleting the words "earlier to occur of either Construction Financial Closing or" in the first and second lines of that Section.
- Section 3.3 (d) of the Agreement is hereby amended by deleting the words "earlier to occur of either Construction-Financial Closing or" in the first and second lines of that Section.

- 3 -

- 12. Section 4.2 (a) of the Agreement is hereby deleted in its entirety.
- Section 4.2 (b) of the Agreement is hereby deleted in its entirety and replaced with the following:
  - "(ii) the failure of the Company to achieve Construction Start and satisfy all conditions precedent to the initial availability of funds necessary to complete construction of the Complex under the Financing Documents within ninety (90) Days following the date of Financial Closing".
- 14. Section 4.2 (k) of the Agreement is hereby amended by deleting the phrase " an Election Notice" (as defined in Section 4.5)" and inserting in its place "a Succession Notice (as defined in Section 4.5)".
- 15. Section 4.5 of the Agreement is hereby amended by:

AL

- 4 -

]

Ĵ

Î

- (I) inserting in the third line of the first paragraph of that Section, following the word "Company" and before the word "without" the parenthetical phrase "(other than a Company Event of Default under Section 4.2 (k)".
- (ii) deleting the words "earlier to occur of either Construction Financial Closing or" in the the second lines of the first paragraph of that Section.

(iii) deleting the words "Construction Financial Closing and" in the fourteenth line of the first paragraph of that Section.

- (iv) deleting the words "Construction Financial Closing and" in the twenty second and twenty third line of the first paragraph of that Section.
- (v) inserting in the first line of the second paragraph of that Section, following the word "KESC" and before the word "shall" inserting the parenthetical phrase "(other than as a result of a Company Event of Default under Section 4.2 (k)".
- (vi) deleting the words "Election Notice" in the fifteenth and sixteenth lines of the second paragraph of that Section and inserting in its place "Succession Notice".
- 16. Section 4.8 of the Agreement is hereby amended by adding before the first words "the exercise" add the brackets and letter "(a)" which shall indicate new numbering. All the contents of existing Section 4.8 shall hereinafter be numbered 4.8 (a), and this shall apply consistently throughout the document wherever reference to Section 4.8 is made.

The following new paragraph shall be added to the same section hereinafter referred to as 4.8(b).

"4.8.(b)Notwithstanding Section 4.8 (a), the Paries agree that the Company may be damaged in amounts that may be difficult or impossible to determine in the event that this Agreement is terminated by the Company as a result of a KESC Event of Default. Therefore the Parties have agreed that the termination of this Agreement and the termination of the Implementation Agreement by the Company and the payment in full to the Company by the GOP of the compensation provided under Article XX of the Implementation agreement as a result of a GOP Event of Default (as defined in the Implementation Agreement) is in lieu of actual damages and any other liquidated damages which would -5 - Marc

otherwise be payable under this Agreement in respect of such termination and the collection of such amount (and the termination of the Implementation Agreement) is the sole remedy of the Company in respect of such termination. The provisions of this Section 4.8(b) shall be without prejudice to any right or remedy of the Company which arises prior to termination of this Agreement".

1.1

- Section 5.2(b)(iii) of the Agreement is hereby amended by deleting the words "Construction Financial Closing and the" in the first line of that Section,
- 18. Section 9.4 (f)(i) of the Agreement is hereby amended by deleting subsection/paragraph (A) in its entirety. Existing subsection/paragraph (B) is hereby amended to become new subsection/paragraph (A); existing subsection/paragraph (C) is hereby amended to become new subsection/paragraph (B); existing subsection/paragraph (D) is hereby amended to become new subsection/paragraph (C); Subsequent references to any of the above existing subsection/paragraphs in Section 9.4 (f)(i) are hereby amended to refer tö the new subsections/paragraphs. References to any of the respective above existing subsections/paragraphs in Section 9.4 (f)(i) in this Agreement No.1 shall refer to the existing numbering appearing in the Agreement.
- 19. Section 9.4(f)(i)(D) of the Agreement is hereby amended by inserting the words "or the Company fails to pay any amounts owed to KESC pursuant to Section 3.5 and 4.7, "after the words and punctuation "when due", and before the word "then" appearing in line 2. After the word "Agreement" and before the word, brackets and number "and (2)" appearing in line 6, the words "or amounts due to KESC under Section 3.5 and 4.7" shall be inserted. After the word "invoice" and before the word "for" appearing in line 7, the words "or demand" shall be inserted.

The following shall be added to the end of Section 9.4(f)(I)(D) following the words and punctuation "by the Company" add:

"KESC shall not be entitled to draw any amounts shown in the invoice or demand to the Company that has been disputed by the Company until such amount is determined to be due and payable to KESC in accordance with the dispute resolution procedures set forth in this Agreement. KESC shall also have the right to draw or collect the full available amount of Company Letter of Credit issued pursuant to Sections 9.4(f) (i)(A) or (B) if it is not replaced as required pursuant to this Section 9,4(f) upon presentation of a certificate of an authorized officers of KESC stating that the term of such Company Letter of Credit will expire within ten (10) Business Days of the date of the certificate, and the Company has failed to deliver a replacement Company Letter of Credit In accordance with this Section 9.4(f) and such security is still required under the terms of this Section 9,4(f). If within thirty (30) Business Days of the any draw by KESC of the Company Letter of Credit pursuant to the immediately preceding sentence, the Company provides a replacement Company Letter of Credit in the required amount, KESC shall return to the start Company the amount drawn by it under the Company Letter of Credit less any amount it would have otherwise been entitled to recover from the Company Letter of Credit arising after the date of encashment under the terms of this Agreement".

- 20. Section 13.1 of the Agreement is hereby amended by deleting the words "earlier to occur of Construction Financial Closing or" in the second and third lines of that Section.
- 21. Section 17.10(e) of the Agreement is hereby amended by adding in line two after the words and punctuation "Financial Closing," and before the words "all such" the following phrase: "and at any refinancing approved by the GOP in accordance with Section 15.3 of the Implementation Agreement"

- 7 -

HE

- 22. Table 1 of Schedule 6 of the Agreement is hereby amended, restated and replaced in its entirety by the attached, signed Table 1 ("ATTACHMENT i").
- 23. Schedule 10 of the Agreement is hereby amended, restated and replaced in its entirety by the attached, signed Schedule 10 ("ATTACHMENT II")

## C. <u>Miscellaneous</u>

ATK.

- 8 -

Ē

-

- 1. This Amendment No: 1 shall be binding upon and inure to the benefit of the parties hereto and their respective permitted successors and permitted assigns.
- 2. This Amendment No.1 shall be governed by and construed in accordance with the laws of England.

In Witness Whereof, the parties hereto have caused this Amendment No.1 to the Agreement to be executed and acknowledged by their respective duly authorized officers or representatives, as of the date first above written. ł

THE KARACHI ELECTRIC SUPPLY CORPORATION LIMITED

A KH L A DUULI II MEH General i Menanger (Generation, Projessionadd By: Name: Private PreserCells) R. E. SSCC. Lidd. Title:

## TAPAL ENERGY LIMITED

By:

Name: THEISH TAPAL Title: DIRECTUR

Witnesses:

Ubini-101 - Reiman 2. MOIZ TAPAL

-9-

ATTACHMENT )

I. •7

Contraction of the second second second second second second second second second second second second second s

.

<u>(</u>می

5

į

.

.\*

schwhule of Reference Tariff and its components for each Agreement Year

TAPAL ENERGY LIMITED

.

*:* 

-

;

1

Estimated initial Dependable Capacity

-119.5 MW

,

Units Sold at 60% Plant Factor

| Autennut         Autention         R.A.W.M.           Autennut         Component         FEM         Total         Example         Ferrigin         Total           1         OGXM         P.Y.750         Component         Local         Foreign         Component         Component           -1         0GXM         P.Y.750         Component         Local         Foreign         Component         Component         Local         Foreign         Component         Componen                                                                                                                                                 |              |          | ENERGY PURCHASE PRICE<br>Rupees per kWir | se price<br>'n | {<br>}<br>} | ; ~      | CAPACITY CHARGE<br>Rupees per kWh | Y CHAR(<br>ser kyyh | ; <del></del> | TOTAL<br>TARIFF |           | CAPACITY CHARGE<br>Rupees per KW /Month | RGE            |               |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------|----------|------------------------------------------|----------------|-------------|----------|-----------------------------------|---------------------|---------------|-----------------|-----------|-----------------------------------------|----------------|---------------|
| Triel         Variable         Total         Escalable         Non-Escalable         Total         Foreign         Component         FERI         Total         Escalable         Component         Escalable         Component         Escalable         Component         Escalable         Component         Escalable         Component         Component         Escalable                                                                                                       | Agreeme      | 13E      |                                          |                |             | •        |                                   | 7                   |               | Rs. RWh         |           | 5                                       |                |               |
| Fuel         Variable         Total         Component         FER         Total         Escatible           0.6500         0.1850         0.1850         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.57100         0.00000         0.77550         0.00000         0.0550         1.7545         99.4260           0.65000         0.1850         0.1850         0.122700         0.00000         0.0551         1.75457         99.4260         94.260         94.260         94.260         94.260         94.260         94.260         95.250         94.260         95.250         94.260         95.250         94.260         95.250 <td>Year</td> <td></td> <td></td> <td>_</td> <td></td> <td>1 Non-És</td> <td>calable  </td> <td></td> <td></td> <td>1</td> <td></td> <td>Non-Escálable</td> <td></td> <td></td> | Year         |          |                                          | _              |             | 1 Non-És | calable                           |                     |               | 1               |           | Non-Escálable                           |                |               |
| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |              | 2 3      |                                          |                |             |          | ment                              |                     |               |                 |           | Component                               |                |               |
| OEMI         Component         Local         Foreign         Capacity         Capacity           0.6500         0.1850         0.1850         0.22700         0.00000         0.53710         0.0000         0.53700         99,4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.53800         0.00000         99,4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.53800         0.00000         99,4260           0.6500         0.1850         0.22700         0.00000         0.53800         0.00000         1.655         1.8800         99,4260           0.6500         0.1850         0.22700         0.00000         0.53800         0.00000         99,4260           0.6500         0.1850         0.22700         0.00000         0.53800         99,4260         99,4260           0.6500         0.1850         0.22700         0.00000         0.71550         90,4260         99,4260           0.6500         0.1850         0.22700         0.00000         0.539470         0.00000         94,260           0.6500         0.1850         0.22700         0.00000         0.539470         0.00000         94,260      <                                                                                                                                                                                                                                                            |              | [Juc]    | Variable                                 | Total          | Escalable   |          |                                   | FER                 | Total         |                 | Escalable |                                         |                | F             |
| 0.6500         0.1850         0.27700         0.00000         6.87100         0.00000         1.055         1.8700         99.4260           0.6500         0.1850         0.1850         0.227700         0.00000         0.53800         0.0055         1.8800         99.4260           0.6500         0.1850         0.1850         0.227700         0.00000         0.53800         0.0055         1.8800         99.4260           0.6500         0.1850         0.1850         0.227700         0.00000         0.33800         0.0055         1.8800         99.4260           0.6500         0.1850         0.1850         0.227700         0.00000         0.33800         0.0055         1.8700         99.4260           0.6500         0.1850         0.1850         0.227700         0.00000         0.33800         0.0552         1.4367         99.4260           0.6500         0.1850         0.1850         0.227700         0.00000         0.77850         0.904260         94.260           0.6500         0.1850         0.1850         0.227700         0.00000         0.5521         1.4367         99.4260           0.6500         0.1850         0.227700         0.00000         0.77850         0.00000         0.5                                                                                                                                                                                                                                               | -1.01        |          | O&N                                      | _              | Component   |          | Foreign                           |                     | Capacity      |                 | Component | Local                                   | Capae          | $\frac{1}{2}$ |
| $ \begin{array}{c ccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |              | , 1<br>, |                                          | \$-2120<br>\$  | 1           |          |                                   | •                   | charge        |                 |           |                                         | chart          | 5             |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 7            | 0.6300   | 0.1850                                   | 0.1850         | - 0.22700   | 0.00000  | 87100                             | 0.00000             | 1.098         | 1.9130          | 99.4260   | 0.0000531.4980                          | 0.00000430.92  | 19            |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | Çi           | 0.6300   | 0.1850                                   | 0.1850         | 0.22700     | 0.00000  |                                   | 0.00000             | 1.065         | 1.3800          | 99.4260   | 0.00000 367.0440                        | _              | 00            |
| 0.6300         0.1850         0.1850         0.22700         0.00000         0.83800         0.00000         1.065         1.8800         99.4260           0.6300         0.1850         0.1850         0.22700         0.00000         0.77850         0.00000         1.9545         99.4260           0.6300         0.1850         0.1850         0.22700         0.00000         0.77850         0.00000         1.7545         99.4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.77850         0.00000         1.7545         99.4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.77850         0.00000         1.4367         99.4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.53470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.53470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.53470         0.00000         0.622         1.4367         99.4260           0.6500                                                                                                                                                                                                                                                    | ۲ň           | 0.6300   | 0.1850                                   | 0.1350         | 0.22700     | 0.00000  | <u> </u>                          | 0.00000             | 1.065         | 1.3800          | 99.4260   | 0.00000367.0440                         |                | 00            |
| 0.6300         0.1850         0.1850         0.22700         0.00000         0.77850         0.00000         1.065         1.8800         99.4260           0.6300         0.1850         0.1850         0.22700         0.00000         0.77850         0.00000         1.5205         99.4260           0.6300         0.1850         0.22700         0.00000         0.77850         0.00000         0.7545         99.4260           0.6500         0.1850         0.22700         0.00000         0.77850         0.00000         0.7547         99.4260           0.6500         0.1850         0.22700         0.00000         0.5927         1.4367         99.4260           0.6500         0.1850         0.21700         0.00000         0.5927         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.522         1.4367         99.4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.522         1.4367         99.4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.522         1.4367         99.4260           0.6500         0.1850         0.1850         0.22700                                                                                                                                                                                                                                                                     | <b>.</b> الع | 0.6300   | 0.1850                                   | 0,1850         | 0.22700     | 0.00000  |                                   | 0.00000             | 1.065         | 1.3300          | 99.4260   | 0.00000367.0440                         |                | 00            |
| $\begin{array}{llllllllllllllllllllllllllllllllllll$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | <i>v</i> ,   | 0.6300   | 0.1850                                   | 0.1350         | 0.22700     | 0.00000  | \$3800                            | 0.00000             | 1.065         | 1.8300          | 99.4260   | 0.00000367.0440                         |                | 80            |
| $ \begin{array}{c ccccccccccccccccccccccccccccccccccc$                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | ¢. >         | 0.6300   | 0.1850                                   | 0-1\$50        | 0.22700     | 0.00000  |                                   | 0.00000             | 1.006         | 1.8205          | 99.4260   | 0.00000540.9330                         |                | 6             |
| 0.6500         0.1850         0.22700         0.00000         0.59470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.59470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.59470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.59470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.522         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.522         1.4367         99.4260                                                                                                                                                                                                                                                        | ~ ~          | 0.6300   | 0.1850                                   | 0.1850         | 0.22700     | 0.00000  | _                                 | 0.00000             | 0.939         | 1.7543          | 99.4260   | 0.00000311.9874                         | _              | ñ             |
| 0 65300         0.1850         0.22700         0.00000         0.59470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.522         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.59470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.59470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.522         1.4367         99.4260           0.65300         0.1850         0.22700         0.00000         0.522         1.4367         99.4260                                                                                                                                                                                                                                                                       | <i>8</i> 2   | 0.6300   | 0.1850                                   | 0.1850         | 0.22700     | 0.00000  |                                   | 0.00000             | 0.622         | 1.4367          | 99.4260   | 0.00000 1.72.8786                       | **             | 55            |
| $ \begin{array}{c c c c c c c c c c c c c c c c c c c $                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | \$           | 0 63 00  | 0.1350                                   | 0.1850         | 0.22700     |          | -                                 | 0.00000             | 0.622         | 1.4367          | 99.4260   | 0.00000172.8786                         |                | 5#0           |
| 0.6500         0.1850         0.22700         0.00000         0.59470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.59470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.6522         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.6522         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.6522         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.6522         1.4367         99.4260           0.65300         0.1850         0.22700         0.00000         0.39470         0.00000         0.6222         1.4367         99.4260           0.65300         0.1850         0.227700         0.000000         0.39                                                                                                                                                                                                                                               | 01           | 0.6300   | 0.1850                                   | 0.1350         | 0.22700     | 0.0000   |                                   | 0.00000             | 0.622         | 1.4367          | 99.4260   | 0.00000172.8786                         |                | 꽃             |
| 0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.65300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.65300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.65300         0.1850         0.227700         0.00000<                                                                                                                                                                                                                                                 |              | 0.6300   | 0.1850                                   | 0.1\$50        | 0.22700     | 0.00000  |                                   | 0.00000             | 0.622         | 1.4367          | 99.4260   | 0.00000172.8786                         |                | 140           |
| 0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.000000         0.39470         0.000000 </td <td>12</td> <td>0.6300</td> <td>0.1850</td> <td>0.1850</td> <td>0.22700</td> <td>0.00000</td> <td></td> <td>0.000000</td> <td>0.622</td> <td>1.4367</td> <td>99.4260</td> <td>0.00000172.8786</td> <td></td> <td>ž</td>                    | 12           | 0.6300   | 0.1850                                   | 0.1850         | 0.22700     | 0.00000  |                                   | 0.000000            | 0.622         | 1.4367          | 99.4260   | 0.00000172.8786                         |                | ž             |
| 0.6300         0.1850         0.1850         0.22700         0.00000         0.39470         0.00000         0.6522         1.4367         99.4260           0.6300         0.1850         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.65300         0.1850         0.22700         0.00000         0.5227         1.4367         99.4260           0.65300         0.1850         0.22700         0.00000         0.522         1.4367         99.4260           0.65300         0.1850         0.22700         0.00000         0.5222         1.4367         99.4260                                                                                                                                                                                                                                                    | 9            | 0.6300   | 0.1850                                   | 0.1850         | 0.22700     | 0.00000  |                                   | 0,00000             | 0_622         | 1,4367          | 99.4260   | 0.00000172.8786                         |                | ž             |
| 0.6300         0.1850         0.1850         0.22700         0.00000         0.39470         0.00000         0.6322         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.000000                                                                                                                                                                                                                                                    | *            | 0.6300   | 0.1850                                   | 0.1850         | 0.22700     | 0.00000  |                                   | 0.000000            | 0.622         | 1.4367          | 99.4260   | 0.00000172.8786                         |                | ž             |
| 0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.65300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.65300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.65300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.65300         0.1850         0.22700         0.000000 </td <td>2</td> <td>0.6300</td> <td>0.1850</td> <td>0.1850</td> <td>0.22700</td> <td>0.00000</td> <td></td> <td>0.000000</td> <td>0.622</td> <td>1.4367</td> <td>99.4260</td> <td>0.00000 172.8786</td> <td></td> <td>¥</td>                    | 2            | 0.6300   | 0.1850                                   | 0.1850         | 0.22700     | 0.00000  |                                   | 0.000000            | 0.622         | 1.4367          | 99.4260   | 0.00000 172.8786                        |                | ¥             |
| 0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.000000                                                                                                                                                                                                                                                     | 10           | 0.6300   | 0.1350                                   | 0.1850         | 0.22700     | 0.00000  | _                                 | 0.00000             | 0.622         | 1.4367          | 99.4260   | 0.00000172.8786                         | 0.00000/272.3( | ž             |
| 0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.6222         1.4367         99.4260           0.65300         0.1850         0.22700         0.00000         0.39470         0.00000         0.6222         1.4367         99.4260           0.65300         0.1850         0.22700         0.000000         0.59470         0.000000         0.6222         1.4367         99.4260           0.63000         0.1850         0.22700         0.000000         0.39470         0.000000         0.6222         1.4367         99.4260                                                                                                                                                                                                                                                                                            | 17           | 0.6300   | 0.1850                                   | 0.1850         | 0.22700     | 0.00000  |                                   | 0.000000            | 0.622         | : 4367          | 99.4260   | 0.00000172.8786                         | 0.000001272.3( | ž             |
| 0.6300         0.1850         0.22700         0.00000         0.59470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6500         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.6222         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.6222         1.4367         99.4260                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | <u>6</u>     | 0.6300   | 0.1850                                   | 0.1850         | 0.22700     | 0.00000  | *                                 | 0.000000            | 0.622         | I.4367          | 99.4260   | 0.00000172.8786                         | 0.00000Jzzz(   | 26            |
| 0.6300         0.1850         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.1850         0.22700         0.00000         0.39470         0.00000         0.622         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.6222         1.4367         99.4260           0.6300         0.1850         0.22700         0.00000         0.39470         0.00000         0.6222         1.4367         99.4260                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 61           | 0.6300   | 0.1850                                   | 0.1850         | 0.22700     | 0.00000  |                                   | 0.00000             | 0.622         | 1.4367          | 99,4260   | 0.00000172.8736                         | 0.00000272.30  | ž             |
| 0.6300 0.1850 0.1850 0.22700 0.00000 0.39470 0.00000 0.622 1.4367 99.4260 0.6300 0.632 1.4367 99.4260                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 50           | 0.6300   | 0.1850                                   | 0.1350         | 0.22700     | 0.00000  |                                   | 0.00000             | 0.622         | 1.4367          | 99.4260   | 0.00000172.8786                         | 0.00000272.30  | 24            |
| 0.6300 0.1850 0.1850 0.22700 0.00000 0.39470 0.00000 0.622 1.4367 99.4260                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 5            | 0.6300   | 0.1850                                   | 0.1850         | 0.22700     | 0.00000  |                                   | 0.00000             | 0.622         | 1.4367          | 99.4260   | 0.00000 172.8786                        | 10.00000272.30 | 346           |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 2            | 0.6300   | 0.1850                                   | 0.1850         | 0.22700     |          |                                   | 0.00000             | 0.622         | 1.4367          | 99.4260   | 0.00000172.8786                         | 0.00000272.30  | 246           |

î

· < <del>} . . .</del>

Levellized 22 year lariff US Cents/KWu 0.0557769

<u>Rs/kWh</u> 1\_674930

¥)

÷

## ATTACHMENT II

## SCHEDULE 10

## TESTING SCHEDULE

# Stage/Event

## Month

| 1. | Later of March 17, 1996 or Financial Closing | 0  |
|----|----------------------------------------------|----|
| 2. | Target Completion Interconnection            | 10 |
| З. | Scheduled Commercial Operations Date         | 13 |

Ak

On behalf of KESC:

On behalf of Tapal Energy Limited:

BY:

TITLE:

TITLE:

BY:



THIS AGREEMENT is made at Karachi on this 11th day of July 2006.

#### BY and BETWEEN

TAPAL ENERGY (PRIVATE) LIMITED, a company incorporated in Pakistan with limited liability under the Companies Ordinance 1984, having its principal office at F-25, Block 5, Kehkashan, Clifton, Karachi (hereinafter referred to as the "TEL", which expression shall where the context permits or requires deem to mean and include its successors-in-interests and assigns), of the One Part

#### AND

KARACHI ELECTRIC SUPPLY CORPORATION LIMITED, a public limited company, having its principal office located in Karachi (hereinafter referred to as the "KESC", which expression shall where the context permits or requires deem to mean and include its successors-in-interests and assigns), of the Other Part

(the TBL and KESC shall hereinafter be collectively referred to as the "Parties" and individually as the "Party")

WHEREAS TEL entered into a Power Purchase Agreement with KESC dated 26<sup>th</sup> September 1995, as amended and/or varied from time to time, (the "PPA");

AND WHEREAS the KESC, under Article II of the PPA, agreed to purchase Dependable Capacity of up to 105% of the Estimated Dependable Capacity and the Net Electrical Output of the Complex to the extent that it is dispatched;

AND WHEREAS under the terms of the PPA, the TEL is required to conduct Annual Dependable Capacity Test of the Complex such that at any point of time, the last tested capacity is deemed to be the prevalent Dependable Capacity of the Complex, which is then used to calculate the monthly Capacity Payments, which KESC is liable to pay to the TEL;

-1-

Que

AND WHEREAS on 15<sup>th</sup> May 2001, the TEL conducted Annual Dependable Capacity test ("the Test") for fifth Agreement Year and demonstrated capacity of 126.8 MW at reference conditions which was restricted to 125.475 MW, as required by the PPA, for billing of Capacity Payments;

AND WHEREAS the KESC raised an Invoice Dispute contesting that the set point for the generators should have been put at 10.3 MW instead of 10.8 MW during the Test and started making deductions from the TEL's monthly Capacity Payment invoices by calculating the Capacity Payment amount on a basis different from the tested Dependable Capacity (the "Dispute");

AND WHEREAS in November 2001, the Dispute was referred to an expert pursuant to Art.15 of the PPA (the "Expert"), who agreed with the contention of TEL that the Test was conducted in accordance with the requirements of the PPA (the "Decision");

AND WHEREAS despite the Decision, the KESC disputed subsequent Tests undertaken in accordance with the PPA, and deducted various amounts from the invoices of TEL such that as on 31<sup>st</sup> May 2006 an amount of Rs. 261,910,187/- is due and payable by KESC to TEL in respect of the Dispute (the "Claimed Amount");

AND WHEREAS the Parties now desire to resolve the Dispute and settle the Claimed Amount and to amend the PPA in accordance with the terms of this Agreement;

NOW, THEREFORE, in consideration of the mutual covenants contained herein, it is hereby agreed as follows:

#### 1. INTERPRETATION

1,1, In this Agreement:

- (a) references to Clauses are to the clauses to this Agreement;
- (b) the singular includes the plural and vice versa;
- (c) all headings are for convenience only and shall not constitute a part of, or be used in construing, this Agreement.
- (d) all capitalised terms used in this Agreement shall have the meaning as ascribed to them under this Agreement and all such terms used and not defined in this Agreement shall have the meaning as given to them in the PPA.

#### 2. EFFECTIVE DATE

This Agreement shall be deemed to take effect from 31" May 2006.

#### 3. THE SETTLEMENT

(c)

KESC acknowledges the Claimed Amount and TEL agrees that the payment by KESC to TEL of Rs.100,000,000 (Rupees One hundred million) in the following instalments shall, if punctually so made, be in full and final settlement of the Claimed Amount:

-2-

- (a) Rs.33,000,000 in the month of July 2006
- (b) Rs.33,000,000 in the month of August 2006

R3.34,000,000 in the month of September 2006.

and upon payment by KESC in accordance with this clause 3, TEL shall have no further claims whatsoever against KESC in respect of the Claimed Amount.

#### 4. AMENDMENTS TO THE PPA

The Parties agree that with effect from 31<sup>st</sup> May 2006:

- (a) In Section 2.1(a)(i) of the PPA, the words "one hundred five percent (105%) of the Estimated Dependable Capacity" shall be replaced by "123.5MW ".
- (b) In Section 10.4 (b) of the PPA, the words "105% of the Bstimated Dependable Capacity" shall be replaced by "123.5MW". The Tests shall be undertaken without any restriction on the set point of the generators."
- (c) If KESC desires, TEL shall have the option, but no obligation, to deliver energy to KESC, at any point of time, in excess of the Dependable Capacity.

And the PPA shall be read and construed as amended herein above.

IN WITNESS WHEREOF the Parties have executed this Agreement on the date first . mentioned above.

For and on behalf of TAPAL ENERGY (PRIVATE) LIMITED

TABISH TAPAC Name: \_ Designation: C.EO.

For and on behalf of KARACHI ELECTRIC SUPPLY CORPORATION

Modern NES Aschoon Name:

Designation: C/6

un Josuan WITNESS/1:

Name: RIZWAN AMBAR ALI PESNANI

Address: TAPAL ENERGY (107) LTS

F-25 BLOCK S, KEHKASHAN CLIPTON, KARACHI.

-3-

WITNESS 2:

Name: <u>ABOUL RAUF YOUSUF</u> Address: <u>Director (RAKSP)</u> KESC.